





Technical Reference Manual



This User Manual contains proprietary information of Medoc Ltd. and may not be reproduced in any form without prior written consent from Medoc Ltd.

DC 00076 Q-Sense Technical Reference Manual 12th Edition, Oct. 2018



1 Ha-Dekel St., PO Box 423 Ramat Yishai 3004627, ISRAEL

Tel.: +972-4-9038800 / Fax: +972-4-9038808

E-Mail medoc@medoc-web.com

www.medoc-web.com

Medoc U.S.A

Compass Medical Technologies, Inc. 1502 West Highway 54 - Suite 404 Durham, North Carolina, 27707, U.S.A Tel.: +1-(919) 402-9600 / Fax: +1-(919) 402-9607 E-Mail: medoc@mindspring.com



European Authorized Representative

CEpartner4U BV Esdoornlaan 13 3951 DB Maarn, The Netherlands Phone: +31 343 442.524 / Fax: +31 343 442 162 Mobile: +31 6 516 536.26 E-Mail: office@CEpartner4U.com



This device complies with 93/42/EEC MDD



Table of Contents

1	S	Safety Guidelines and Regulations	. 14
	1.1	User Manual Icons	14
	1.2	Intended use	15
	1.3	Target Population and Contraindications	15
	1.4	Safety and Regulatory Summary	15
	1.5	Safety Requirements	16
	1.5	5.1 Warnings	16
	1.5	5.2 Cautions	17
	1.5	5.3 Equipment Classification	18
	1.6	System Protection	18
	1.6	5.1 System Self-Test	18
	1.6	5.2 Temperature Safety Mechanisms	18
	1.6	5.3 Thermode Detection	18
	1.7	Equipment Labels, Symbols, Warning Statements and Abbreviations	19
	1.8	Electromagnetic immunity	20
	1.9 Comr	Recommended Separation Distance between Portable and Mobile RF munications Equipment and Q-Sense	74
	1.10		
2	_	Overview	
_	ں 2.1	Training	
	2.1	System Description	
	2.2 2.3	Q-SENSE Thermode	
	_	Q-SENSE Thermode	
	2.4 2.5	System Status	
	2.5 2.6	-	
	2.0	System States Application Methods	
	2.7		
	2.7		
	2.7		
	2.7		
	2.7		
	2.7		
3		System Components	
-	3.1	Power side Panel	
	3.2	Connectors side panel	
4	-	Setup and Installation	
-	د 4.1	Computer Requirements	
	T. I		30

Advanced medical systems

4.2	Ins	stallation USB Content	37
4.3	Set	tup and Installation Main Steps	37
4.4	Se	lecting a Location	37
4.5	De	vice Connections	38
4.6	So	ftware Installation	39
4.7	Un	it Protection	40
4.8	Pro	oduct Activation	40
5 C	Ope	ration	42
5.1	Po	wer up Q-SENSE System	42
5.2	Sta	arting Q-SENSE	42
5.3	Sys	stem Self-Test	44
5.4	Ba	sic Operation	45
5.4	.1	Menu Bar	46
5.4	.2	Toolbar	47
5.4	.3	Search Row	47
5.4	.4	Current Selection Information	48
5.4	.5	Data Display Area	48
5.4	.6	Q-Sense Status Line	48
6 7	Test	t Procedure and Management	49
		-	
6.1	Se	lecting a Patient	49
6.1	Se	lecting a Patient	50
6.1 6.2	Sel Sel	lecting a Patient	50 52
6.1 6.2 6.3	Sel Sel Ru	lecting a Patient lecting a Program lecting a Body Site	50 52 53
 6.1 6.2 6.3 6.4 6.4 	Sel Sel Ru	lecting a Patient lecting a Program lecting a Body Site nning the Test	50 52 53 55
 6.1 6.2 6.3 6.4 6.4 	Sel Sel Ru .1	lecting a Patient lecting a Program lecting a Body Site nning the Test Test and Post-Test Options	50 52 53 55 56
 6.1 6.2 6.3 6.4 6.4 6.4 6.4 	Sel Sel Ru .1 .2 .3	lecting a Patient lecting a Program lecting a Body Site nning the Test Test and Post-Test Options Temperature Graph Display.	50 52 53 55 56 56
 6.1 6.2 6.3 6.4 6.4 6.4 6.4 	Sel Sel Ru .1 .2 .3 Prog	lecting a Patient lecting a Program lecting a Body Site. nning the Test Test and Post-Test Options. Temperature Graph Display. Test Run Keyboard Shortcuts (Hot Keys)	50 52 53 55 56 56 58
 6.1 6.2 6.3 6.4 6.4 6.4 6.4 7 	Sel Sel Ru .1 .2 .3 Prog Prog	lecting a Patient lecting a Program lecting a Body Site. Inning the Test Test and Post-Test Options. Temperature Graph Display. Test Run Keyboard Shortcuts (Hot Keys)	50 52 55 56 56 58 58
6.1 6.2 6.3 6.4 6.4 6.4 6.4 7 7 7	Sel Ru .1 .2 .3 Prog Cre	lecting a Patient lecting a Program lecting a Body Site nning the Test Test and Post-Test Options Temperature Graph Display Test Run Keyboard Shortcuts (Hot Keys) gram Management	 50 53 55 56 58 58 60
6.1 6.2 6.3 6.4 6.4 6.4 6.4 7 7 7 7 7 7 7	Sel Ru .1 .2 .3 Prog Cre Re	lecting a Patient lecting a Program	 50 52 53 55 56 58 58 60 61
6.1 6.2 6.3 6.4 6.4 6.4 6.4 7 7 7 7 7.1 7.2 7.3	Sel Ru .1 .2 .3 Prog Cre Rei Du	lecting a Patient	 50 52 53 55 56 58 60 61 61
6.1 6.2 6.3 6.4 6.4 6.4 6.4 7 7 7 7 7 7.1 7.2 7.3 7.4	Sel Sel Ru .1 .2 .3 Proc Cre Rel Du Edi	lecting a Patient lecting a Program	 50 52 53 56 58 60 61 61 61
6.1 6.2 6.3 6.4 6.4 6.4 7 7 7 7 7 7 7.1 7.2 7.3 7.4 7.5	Sel Sel Ru .1 .2 .3 Prog Cre Rel Du Edi Pro	lecting a Patient lecting a Program lecting a Body Site Inning the Test	50 52 53 55 56 58 60 61 61 61 61
6.1 6.2 6.3 6.4 6.4 6.4 7 7 7 7 7 7 7 7 7 3 7.4 7.5 7.6	Sel Ru .1 .2 .3 Prog Rei Du Edi Pro	lecting a Patient lecting a Program lecting a Body Site nning the Test Test and Post-Test Options Temperature Graph Display Test Run Keyboard Shortcuts (Hot Keys) Gram Management ogram List bgram List plicate A Program ogram Details	 50 52 53 56 56 58 60 61 61 61 61 62
6.1 6.2 6.3 6.4 6.4 6.4 7 7 7.1 7.2 7.3 7.4 7.5 7.6 7.6	Sel Ru .1 .2 .3 Prog Rei Du Edi Pro	lecting a Patient lecting a Program lecting a Body Site Inning the Test	 50 52 53 56 58 60 61 61 61 62 66
6.1 6.2 6.3 6.4 6.4 6.4 7 7 7 7 7 7 7 7 7 3 7.4 7.5 7.6 7.6 7.6 7.6	Sel Ru .1 .2 .3 Prog Cre Rei Du Edi Pro	lecting a Patient	 50 52 53 56 58 58 60 61 61 61 62 66 70



Q-Sense Technical Reference Manual		
	7.6.6	Chaining Programs
	7.6.7	Next Sequence Definition
7	.7 9	Standard Events
	7.7.1	Standard Event Configuration 80
	7.7.2	List of Standard Events according to Program Type
7	.8 1	۲est Instructions81
7	.9 1	Import / Export Program81
8	Pa	ntient Management
8	.1 6	Patients List
8	.2	Add New Patient85
8	.3 F	Remove Patient
8	.4 E	Edit Patient Details86
8	.5 [Department Management86
8	.6 1	ICD 9 Management
8	.7]	Import / Export Patient Lists88
9	Re	esults Management
9	.1 P	Results List90
9	.2 1	Import / Export Results
9	.3 F	Print Results and Reports95
	9.3.1	Analysis Report Dialog
10	Sy	stem Configuration
1	0.1	Software Settings105
	10.1.	1 Communication
	10.1.	2 Patient Fields106
	10.1.	.3 Results Fields
	10.1.	4 Report Options110
	10.1.	5 User Management
1	0.2	Hardware Settings
	10.2.	1 General
	10.2.	.2 Safety
	10.2.	3 Thermode124
1	0.3	Test Settings
	10.3.	5
	10.3.	
	10.3.	
11	Sy	rstem Information
12	Th	ne Black box (Logger)135
1	2.1	Exporting the Black Box File135



13	App	endix A – Normative Data136
13	5.1 N	Normative Data License Agreement136
13	5.2 E	nable Normative Data137
13	.3	
13	8.4 N	Normative Data Editor139
:	13.4.1	Selecting the Normative Data Source (Built-In Tables)139
-	13.4.2	Normative Data Tables140
13	3.5 C	Display of Normative Data in Tests, Results and Reports
14	Арр	endix B – Backing Up and Restoring the Database
15	App	endix C - Accessories145
16	App	endix D – System and Module Description148
16	5.1 S	System Description148
17	App	endix E – Preventive and Periodical Maintenance
17	'.1 F	Periodical System Run and Visual Check149
17	'.2 F	Periodical Thermode Calibration150
17	'.3 F	Periodical Thermode Maintenance150
18	App	endix F – Cleaning and Disinfecting151
19	App	endix G - Thermode straps assembly instructions
20	App	endix H – List of Accompanying Documents
21	App	endix I – Transportation and Storage conditions
22	Арр	endix J – Environmental Conditions155
23		endix K – Upgrading procedure156
24	••	endix L – Troubleshooting – Operator Level
24		Numbered Software Error and Warning Messages
-	24.1.1	Frequent Warning Messages
-	24.1.2	Frequent Error Messages162
24	.2 F	aulty System Responses171
25	Арр	endix M – Licensed Options
25	5.1 F	TP Support174
-	25.1.1	Export Configuration174
25	5.2 0	CoVAS
-	25.2.1	CoVAS Calibration
25	5.3 F	Patient Record Control177
2	25.3.1	Control Patient Field Entry Format with Input Masks177
-	25.3.2	Add/Remove Patient Fields
25	5.4 F	Pain Rating
-	25.4.1	Pain Rating Parameters Configuration



26 Ap	pendix N – fMRI functionality	
26.1	fMRI Thermode	
26.1.	1 Q-sense system configuration in fMRI mode:	
26.1.	2 Installation –MR Filter	
26.2	TTL functionality	
26.2.	1 Q-sense fMRI side panel	
26.2.	2 TTL set up	
26.2.	3 Standard Event Configuration	
26.2.	4 TTL specifications	
26.3	Blower control	



List of Figures

Figure 1: Q-Sense Label
Figure 2: Ramp &Hold Sequence – Return to Baseline
Figure 3: Ramp & Hold Sequence – Return to Next Destination
Figure 4: VAS Search – Manual Mode 31
Figure 5: Power side panel
Figure 6: Connectors side panel
Figure 7: Q-SENSE System Wiring Schema
Figure 8: Connecting Wall hanger to the wall
Figure 9: Q-SENSE System Wiring Schema 38
Figure 10: DC power port
Figure 11: Home screen product registration link 41
Figure 12: Q-sense COM port 42
Figure 13: Home Screen
Figure 14: Login Screen
Figure 15: Q-Sense Splash Screen 43
Figure 16: Self-Test Prompt 44
Figure 17: Q-Sense in Rest Mode 45
Figure 18: Test Navigator – Main Menu 45
Figure 19: Test Management Main Screen 49
Figure 20: Patient Selection Menu 50
Figure 21: Program Selection Menu 51
Figure 22: Select Body Site screen 52
Figure 23: Select Body Site – Specific site selection
Figure 24: Pre-Test Screen
Figure 25: Real-Time Test
Figure 26: Program Management screen 58
Figure 27: Program Management – Program Preview
Figure 28: Program graph preview 60
Figure 29: New Program Screen 60
Figure 30: Program Menu – Limits
Figure 31: A Limits Method Test
Figure 32: Statistical analysis of results
Figure 33: Program Menu – Levels 66
Figure 34: A Levels Method Test
Figure 35: Statistical Analysis of the Results 70
Figure 36: Program Menu – TSL 70
Figure 37: A TSL Method Test



Q-Sense Technical Reference Manual	
Figure 38: Statistical Analysis of the Results7	'3
Figure 39: Program Menu – Ramp and Hold7	'3
Figure 40: Program Editor – VAS Search 7	'5
Figure 41: Program Chain Editor7	7
Figure 42: Program Editor Next Sequence Definition7	'9
Figure 43: Manual Next Sequence definition7	'9
Figure 44: Standard Events in Program Editor	30
Figure 45: Test Instructions Editor	31
Figure 46: Patients List	33
Figure 47: Patient Management – Patient details preview	34
Figure 48: Patients Editor Screen	35
Figure 49: Edit Department Menu 8	37
Figure 50: Edit ICD Menu	38
Figure 51: Results List	90
Figure 52: Result preview display)1
Figure 53: Export Result File Opened In Excel – Description Tab Sample)3
Figure 54: Export Result File Opened In Excel – Data Tab Sample)3
Figure 55: Analysis Reports Dialog	95
Figure 56: Analysis Report	96
Figure 57: Single Results Report	98
Figure 58: Multiple Results Report	98
Figure 59: Site to Site Results Report10)0
Figure 60: Pre / Post Results Report10)0
Figure 61: Trend Results Report10)0
Figure 62: Custom Report Results10)0
Figure 63: Bilateral Results Report10)1
Figure 64: Custom Report Configuration10)3
Figure 65: Custom Report Configuration10)4
Figure 66: Custom Report Configuration10)4
Figure 67: Software Settings Dialog10)5
Figure 68: Open Software Settings10)6
Figure 69: Software Settings - Patient Fields tab selection10)6
Figure 70: Software Settings - Patient Fields10)7
Figure 71: Results Fields Configuration Screen11	.0
Figure 72: Report Options tab11	.1
Figure 73: Detailed Report example11	.2
Figure 74: Detailed Report Options11	.3
Figure 75: Custom made Report Name dialog11	.4
Figure 76: Report Editor Screen11	.5
Proprietary 9 of 18	38



Figure 77: Open Software Settings	116
Figure 78: Software settings users tab selection	116
Figure 79: Software Settings Users Tab	116
Figure 80: Software Settings Add New User	117
Figure 81: Software settings filled	118
Figure 82: Hardware settings – General tab	123
Figure 83: Hardware settings – Safety tab	124
Figure 84: Hardware settings – Thermode tab	125
Figure 85: Test Configuration Settings	126
Figure 86: Instructions Configuration	129
Figure 87: Test Instructions Editor	130
Figure 88: Audio save button	130
Figure 89: Body Sites Editor	131
Figure 90: System Information	133
Figure 91: Sysinfo File Sample	134
Figure 92: Black Box – Software	135
Figure 93: Normative Data Editor - Enable Normative Data	137
Figure 94: Normative Data License Agreement	138
Figure 95: Normative Data Editor - Learn More	139
Figure 96: Normative Data screen select source	140
Figure 97: Normative Data Editor	141
Figure 98: Database Utility – Backup/Restore Menu	144
Figure 99: System Diagram	148
Figure 100: Previous version message	156
Figure 101: Removing previos version screen	156
Figure 102: Data Backup Path	
Figure 103: MMS Uninstall	
Figure 104: Uninstall Completed	
Figure 105: MMS successfully removed	157
Figure 106: Error and Warning Messages Sample	158
Figure 107: Open Software Settings	174
Figure 108: Software settings export tab selection	174
Figure 109: Software Settings Export Tab	175
Figure 110: CoVAS Calibration Tab	176
Figure 111: Patient Fields with Input Masks	
Figure 112: Patient Card with Selected Patient Fields	179
Figure 113: Pain Rating Parameters	
Figure 114: Pain Rating Limits	
Figure 115: fMRI room setup	184
Proprietary 1() of 188



Figure 116:	Q-sense fMRI Thermode Control Room part WA 002811	.84
Figure 117:	Q-sense fMRI Thermode Shielded room part AS 004121	.85
Figure 118:	Q-sense fMRI side panel1	.86
Figure 119:	HW settings for Q-sense fMRI1	.86
Figure 120:	Standard Events in Program Editor1	.87
Figure 121:	Standard Event Editor1	.87



List of Tables

Table 1: Equipment Labels 19
Table 2: Q-SENSE Specifications 25
Table 3: Terminology 27
Table 4: Q-SENSE Connectors side Panel receptacles 33
Table 5: Connectors to power side panel
Table 6: Menu Bar options 46
Table 7: Status line parameters 48
Table 8: Test Run options 55
Table 9: Hot Keys 56
Table 10: Probe Type options
Table 11: Program Details 63
Table 12: Program Parameters 64
Table 13: Program Details 67
Table 14: Program Parameters 67
Table 15: Program Details 71
Table 16: Program Parameters 71
Table 17: Program Parameters 74
Table 18: Program Parameters 75
Table 19: Chain sub program parameters 77
Table 20: Program Details 77
Table 21: Program Table 78
Table 22: List of standard events according to program type80
Table 23: Report Analysis Configuration
Table 24: Report Analysis Configuration101
Table 25: Software Settings - General106
Table 26: Patient's Field 110
Table 27: Report Options
Table 28: Report Options
Table 29: Authorization Levels (main actions) 119
Table 30: Hardware Settings Tab Sheets 123
Table 31: Hardware Settings – General tab options 123
Table 32: Pre-test parameters 126
Table 33: Test configuration parameters 126
Table 34: Post-test parameters 127
Table 35: Test Display configuration
Table 36: Temperature Scale Configuration 127
Table 37: Temperature Limits 128
Table 38: Normative Data Fields 141
Proprietary 12 of 188

Table 39: Accessories	145
Table 40: Q-sense Accompanying Documents	153
Table 41: Warnings	158
Table 42: Errors	162
Table 43: Faulty System Responses	171
Table 44: Regular Characters	177
Table 45: Special Characters	177



Disclaimer

Medoc assumes no liability for use of this document if any unauthorized changes to the content or format have been made. Every care has been taken to ensure the accuracy of the information in this document. However, Medoc assumes no responsibility or liability for errors, inaccuracies, or omissions that may appear in this document. Medoc reserves the right to change the product without further notice to improve reliability, function or design. This manual is provided without warranty of any kind, implied or expressed, including, but not limited to, the implied warranties of merchantability and fitness for a particular purpose.

1 Safety Guidelines and Regulations

This manual is written for trained users of Medoc Products. The user includes the body with authority over the equipment and those persons who actually handle the equipment.

Before attempting to work with this equipment, read, understand, note and strictly observe all Warning notices, Cautions and Safety markings on the equipment. The manual and software are available in English.

Before attempting to work with this equipment, make sure that this manual and any Release Notes delivered with the software media pack have been thoroughly read and fully understood, paying particular attention to all:

- 1. Warnings
- 2. Cautions
- 3. Notes
- 4. Important Notices
- 5. User Notices

1.1 User Manual Icons

The following icons are used throughout the user manual:



	Warning: A condition that could cause serious injury or death to a patient and/or operator if instructions are not followed.
	Caution: A condition that could cause possible damage to equipment or cause the system to function inaccurately.
\bigcirc	Note: Indicates important user information regarding the use of the system.
	Advice: Refer to instruction manual/ booklet
	INSTRUCTION: Indicates an instruction where it is important to follow the user manual literally as described.

1.2 Intended use

The Q-Sense is a device for quantitative thermo-testing in the context of quantitative sensory testing battery in compliance with Quantitative Sensory Testing (QST) protocols to detect and quantify sensory loss and sensory gain aimed to precisely characterize somatosensory function in patients.

1.3 Target Population and Contraindications

The Q-Sense can be used on both healthy subjects and patients suffering from wide range of neuropathic, CNS and pain conditions. Minimum age reported in the literature is 6.

The only known contraindication is related to the skin on which the thermode is applied – it must be used in contact with intact skin only. The thermode also intended to be applied on the skin only, not on the eyes and the mouth.

1.4 Safety and Regulatory Summary

Read and follow all WARNINGS, CAUTIONS and NOTES provided in this manual. To avoid the possibility of injury, damage to your system, or loss of data, always follow these precautions during system operation.

- The Q-SENSE system complies with safety requirements for medical electrical systems (based on the IEC 60601-1 standard).
- The Q-SENSE system complies with electromagnetic emission levels (based on table 201 in the IEC 60601-1-2 standard).
- The Q-SENSE system complies with electromagnetic immunity levels (based on tables 202 and 204 in the IEC 60601-1-2 standard).
- This device complies with 93/42/EEC MDD.
- It is recommended to keep a distance of 3 meters between portable and mobile RF communications equipment and the Q-SENSE (based on table 206 in the IEC 60601-1-2 standard).



1.5 Safety Requirements



The Q-SENSE system can be tested according to IEC 62353 Recurrent test and test after repair of medical electrical equipment.



Do not modify or replace any component of the Q-SENSE system. Connecting or replacing external Q-SENSE accessories is allowed.

No modification of this equipment is allowed.

To avoid the risk of electric shock, this equipment must only be connected to a supply mains with protective earth



Keep all liquids away from the Q-SENSE system. Unplug the Q-SENSE system if it is not to be used for a long period of time.

Do not block airflow anywhere around the Q-SENSE system.

1.5.1 Warnings

- Only personnel properly trained to operate the Q-SENSE system should use this system.
- Do not turn on system power until all cables have been properly connected and verified.
- To avoid the risk of electric shock, this equipment must only be connected to a supply mains with protective earth
- Do not use any power supply which not supplied by Medoc (Medoc's power supply Cat.# DT 00017).
- Do not use any electrode paste, gel, or other materials on the contact point between the Thermode plate and the skin of the tested subject.
- To reduce the risk of injury, place the Thermode on the subject only before starting the test and while the test screen is visible. After the test is complete, remove the Thermode from the subject's skin while the test screen is still visible.
- Connect the Thermode to the patient's skin ONLY during the test; not during system Self-Test, programming or maintenance.
- The computer that is used to operate the Q-SENSE system must be powered through a Medical Grade Isolation Transformer only.
- The use of accessories or cables other than those specified, with the exception of accessories or cables sold by the manufacturer as replacement parts, may result in increased emissions or decreased electrical immunity of the device.
- Connecting any device or accessory that has no medical grade certificate to the Q-SENSE system is not allowed.
- Using a Thermode without the appropriate calibration table may result in potential harm or injury.
- Using the Q-SENSE system not according to instructions may result in potential harm or injury.



- Adverse Reaction: Skin irritation (in addition to pain sensation) beneath the probe has been reported with the use of a stimulator, which was based on similar technology as the Q-Sense device.
- Be aware of potential risk of skin damage caused by wrong parameter combination.
- Use of this equipment adjacent to or stacked with other equipment should be avoided because it could result in improper operation. If such use is necessary, this equipment and the other equipment should be observed to verify that they are operating normally.
- of accessories, transducers and cables other than those specified or provided by the manufacturer of this equipment could result in increased electromagnetic emissions or decreased electromagnetic immunity of this equipment and result in improper operation.
- Portable RF communications equipment (including peripherals such as antenna cables and external antennas) should be used no closer than 30 cm (12 inches) to any part of the [ME EQUIPMENT or ME SYSTEM], including cables specified by the manufacturer. Otherwise, degradation of the performance of this equipment could result.
- NOTE The EMISSIONS characteristics of this equipment make it suitable for use in industrial areas and hospitals (CISPR 11 class A). If it is used in a residential environment (for which CISPR 11 class B is normally required) this equipment might not offer adequate protection to radio-frequency communication services. The user might need to take mitigation measures, such as relocating or reorienting the equipment
- •

1.5.2 Cautions

- Proper use of this device depends on careful reading of all instructions and labels.
- Turn OFF system power before connecting or disconnecting any system component(s) or accessories. Otherwise, you may damage the device(s).
- The Thermode is very delicate and can easily be damaged. Therefore, handle with care.
- If you disconnect any cables, take care to reconnect them correctly to prevent damage to the system or components.
- Inspect the power cord often for fraying or other damage. DO NOT operate the apparatus if the power cord or plug is damaged.
- The computer that is used to operate the Q-SENSE system must not be connected to a network while it is used for running tests.
- The Q-SENSE system does not require special precautions regarding EMC, and needs to be installed and put into service according to this manual.
- Portable and mobile RF communications equipment can affect medical electrical equipment.
- Use caution when using the Q-SENSE Thermode on patients with suspected neuropathies as they may be more susceptible to soft tissue or nerve damage at extreme temperatures. Also, patients with neuropathies may not be able to properly discontinue use of the device during prolonged hot or cold stimulation.



1.5.3 Equipment Classification

- Degree of protection against electric shock: Class I
- Type of protection against electric shock: BF
- Type of Operation: Continuous
- Protection against ingress of liquids: Not protected against ingress of liquids
- Ordinary equipment.
- Computer must comply with IEC 950 EN 60950 UL 60950.

1.6 System Protection

1.6.1 System Self-Test

Upon start-up the system performs a self-test in which system sensors are being tested. If a malfunction is detected, an appropriate message is displayed and the system cannot operate until that malfunction is resolved.

1.6.2 Temperature Safety Mechanisms

Several safeguard mechanisms have been implemented in the system to safeguard against extreme temperatures and to protect the tested subject and the unit.

Software protection mechanisms include:

- Temperature upper and lower limits In normal operation, Thermode temperature will always be within these limits.
- Time duration limits Thermode temperature is limited in duration. If the Thermode temperature maintains a specific temperature (or above) for a longer period of time then specified for that temperature, the system will go into Safe Mode.
- Safe Mode a protective state of the system in which it is not possible to run tests. In any case of suspected malfunction or if any modification is made to system hardware settings, system will remain in safe mode until a system Self-Test is performed.

Hardware protection mechanisms include:

• If the Thermode temperature reaches 57°C an analog circuit overrides the system and lowers the temperature gradually.

1.6.3 Thermode Detection

The system automatically detects that a Thermode is missing, and disables it in order to protect both system and user.



1.7 Equipment Labels, Symbols, Warning Statements and Abbreviations



Figure 1: Q-Sense Label

Table 1: Equipment Labels

Equipment Label	Description
I	Power switch ON/OFF
0	
СОМ	Communications connector.
\sim	Date of Manufacture (YYYY-MM)
	Manufacturer
†	Degree of protection against electric shock – Applied Part Type BF.
\triangle	Warning - Connect the power cord to the power outlet, according to the local electrical standards.
Ĩ	Refer to Manual
	Disposal according to electronic scrap ordinance





Manual edition refers to current version of manufactured system



Medoc reserves the right to change specifications without prior notice, in line with the company policy of constant product improvement

1.8 Electromagnetic immunity

The Q-Sense is intended for use in the electromagnetic environment specified below. The customer or the user of the Q-Sense should assure that it is used in such an environment.

Immunity test	IEC 60601 test level	Compliance level	Electromagnetic environment –guidance
Electrostatic discharge (ESD) IEC 61000-4-2	±6 kV contact ±8 kV air	±6 kV contact ±8 kV air	Floors should be wood, concrete or ceramic tile. If floors are covered with synthetic material, the relative humidity should be at least 30 %.
Power frequency (50/60 Hz) magnetic field IEC 61000-4-4	3 A/m 50 & 60 Hz	3 A/m 50 & 60 Hz	Power frequency magnetic fields should be at levels characteristic of a typical location in a typical commercial or hospital environment.
Electrical fast transient/burst IEC 61000-4-4	±2 kV for power supply lines	±2 kV for power supply lines	Mains power quality should be that of a typical commercial or hospital environment.
Surge IEC 61000-4-5	±1 kV line(s) and neutral	±1 kV line(s) and neutral	Mains power quality should be that of a typical commercial or hospital environment.
Voltage dips, short interruptions and voltage variations on power supply input lines IEC 61000-4-11	<5 % UT (>95 % dip in UT) for 0,5 cycle 40 % UT (60 % dip in UT) for 5 cycles 70 % UT (30 % dip in UT) for 25 cycles <5 % UT	<5 % UT (>95 % dip in UT) for 0,5 cycle 40 % UT (60 % dip in UT) for 5 cycles 70 % UT (30 % dip in UT) for 25 cycles) <5 % UT	Mains power quality should be that of a typical commercial or hospital environment. If a dip or an interruption of mains power occurs, the current of the Q- sense may be dropped off from normal level, it may be necessary to use uninterruptible power supply or a battery.



(>95 % dip in UT)	(>95 % dip in UT) for 5s		
for 5s			



Immunity test	IEC 60601	Compliance	Electromagnetic environment –
	test level	level	guidance
			Portable and mobile RF communications equipment should be used no closer to any part of the Q-sense, including cables, than the recommended separation distance calculated from the equation applicable to the frequency of the transmitter.
			Recommended separation distance
			$d = \left[\frac{3.5}{V_1}\right]\sqrt{P}$
Conducted RF IEC 61000-4-6	3V, 6V	3Vrms, 6V	$d = \left[\frac{12}{V_2}\right]\sqrt{P}$
			$d = [\frac{12}{E_1}]\sqrt{P}$ 80 MHz to 800 MHz
Radiated RF IEC 61000-4-3	3V/m	3V/m	$d = \left[\frac{23}{E_1}\right]\sqrt{P} 800 \text{ MHz to 2,5 GHz}$ is the maximum output power rating of the transmitter in watts (W) according to the transmitter manufacturer and d is the
			recommended separation
	3V from 0.15 to	3V from 0.15 to	distance in metres (m). Field strengths from fixed RF transmitters, as determined by an electromagnetic site survey,
	80MHz;	80MHz;	should be less than the compliance level in each frequency range.
	6V from 0.15 to	6V from 0.15 to	D Interference may occur in the vicinity of equipment marked with the following
	80MHz and 80% AM	80MHz and 80% AM	symbol:
	at 1kHz	at 1kHz	
	10V/m from 80MHz	10V/m from 80MHz	
	to 2.7GHz	to 2.7GHz	
A A		Proprietar	y 22 of 188



NOTE 1 At 80 MHz and 800 MHz, the higher frequency range applies.

NOTE 2 These guidelines may not apply in all situations. Electromagnetic propagation is

affected by absorption and reflection from structures, objects and people.

* Field strengths from fixed transmitters, such as base stations for radio (cellular/cordless) telephones and land mobile radios, amateur radio, AM and FM radio broadcast and TV broadcast cannot be predicted theoretically with accuracy. To assess the electromagnetic environment due to fixed RF transmitters, an electromagnetic site survey should be considered. If the measured field strength in the location in which the Q-Sense is used exceeds the applicable RF compliance level above, the Q-Sense should be observed to verify normal operation. If abnormal performance is observed, additional measures may be necessary, such as reorienting or relocating the Q-Sense.

** Over the frequency range 150 kHz to 80 MHz, field strengths should be less than 3 V/m.

Declaration – electromagnetic emissions			
Emissions test	Compliance	Electromagnetic environment – guidance	
RF emissions CISPR 11	Group1 Class A	The Q-sense uses RF energy only for its internal function. Therefore, its RF emissions are very low and are not likely to cause any interference in nearby electronic equipment.	
Harmonic emissions IEC 61000-3-2	Class A	The Q-sense is suitable for use in all establishments other than domestic, and may be used in domestic establishments and those	
Voltage Fluctuations And Flicker IEC 61000-3-3:2013	Complies	directly connected to the public low-voltage power supply network that supplies buildings used for domestic purposes, provided the following warning is heeded: Warning: This equipment/system is intended for use by healthcare professionals only. This equipment/ system may cause radio interference or may disrupt the operation of nearby equipment. It may be necessary to take mitigation measures, such as re-orienting or relocating the Q-sense or shielding the location.	



Medoc

1.9 Recommended Separation Distance between Portable and Mobile RF Communications Equipment and Q-Sense

The Q-Sense is intended for use in an electromagnetic environment in which radiated RF

Rated maximum	Separation distance according to frequency of transmitter (m)				
output power of transmitter	150 kHz to 80 MHz outside ISM	150 kHz to 80 MHz in ISM bands	80 MHz to 800 MHz	800 MHz to 2,5 GHz $d = \left[\frac{23}{E_1}\right]\sqrt{P}$	
(W)	bands $d = [\frac{3,5}{V_1}]\sqrt{P}$	$d = \left[\frac{12}{V_2}\right]\sqrt{P}$	$d = [\frac{12}{E_1}]\sqrt{P}$	E1	
0.01	0.12	0.2	0.4	1	
0.1	0.37	0.64	1.3	2.6	
1	1.17	2	4	8	
10	3.7	6.4	13	26	
100	11.7	20	40	80	

For transmitters rated at a maximum output power not listed above, the recommended separation distance d in metres (m) can be estimated using the equation applicable to the frequency of the transmitter, where P is the maximum output power rating of the transmitter in watts (W) according to the transmitter manufacturer.

NOTE 1: At 80 MHz and 800 MHz, the separation distance for the higher frequency range applies.

NOTE 2: These guidelines may not apply in all situations. Electromagnetic propagation is affected by absorption and reflection from structures, objects and people.

disturbances are controlled. The customer or the user of the Q-Sense can help prevent

electromagnetic interference by maintaining a minimum distance between portable and mobile

RF communications equipment (transmitters) and the Q-Sense as recommended below,

according to the maximum output power of the communications equipment.



1.10Technical Data



Medoc reserves the right to change specifications without prior notice, in line with the company policy of constant product improvement

The table below describes the technical specifications and capabilities of the Q-Sense system.

Parameter	Specification	
Thermode Active Area	30x30 mm	
Thermode temperature Range	20 – 50 °C	
Heating Rate	0 - 2 °C/Sec ± 10%	
Cooling Rate	0 - 1 °C/Sec ± 10%	
Stimulation Methods	Limits, Levels & TSL Ramp & Hold (optional)	
Set-Point Resolution	0.1 °C	
Display Resolution	0.1 °C	
Repeatability	± 0.3 °C	
Absolute Accuracy	±0.3 °C	
Operating Mode	Intervals* *(Short pause between patients may be required)	
Minimum Inter Stimuli Interval (ISI)	2 sec	
Maximum Consecutive Stimuli	10	
Baseline Thermode Temperature Range	28 – 40 °C	
Programmable Parameters	Method, Adaptation Temperature, Heating Rate, Cooling Rate, Number of Stimuli, ISI, Sound Option, Randomize Option	
Communication with PC	USB	
Communication with Printer	Via PC	
Standards	IEC 60601-1; IEC 60601-1-2	
Approvals	CE, FDA	
Operating Voltage	12 VDC, 5A	
Power Supply	100 – 240 VAC, 4 – 2 A, 50/60 Hz	
Power Consumption	Approx. 60W	
Dimensions	31cm X 28 cm X 16cm (table or wall mount) Shipping: 53cm X 34cm X 22cm	
Weight	Approx. 4.5 Kg Shipping: Approx. 10 Kg	
Software	Propriety Software Application (Included)	
Computer Requirements (PC not included)	Notebook PC (Atom CPU not supported) At least 2 GB RAM, 6 GB Free disc space. Windows 7/8 (32 & 64 bit), Windows XP (32bit) sp. 3 and above.	

Table 2: Q-SENSE Specifications



Parameter	Specification
Screen Resolution	1366X768 and higher
Patient Response Unit	2 buttons (Yes / No) Response Unit
Product life time	10 years
Thermode life time	1 year



2 Overview

The Q-Sense Thermal Sensory Analyzer is a QST (Quantitative Sensory Testing) device including advanced software package designed for clinical use and research in the field of Pain Management.

2.1 Training

Users of Q-Sense shall have received adequate training on its safe and effective use before attempting to work with it. Training requirements may vary from country to country. The User shall make sure that training is received in accordance with local laws or regulations that have the force of law. Information on training is available from your local Medoc representative.

2.2 System Description

The Q-Sense is a TSA 2001 model, which enables the user to perform various thermal test paradigms including the method of Limits, Levels, and TSL. These test paradigms can be utilized for a wide range of thermal QST pain measures such as thermal pain threshold and tolerance.

Intended use: The Q-Sense is a pain management and pain research intended to be used for the quantitative assessment of small nerve fiber dysfunctions. It measures sensory thresholds such as the cold and warm sensation, and heat-induced pain.

2.3 Q-SENSE Thermode

The Thermode is the active component of the system which delivers the actual stimulus. The main components of the Q-Sense Thermode include:

- Active Thermal Element The active Thermal element is a Thermoelectric Cooler (TEC), a peltier element, which generates a temperature gradient according to the direction and amplitude of power (electrical current) it is given.
- Temperature Sensors (Thermistors) The Thermistors are resistors which change their resistance with temperature. Two thermistors are used to sense the slightest change in temperature and deliver feedback to the temperature control unit. The temperature control unit adjusts the power output to this feedback.
- Contact Plate Contact plate is the external area of the Thermode which comes in contact with the skin of the tested subject. The plate is aluminum coated for heat dispersion.
- Cooling Elements The cooling elements disperse heat from the TEC

For more detailed specifications of the Q-Sense please refer to the Technical Specifications in section 1.10, page 25

2.4 Terminology

The following terms are used throughout this manual:

Table 3: Terminology

Description
The probe which is used to apply the thermal stimulation on the subject's skin.
Thermo Electric Cooler (TEC), a Peltier element used as the active heat generator in the Thermode.
The initial/rest temperature of the Thermode, after system initialization and in between tests. Also referred to as Adaptation Temperature.
-



Term	Description
Baseline Temperature	The initial temperature the stimulus starts from
Destination Temperature	The final temperature the stimulus ends with.
Duration Time	Indicates the time the probe will remain at the destination
Trial	Represents one stimulus
Sequence	A set of trials with the same specifications
Test	A set of sequences.
Event	A specific occurrence during a test such as the onset of a stimulus, reaching a specific temperature and etc.
LUT	A Look Up Table (LUT) maps one set of values to another. A LUT is used to map between the readings from the temperature sensors to the actual temperature it represents.
Firmware	The program embedded on the internal processor of the system

2.5 System Status

- **Online:** This is the Q-SENSE system full operating mode and indicates full communication between MEDOC Main Station and Q-Sense System. While on-line the Q-Sense system can operate in several different states as listed in the following section.
- **Demo Mode:** This is a demo mode of operation. There is no communication between the software and the Q-SENSE system. All tests performed while working in this mode are simulated. No actual thermal stimulation is performed and there is no access to the Q-SENSE hardware settings. The Simulator is useful for demonstration, training and for protocol design without executing actual thermal stimulation. It is recommended to turn off the Q-SENSE system while working in Simulator mode.

2.6 System States

System State is the active state in which the system is currently operating. System states refer to the **online** status only.

The Q-SENSE active states are as follows:

- **Rest Mode**: The Q-SENSE system is active and ready to perform a test. The temperature is maintained at the specified Rest Mode temperature. In this mode, patient, program and result management are possible.
- **Test Run**: The Q-SENSE system is running a (predefined) test. The Thermode is active, and the temperature varies according to the active program.
- **Safe Mode**: No test can be performed and safety mechanisms are activated. The system must perform a Self-Test before resuming normal operation.
- **Engineering**: During Thermode Calibration the system state is set to Engineering. In this state safety limitations are lowered and it is not allowed to run any actual tests.
- **Test Initiation**: The system raises the temperature from Rest Mode temperature to Baseline level. A stabilization test is then performed, to make sure that the temperature is maintained according to the program requirements.
- **Self-Test**: The system performs a Self-Test in order to check whether the connected Thermode is functioning according to system requirements.



• **Black Box**: The system has detected an error and is currently logging data in the internal memory. The log file can be exported later for service use.

2.7 Application Methods

Method is the type of the test program. Q-SENSE supports the following methods:

- **Limits**: A set of predetermined stimuli is emitted, and for each stimulus, the tested subject is required to respond (stop the stimulus) when perceiving a predefined sensation. For more details refer to section 2.7.1.
- **Levels**: A set of stimuli is emitted, and for each stimulus, the subject is required to respond. The following stimulus is calculated according to the subject's response. For more details, refer to section 2.7.2.
- **Thermal Sensory Limen (TSL)**: The temperature direction changes according to the subject's response, alternating between increasing temperature and decreasing temperature. For more details, refer to section 2.7.3.
- **Ramp & Hold** (optional, available with specific license): The temperature of the stimulus is maintained for a predetermined duration.
- **VAS Search** (optional, available with specific license): A set of Ramp & Hold stimuli is emitted, and for each stimulus, the subject is required to report the perceived level of pain. The following stimulus is calculated according to the subject's response.



Before using each of these methods, make sure you have read section 6, Page 49, Test Procedure and Management, which addresses the following topics: Tested subject selection, Program selection, and Stimuli administration.

2.7.1 Limits Method

This reaction time inclusive method consists of stimuli of continuously changing intensity. With each stimulus, when the subject perceives a predefined sensation, he/she manually stops the stimulus.

The Limits method is the most widely used method for threshold determination, because it requires the shortest test procedure of all methods, and it can measure not only pain thresholds, but also non-painful thermal thresholds.

For detailed information refer to section 7.6.1, page 62.

2.7.2 Levels Method

In the Levels method, stimuli are increased by a predetermined initial step until the first YES response. Stimuli then decreased by one half the initial step until a NO is given. Subsequently, the direction changes according to the response: increase for NO and decrease for YES. The step is halved at every direction change. The test is terminated when the step reaches a small enough size, as pre-determined by the user. Threshold is determined by taking the mean of last YES and the last NO. This is the shortest of all constant stimuli methods.

A modified Levels method allows you to hold Temperature for predefined Time at destination.

For detailed information refer to section 7.6.2, page 66.

2.7.3 Thermal Sensory Limen (TSL)

The TSL is another reaction time dependent method is the detection of **Thermal Sensory Limen (T.S.L.)**. In this algorithm, once started, the stimulus temperature does not stop at adaptation, rather, it continues directly from the warm threshold to the cold threshold and vice versa. After several oscillations between warm and cold sensory thresholds, the 'Limen' of no



thermal sensation is calculated as the difference between means of cold-sensation and warmsensation. Using the TSL method, allows the Limen or 'no thermal sensation' area to be calculated.

For detailed information refer to section 7.6.3, page 70.

2.7.4 Ramp & Hold Method

In this method, the temperature rises (or falls) to a predetermined destination, at a predetermined rate, remains there for a predetermined duration and proceeds to the next destination.

There are two main options: Return to Baseline, and Continue to Next Destination.

"Return to Baseline" option

The diagram below shows a general view of the sequence and predefined parameters. For the definition of each parameter, see 7.6.4, page 73.



Figure 2: Ramp & Hold Sequence – Return to Baseline

Each sequence consists of a number of similar trials (one trial is also allowed).

- The start point of each trial is called onset. The temperature at this point (the onset point) is the *Baseline Temperature* defined for this sequence.
- 2. From this point the temperature changes at a predefined rate, until it reaches the predefined *Destination Temperature*.
- 3. Then, the temperature remains at the destination temperature level for the predefined duration time. The point at the end of the duration time is called end of duration. Setting duration time to zero produces peaks.
- 4. Finally, the temperature returns at the return rate, to the baseline temperature. The temperature remains at this level for a duration that can be specified in one of two ways: as an end-to-onset time or an onset-to-onset time.

At the start of each sequence the temperature should remain at the *Baseline Temperature* level for the time given by the parameter *Time Before Sequence*.



"Return to the Next Destination" option

The temperature at the End of Duration point goes directly to the Destination Temperature of the next sequence when this option is used.

This is available only for sequences consisting of one trial.

See definition of each parameter in section 7.6.4, page 73.



Figure 3: Ramp & Hold Sequence – Return to Next Destination

2.7.5 VAS Search Method

In VAS Search method, first Ramp & Hold stimulus with predefined intensity and duration is given and the subject is asked to estimate the perceived magnitude of pain. The intensity of the next stimulus is set according to the response – increases if the reported pain rating is lower than the desired, decreases if the reported pain rating is higher than the desired. The step size by which the intensity of the next stimulus is changed can be predetermined during program creation or set by the operator during the test. The VAS Search method can be used to individually define what temperature is associated with patient's pain rating.

The temperature associated with the subject individualized rating of pain can be determined in one of two modes – automatic and manual.

In Manual mode only the first stimulus is predefined, the intensity of the next stimulus is set by the operator during the program run using the following window that pops up after each trial:



Figure 4: VAS Search – Manual Mode



In this window the user selects if to increase or decrease the destination temperature of the next trial and by how much (determined by the 'Step Size' value). There is also an option to resume the program if the operator decides that the target pain rating was reached.

In Automatic mode the target pain rating and the step size are predetermined during program creation and the temperature of each trial is defined according to these parameters and the subject reported pain rating - increases by the step size value if the reported pain rating is lower than the target and decreases by the step size value if the reported pain rating is higher than the target.

See section 7.6.5, page 75 for full description of program details and parameters.

2.7.6 Chain Option

The chain option enables to chain any of the above methods, any number of times, in any order.

For detailed information refer to section 7.6.6, page 77.



3 System Components

Q-SENSE components contain the following receptacles and controls:

Table 4: Q-SENSE Connectors si	ide Panel receptacles
--------------------------------	-----------------------

#	Description	Mate with
1	Q-sense Thermode Main receptacle Blue – power connector Green – Sensor connector	
2	Response Unit receptacle Use for recording patient response during test	

Table 5: Connectors to power side panel

1	Power Supply Power supply is used to connect electricity port and to the power port in the Q-sense system.	
2	Communication USB Cable This cable is used to connect the Q-sense com port with a computer.	

3.1 Power side Panel





Figure 7: Q-SENSE System Wiring Schema



4 Setup and Installation

The Q-SENSE system installation is designed to be simple and fast. Follow the instructions given below for selecting a location, installing and initial. Proper installation and operation of the equipment is important for ensuring the safety of the tested subjects.



Complete and verify the entire Q-Sense setup and installation process before the using the system

4.1 Computer Requirements

The requirements below are **minimum** requirements intended to be used as a guideline for selecting a computer to use with Medoc Main Station software.

Safety	Computer must comply with IEC 950 – EN 60950 – UL 60950
Processor	Intel Core i3 4xxx or better (avoid low power processors such as Atom, Pentium or Celeron)
Memory	4 GB RAM
Storage	128 GB
USB slots	2 free USB slots (3 for AlgoMed and Q-Sense CPM)
Monitor/Display	13" (15" recommended)
Screen Resolution	1366 x 768 pixels, 96 DPI (max. 1920x1080)
Operating System	Windows 7, 8, 10 32/64 bit (Not recommended: Windows XP service pack 3. Please see note below)
Other	Microsoft Office 2007 or above (required for viewing results in excel)



The following Medoc Main Station updates will no longer support Windows XP. Please avoid using Windows XP operating system if possible.



The computer used for Medoc Main Station software should be dedicated for this purpose only. Medoc takes no responsibility for any conflicts which may occur with other programs



For optimal performance disable any other resource heavy programs running on the same computer such as Anti-Virus, Screen Saver, Hibernate Mode, and Network connections.


4.2 Installation USB Content

The installation USB contains the MEDOC Main Station setup program and additional features that are required for system normal operation, as detailed below:

- 1. MEDOC Main Station setup program.
- 2. Software license files.
- 3. Technical Reference Manual.
- 4. Technical Reference Service Manual.
- 5. Remote assistance tools.



The installation USB contains all of the unique system files. Keep it in a safe place. It may be required for restoring the system configuration in case of malfunction and/or service during remote control session by Medoc.

4.3 Setup and Installation Main Steps

The assembly procedure consists of the following steps:

- 1. Selecting a location for the system. See section 4.4, page 37.
- 2. Connecting the Thermode and any additional accessories to the system. See section 4.5, page 38.
- 3. Connecting the laptop and installing the MEDOC Main Station software. See section 04.6, page 39.
- 4. Configuring the system.



See Q-Sense Technical Reference Service Manual for further details on the system components.

4.4 Selecting a Location

An appropriate location for the Q-SENSE system should meet the following criteria:

- 1. The room temperature should be kept between 18°C and 24°C.
- 2. The Q-Sense system should not be placed in a location, which is exposed to direct sunlight or constant vibrations, or close to heaters or air conditioners.
- 3. The space allocated for the system should allow enough room to perform regular system maintenance procedures.
- 4. The room in which the system will be located should be quiet, allowing the tested subject to concentrate on the examination with minimum distractions.
- 5. It is possible to hang the Q-Sense system using Medoc Wall Hanger. To hang the system: Connect Medoc Wall Hanger to the specific site on the wall.
 - 5.1. Mark the screw's holes on the wall, and take the hanger off the wall.
 - 5.2. Use a 6 mm drill bit to create holes on the wall in the points you marked in b section.
 - 5.3. Insert masonry anchors to the drilled holes.
 - 5.4. Connect the hanger by screwing the attached screws to the masonry anchors.



Different type of walls requires different masonry anchors.





Figure 8: Connecting Wall hanger to the wall

4.5 Device Connections

- 1. Ensure that the laptop meets the requirements set by Medoc (See section 4.1, page 36).
- 2. Make sure that your Laptop, Q-SENSE unit and printer (optional) are all OFF.
- 3. Connect the system, Thermodes and accessories. See Figure 9





Figure 9: Q-SENSE System Wiring Schema



Do NOT connect the USB cable to the Q-SENSE system before <u>complete</u> installation of the MEDOC Main Station software.



Thermode and Response Unit connectors are color coded. The Response Unit plug should be connected to yellow receptacle. The Thermode green and blue plugs should be connected to the green and blue receptacles respectively.



The SIP/SOPs (signal input and outputs) must be connected to medical grade safety approved equipment for Canada and the United States by a recognized certification agency, which is connected to the common protective earth of the system.



Do not connect any equipment to a Q-SENSE system that has not been specifically authorized and approved by Medoc.



Figure 10: DC power port

4. Connect the Q-SENSE power supply to an AC outlet and the cable to the DC power port.

4.6 Software Installation

The installation procedure of MEDOC Main Station software can only be performed by a user with **windows system administrator rights**. Without appropriate access rights, the installation procedure may fail.

Users with previously installed versions of Medoc Main Station should review the upgrading instructions in section 23, page 156.



Depending on the current configuration of the computer, MEDOC Main Station installer may require the installation of additional components. The installation procedure may take several minutes.



Disable any antivirus that runs on the PC before installing Medoc Main Station.

The following instructions will guide you through the installation procedure:

- 1. Insert the **USB drive** into an available USB port on the computer
- 2. Open the folder to view files.



- 3. Double-click **Setup.exe** to start software installation
- 4. Click **Next** to continue the installation.
- 5. Select the file location where Medoc Main Station should be installed. You can use the location recommended by the installation. Click **Next** to continue.
- 6. Select the location of the Medoc Main Station shortcuts in the start menu. You can use the location recommended by the installation. Click **Next** to continue.
- Installing normative data please read the agreement carefully. If you agree to the term please choose to install normative data. Click **Next** to continue.
- 8. Privacy please read the terms carefully. If you agree please choose to allow data collection. You may cancel this option any time later if you choose to.
- 9. The installation is now in progress. Depending on your computer configuration it may take up to several minutes.
- 10. After installation has completed, click **Finish** to close the wizard.
- 11. **Restart** your Computer.
- 12. Medoc Main Station software is now ready to be used.

4.7 Unit Protection



- Keep all liquids away from the Q-SENSE system.
- Unplug the power supply of the Q-SENSE system from electricity if it is not to be used for a long period of time.
- **Do not block airflow anywhere around the Q-SENSE system and Thermode.**



When all setup and installation procedures have been completed you may start using the system in accordance with the operation instructions in section **5**

4.8 Product Activation

When starting Medoc Main Station software for the first time, you will be asked to register your new system and activate the software. Product registration and software activation is simple and fast and it will enable you to maintain full use of Medoc Main Station software as well as:

- ✓ Warranty on your newly purchased Medoc systems
- ✓ Access to Technical Support and Customer Service assistance
- ✓ Software updates and new product notifications
- ✓ Product support alerts

Please note that if you don't activate the software within the designated time, the software will work in demo mode only.

To complete the activation process, simply start Medoc Main Station software and follow the instructions given in the activation wizard. If the activation wizard is not displayed when you start Medoc Main station Software, go to the Home screen, select Options and then select Product Registration. See Figure 11.



If at any point during the product activation process you encounter any difficulties please contact Medoc support at support@medoc-web.com for assistance.



Figure 11: Home screen product registration link



5 Operation

The purpose of the following chapters is to specify the Q-SENSE applications and to bring indetail step-by-step instructions for conducting thermal stimuli.

This chapter specifies Q-SENSE behavior at start up and MEDOC Main Station basic components.

5.1 Power up Q-SENSE System



Make sure that all communication and power cables, accessories and Thermode are properly connected to the system!



It is not allowed to connect the Q-Sense computer to a network while running tests with the system!

1. **Connect** your PC using USB cable to the **COM** connection, communication port, at the side of the Q-SENSE system.



Figure 12: Q-sense COM port

2. Power up the Q-SENSE system by using the main switch on the side of the Q-SENSE.

5.2 Starting Q-SENSE



Before initializing the Q-SENSE system, ensure that the Thermode is connected correctly.



The Thermode should be attached to the subject's skin <u>only</u> when performing a test. <u>Do not</u> attach the Thermode to the subject during system Self-Test, while programming the system or performing maintenance.

- 1. Turn **ON** the Q-SENSE unit, by using the main power switch
- 2. Turn **ON** the laptop
- 3. Start Medoc Main Station software by double-clicking the icon
- 4. The Home screen is displayed showing the available devices (license dependent).





Figure 13: Home Screen

- 5. To select Q-Sense, click on the Q-Sense image.
- 6. When the login screen is displayed, enter your user name and password and select "**Q**-**SENSE**" in the "**connect to**" field.



Initial login Username and Password are both admin.

For more information on user management please refer to section 10.1.5



Figure 14: Login Screen

7. While the software is establishing communication with the device the following screen may be displayed:



Figure 15: Q-Sense Splash Screen



- Q-Sense Technical Reference Manual
- 8. When communication is established, the system will enter SAFE MODE and remain in this state until completing the system Self-Test, see section 5.3 below. Make sure the Thermode is not attached to the Subject and click on OK on the following message:

5.3 System Self-Test

Upon system start up, Q-SENSE will automatically enter SAFE MODE state. The system performs a self-test in which system sensors are being tested. If a malfunction is detected, an appropriate message is displayed and the system cannot operate until that malfunction is resolved.

System self-test must be completed successfully before Q-SENSE can be used. The following message prompting for system self-test is displayed. Make sure the Thermode is not attached to a patient and click **OK**.



Figure 16: Self-Test Prompt

The system will perform a complete self-test of all enabled Thermodes according to your system settings.

After self-test has completed successfully, Q-SENSE State will change to **Rest Mode**.



Make sure that Thermode is not attached to human subject before self-test is complete.



TEST								Medo
	PATIENTS	PROG	RAMS	RESULTS			User Na	me:Administra
	Select Patient					N. II		
	Department	ID	Last Name	First Name	Gender	Date of Birth	His	ory C
PR ST	Department 1	8787878	Doe	Isabel	Male	11/18/2012		A
SCT PP	Department 1	12345678	Smith	John	Male	11/26/1982		A
SRUTCI BOD								
REACT BOOT								
Current Selection Patient Undefined Program Undefined								
Current Selection Patient Undefined								(ii

Figure 17: Q-Sense in Rest Mode

5.4 Basic Operation

As soon as the self-test is successfully completed the **TEST** screen is displayed:



Figure 18: Test Navigator – Main Menu





Note that the status line also indicates the type and model of the Thermode, you are currently applying to (see red circle, above).

The Test screen enables you to select a patient, program and body site before starting the test procedure.

The **TEST** screen menu consists of the following components:

- Menu Bar Perform general and software related operations
- Toolbar Quick access to patient and program editors and saved test results
- Search Row Filter the data for quick access to patients and programs
- Current Selection Information Display of current selections made by user
- Data Display Area View list of patients, programs or body sites
- Status Line Continues real-time display of Q-SENSE status and Thermode temperature.

The components are detailed in the following sections.

5.4.1 Menu Bar

The Menu bar consists of the following options:

Menu	Option	Description
File	Export	Export selected Patient data, Program parameters or Test Results to *.ats file format. The *.ats file format is encoded and can be read only by MEDOC software, enabling simple backup, export and import options while maintaining patient confidentiality. For detailed export and import instructions see relevant sections in Patient, Program and Results Management chapters.
	Import	Import Patients, programs and test results saved in *.ats file format. See above description on Export.
	Export to Excel	Export Tests Results to MS Excel format, for data analysis.
	Print	Print Test Results as a Report (enabled in the Results screen). Refer to section 9.3 for further information.
	Logoff	Log off and switch users without exiting the software or shutting down Q-SENSE.
	Switch Device	Return to Home Screen and select another device or change preferences.
	Exit	Exit MEDOC Main Station.
View	Test	Switch to Test Editor screen to perform and manage tests from the available patient and program library.
	Patients	Switch to Patient Editor screen for managing Patient lists.
	Programs	Switch to Program Editor screen for managing programs and protocols.

Table 6: Menu Bar options



Menu	Option	Description
	Results	Switch to Results screen to view and manage saved test results.
Settings	Software settings	MEDOC software related configuration and preferences.
	Hardware settings	Q-SENSE system and Thermode hardware configuration and management.
	Test settings	Test configuration and display preferences
	Status	Status configuration- On line\Demo.
Utilities	Department	Add/Remove Departments for patient list management.
	ICD 9	Add/Remove ICD 9 codes.
	Normative Data	Normative Data database and Editor.
	Black box	System hardware and software logs
	Device recovery	Initiate Q-SENSE system self-test in case safety was triggered.
	Log Files	View software log files
	Probe Files	Import or export the termode calibration details.
Help	Operating Manual	Link to the Q-SENSE Technical Reference Manual
	Service Manual	Link to the Q-SENSE Technical Reference Service Manual
	About	System and software information
	Service Call	Submit a Service Call via Medoc web site*.
	FAQ	FAQ on Medoc web site*.

*internet connection required

5.4.2 Toolbar

The Toolbar consists of shortcuts to the different views:

- Test Navigator Select test parameters
- Patients View Manage patient database
- Programs Manage program list
- Results View, export or print test results

5.4.3 Search Row

The Search Row allows you to filter the displayed data, in order to find the required data quickly. To use the filter, place the cursor in the in search row, in the column you want to filter by and start typing the beginning of the word.



5.4.4 Current Selection Information

Current selection panel displays the currently selected items. The data may vary depending on the current view. While in test navigator for example, the current selection will show the currently selected patient, program, body site and enabled Thermode.

5.4.5 Data Display Area

Displays list of data which is relevant to the current screen.

5.4.6 Q-Sense Status Line

The Status bar displays important information about the status of the connected device. If the status line is empty, MEDOC Main Station is not connected to any device. When connected to the PATHWAY system, the status line displays the following information as described in Table 7.

Information	Description
Device	The device currently in use and connected to Medoc Main Station
Status	The current status of the system, either Demo or On-Line.
Com	Communication status between the system and the laptop. A green indicator is displayed for ON and a red indicator for OFF.
State	The state of the Temperature Control Unit. The following modes are available: Rest, Engineering, Safe and Test run.
Thermode	Indicates the type and model of the Thermode currently connected to the system: Q-sense 30x30mm.
ТЕС	Temperature of the Thermode.
HS1/HS2	Temperature of heat-sink (cooling element)
Fan Mode	Indicates the Fan mode.
Safety	Safety status. A green indicator is displayed for OK and a red indicator for Safe Mode activation – Thermode is disconnected from input. System remains non-functional, until malfunction is resolved.

Table 7: Status line parameters



6 Test Procedure and Management

Starting a test procedure is only possible from the Test Navigator screen



Figure 19: Test Management Main Screen

The stimulation procedure (test) consists of the following main steps:

- Select Patient Select the tested subject's file from the database.
- Select Program Select the relevant program that will be used for the test.
- Select Body Site Select the body site and its details for the current test. This step is optional.
- Go to Test and Apply Stimuli Perform the test
- **Save/Print Test Results** –Print test report and/or save test data for future reference (This step is optional).



Make sure you are familiar with the stimulation method before you start with the test procedure



Inform the subjects how to operate the applicable controls, and request them to concentrate on the perceived sensations throughout the full duration of the test.

The stimulation procedure steps are described in detail in the following sections.

6.1 Selecting a Patient

- 1. Click the Select Patient button on the Test Navigator
- 2. Use the Search Row (see section 5.4.3 above) to filter and sort list in order to find the relevant patient entry.



	atient List can be sorted according to any he relevant column header to sort the lis	
MMS - v.6.1.19.1	File View Settings Utilities Help PATIENTS PROGRAMS RESULTS Select Patient Department ID Last Name First Name Gr Department 1 12345678 Smith John	Male 11/18/1982
Program Undefined Body Site Undefined		Show/Hide Details Button
Device: Q-Sense Status:Or	lline Com: 🔶 State:Rest Mode Thermode:Q-Sense-30x30 TEC:32.00 °C HS1:2	24.02 °C HS2:45.46 °C Fan Mode:High Safety: 🔶 📃

Figure 20: Patient Selection Menu

- 3. Click the patient entry to select (or double-click to select and move on to the next step). The selected entry will be displayed in the current selection details in the left panel.
- 4. In order to view patient details, click on the **Show Details** 😓 button.
- 5. Click the **Hide Details** button to view the patients list only.



Patient details can only be viewed from the Test screen. For editing, refer to Patient Management section (section 8, page 83).

- 6. Click the **Clear** button to clear Current Test information if required.
- 7. Proceed to the **Selecting a Program** section.



It is not possible to run a test before you select the tested subject and program. The "Go to Test" button will remain disabled as long as test details are not complete.

6.2 Selecting a Program

- 1. Click the Select Program button on the Test Navigator
- 2. Use the Search Row (see section 5.4.3 above) to filter and sort list in order to find the relevant patient entry.





Figure 21: Program Selection Menu

- 3. Click the program entry to select (or double-click to select and move on to the next step). The selected entry will be displayed in the current selection details in the left panel.
- 4. In order to view patient details, click on the **Show Details**

🕏 button.

5. The following options are available in the program details view:

Display program details.
Display program graph preview. Available on selected methods.
View selected test instructions.
View recent program modification dates.



Program details can only be viewed from the Test screen. For editing, refer to Program Management section (section 7, page 58).



button to view the program list only.

7. Click on the **Clear** button to clear current selection if required.



6. Click the **Hide Details**

- Q-Sense Technical Reference Manual
- 8. Proceed to the **Selecting Body Site** section if you want to select site for the specific test, or click **Go to Test** in order to run the test without specific body site selection.



It is not possible to run a test before you select the tested subject and program. The "Go to Test" button will remain disabled as long as test details are not complete.

6.3 Selecting a Body Site

Selecting a tested body site is optional. A test can be initiated without selecting a body site.

1. Click on the Select Body Site button in the Test Navigator.



Figure 22: Select Body Site screen

 Click on the Dermatome (Dermatomic area) you wish to use in the current test. An enlarged view of the selected dermatome and a list of specific sites will be displayed on the right.





Figure 23: Select Body Site – Specific site selection

3. Use the Zoom In/Out controls for a more precise selection



- 4. Use the Flip Image control to switch between Anterior / Posterior view
- 5. Complete the site selection by selecting one of the Specific Sites. The selected site will be displayed in the current test details in the left side bar.
- 6. Click the "Go to Test" button to proceed to the test.

6.4 Running the Test



Make sure you have selected both Patient and a Program before you try to proceed with an actual test.

- 1. Click the **Go to Test** button The following screen is displayed:





Figure 24: Pre-Test Screen

- 2. Click the **Start Pre-test** button (). The system performs a short **pre-test**, and stabilizes on the *Baseline* temperature.
- 3. After a successful pre-test, the **Pre-test** button is replaced by a **Test Run** button (



At this point attach the Thermode to the tested subject.

Caution! DO NOT use any electrode paste, jell or other material on the contact point between the Thermode and the tested subject's skin.

Place the Thermode on the intended stimulation site so that the contact plate makes the best possible contact with the skin, fastening it lightly with the attached strap.

It is recommended to first fasten the strap around the area to be stimulated and then tighten the strap by pulling it an additional 2 cm before fastening the two ends.

- 4. Click the **Start** button (**D**).
- 5. To pause the stimulation, click the **pause** button (\bigcirc).
- 6. To stop the stimulation, click the **stop** test (button. The first click stops the test, and initiates the Finalizing test procedure (accompanied by a message), which saves relevant information in the Black Box in case of an error. A second click aborts the test immediately.
- 7. Once a test has finished, it is possible to use the mouse curser to display the exact time, temperature and other available information at any given point.
- 8. Click on the **Save** button to save the test results and stimulation data.



The system will prompt you to save the stimulation data before starting a test or exiting the current session.





After concluding a stimulation session, always clean the Thermode contact plate. For further details, refer to section 18, page 151.



Do NOT immerse the Thermode in any kind of liquid.

6.4.1 Test and Post-Test Options

Table 8: Test Run options

	Print test results as a report. Refer to section 9.3 for details.
	Save the stimulation data.
	Display test instructions for the subject. To create or edit the default instructions, refer to section 10.3.2.1
	Play audio test instructions for the subject. To assign an audio file with default instructions, refer to section $10.3.2.2$.
Α	Automatic zoom along the temperature axis. The temperature axis scale expands according to the highest and lowest temperatures displayed on screen. Button is located at the bottom left of the graph. Alternatively, change zoom manually by using the +/- buttons.
D	Reset zoom to the original (default) scale. Button is located at the bottom left of the graph.
16	Change the zoom along the time/trial axis, by changing the value in the spin box at the bottom right of the graph. Use the up/down arrows or type the value in the box.
New Test	Return back to the test screen to start a new test with or without clearing certain test details such subject and program. Last test results are not saved automatically.
\bigcirc	Indicate positive subject response in Limits, Levels, and TSL protocols.
N	Indicates negative subject response in Limits, Levels, and TSL protocols.
8	Trigger the next stimulus. Available only when the manual trigger option is selected. The button is enabled only when the system expects a trigger.
	Edit the program locally. The edited parameters are effective only in the specific test and the program parameters are not changed in the data base.



6.4.2 Temperature Graph Display

During the test, the test graph is generated and displayed on the screen. Following is an example:



Figure 25: Real-Time Test

The+ following information is displayed:

- 9. Stimuli display (including CoVAS and events where applicable)
- 10. Current probe and program details.
- 11. Current sequence and trial data
- 12. Legend
- 13. Axes labels: Temperature axis on the left and CoVAS on the right (where applicable)
- 14. Statistics Panel: hidden by default. To view in full click on it. To keep the display permanently open, use the pin icon (^{III}) on the top right corner. Click again to hide.
- 15. Test and Patient Details Panel: hidden by default. To view in full click on it. To keep the display permanently open, use the pin icon () on the top right corner. Click again to hide.

6.4.3 Test Run Keyboard Shortcuts (Hot Keys)

Keyboard shortcuts (hot keys) are available for several test functions enabling control of the test from the keyboard. The table below describes the available options.

Button	Кеу	Action
	S	Can be used instead of the Stop button to finalize; then stop the test.

Table 9: Hot Keys



Button	Кеу	Action
, , <	Spacebar	Can be used instead of the following buttons in sequence: pre-test / run test / pause / resume test.
Ŷ	Y	Can be used instead of the YES button. Not available in Search method.
	N	Can be used instead of the NO button. Not available in Search method.
8	т	Can be use instead of the Trigger button.
	Tab	Wrong Sensation option. Available in Levels and TSL only.
	Delete	To Delete / Undelete a trail from statistics of a Limits test. Available only after test is complete and before it is saved. Delete trail is colored in Yellow.



7 Program Management

This chapter describes how to manage your stimulation programs. Programs can be created, duplicated or removed from the database. Following are the detailed individual parameters for program.

7.1 Program List

The Program screen enables you to manage programs (add, remove, edit etc.).

1. To access the program database, click the **Programs** button at the toolbar. The following screen is displayed:

MMS - v.60.80				I Help
		NTS PROGRAMS	RESULTS	
Current Selection	Select a Program	Name	Modification Date	Description
	T			
Probe Undefined	Limits	wwe	04/07/2012	(A)
Method Undefined	TSL	111111111	05/07/2012	(A)
Program Undefined	TSL	TSL Demo	01/07/2012	(A)
Designer Undefined	Levels	Levels Demo	01/07/2012	(A)
	Limits	Limits CS & WS	01/07/2012	a)
	TSL	tsl	12/07/2012	A
	Levels	tz	12/07/2012	(A)
	TSL	tzahitsa	12/07/2012	
@ & ×				
Device: Q-Sense Status;Online				¢

Figure 26: Program Management screen

2. To narrow down the displayed list use the filter row.



The more criteria you define using the filter, the shorter the list that will be displayed.

3. To clear the filtered list, select the (All) option from the filter drop down list.



It is possible to change column order by drag and drop it to required place.

The following options are available in the Program Management screen:

	Create a new program
	Duplicate selected program
\otimes	Delete selected program
	Edit selected program





Show or Hide program preview

- Current Selection control displays the selected program information.
- Current Thermode control displays the Thermode that is used for the selected program.

When you open the Program Preview bar, the following screen is displayed:



Figure 27: Program Management – Program Preview

The following options are available in the Program Preview:

	Display program details.
	Display program graph preview. Available on selected methods.
()	View selected test instructions.
	View recent program modification dates.



Program details cannot be edited from the preview screen.



Figure 28: Program graph preview

7.2 Creating a New Program

To create a new program, perform the following procedure:

1. From the **Programs** screen, click the Create New Program (¹) icon. The following dialog is displayed:



Figure 29: New Program Screen

2. In the **Probe Type** drop-down list, select the Q-sense probe. The following options are available (in accordance with your licensing agreement):

Table 10: Probe Type options

Parameter	Description
Q-sense	Can operate with one of the following methods: Levels, Limits, TSL, Ramp & Hold (optional, license is needed), Chain.

- 3. From the **Method** drop-down list, select the required stimulation method (Limits, Levels TSL or Chain).
- Type the program name.
 For faster identification, it is advisable to type a meaningful name.



 Click OK. The Program Details screen is displayed (see below).

7.3 Removing a Program

To remove a program from the program list:

- 9. Click the **Remove** icon (
- 10. Click **Yes** to the prompt. The selected program is deleted.

7.4 Duplicate A Program

To duplicate a program from the program list:

- 1. Click the **Duplicate** icon (
- 2. The selected program was duplicated and given a unique name automatically (the name is editable).

7.5 Edit A Program

To edit a program from the program list:

- 1. Click the **Edit** icon (
- 2. The selected program details screen is open and can be edited.

7.6 Program Details

The following screens enable the formation of program protocols. There are two test types: Programmed – the Thermode temperature varies according to the program parameters.

Manual – the Thermode temperature varies according to selected accessory.

Following are sampled menus:

- Limits
- Levels
- TSL
- Ramp and Hold (optional)
- VAS Search (optional)
- Chain

The following functions are available in the program menus:



Displays a list of the program revision history by version date and username.



- Displays the Test Instructions Editor for editing test instructions. Refer to section 7.8, page 80.
- Returns to the previous menu.
- Copies the selected record (sequence). Not available in Levels, TSL and Chaining programs.
 - Pastes the copied record to the right of the selected record. Not available in Levels, TSL and Chaining programs.
- Adds a new record to the end of the current program. Not available in Levels, TSL and Chaining programs.
 - Adds a new record to the left of the selected record. Not available in Levels, TSL and Chaining programs.
- •

Previews the program graph.

- Checks program validity, and marks missing or incorrect parameters.
 - Saves the information displayed.

7.6.1 Limits

In this method, the system generates a number of stimuli of continuously changing intensity. With each stimulus, when the subject perceives a predefined sensation, he/she manually stops the stimulus. For method explanation, refer to section 2.7.1, page 29.



MMS - v.6.1.17.0							00
							Medo
		PROGRAMS	RESULTS				
	Program Name: Im	Probe:	Q-Sense Start Test:	Sound			
Current Selection Probe Q-Sense							
Method Limits	Updated On: 13/12/2012	Description:	End Test:	Sound			
Program lim Designer Administrator		-					
Cear J	Sequence Modality	1 Warm Sensat 💌					
	Baseline	0 0					
	Trigger	Auto 👻					
	Rate	00					
	Return Rate	00					
	Number of Trials	1					
	Randomize with Next	•					
	ISI Min (sec) ISI Max (sec)	00					
	Standard Events						
	Number of sequences in the test : 1						
Device: Q-Sense Status:D	emo Com: 🖶 State:Rest Mod	e Thermode:Q-Sen	e-30x30 TEC:32.00 °C HS1:1	15.00 °C HS2:10.00 °C	Fan Mode:OFF Safety:	8	

Figure 30: Program Menu – Limits

When adding or editing programs, follow these rules:

Enter meaningful information in the Program Description section of the screen.

Define all the parameters of a sequence for that sequence to be valid.

If not all the sequences are displayed, use the horizontal scroll bar to display the ones that are out of sight.



Missing or incorrect parameters are identified by an 60 icon that flashes next to the relevant field. A short description of the malfunction can be displayed by holding the cursor over the relevant 60 icon.

The program menu varies according to the probe, and the method of stimuli intended for use, as shown on the left side of the screen.

The parameters available for program creation at this stage are as follows:

Table	11:	Program	Details
-------	-----	---------	---------

Parameter	Description
Program name	The program name (editable).
Probe	The probes used with this program.
Description	Under Description, click the arrow text to describe the program. The Description icon is indicated as active (



Parameter	Description
Update on	The date on which the latest changes were made and saved.
Start test/End test	Defines if these events are indicated by sound.

Table 12: Program Parameters

Parameter	Description
Modality	Four different types of sensation can be delivered to the subject: Warm sensation; Cold sensation; Hot pain.
Baseline (°C)	Temperature from which each new stimulation sequence begins – between 28°C to 40°C for Q-sense thermode.
Rate (°C/sec)	Defines the temperature rate of increase per second, from the baseline until the subject's response. Type an acceptable value: 0.1 to 2°C/Sec for Q-sense thermode
Return Rate (°C/sec)	Defines the temperature rate of decrease per second, from the temperature at which the subject responded, down to the baseline. Type an acceptable value:0.1-1°C/sec for Q-sense thermode.
Number of Trials	The number of trials in the sequence – between 1 to 100.
Randomize with next	Randomizes the next trial or sequence.
ISI Min (sec) / ISI Max (sec)	Variable time between trials. Type acceptable value: 2-100.
Standard Events	Enables the system to alert you when stimulation procedure arrives at some standard events.



The following screen shows a test consisting of 2 sequences, each comprising of 4 trials:



Figure 31: A Limits Method Test



Test statistics are hidden by default.

The bottom of the screen displays the total records (sequences) in the current program.

To view the statistics details, place the cursor at the bottom end of the statistics tab and drag it down.

Click the **Tack** icon at the upper right of the screen to lock or unlock the statistics display (indicated by the circle in Figure 31).

Following are the statistic details:

Statistics				
Sequence	1	2		
Modality	WS	CS		
Trials	3	0		
Average	36.41	0		
Variance	15.04	0		
STD	3.88	0		

Figure 32: Statistical analysis of results

The following information is displayed:

• Sequence index.



- Modality WS (warm sensation); CS (cold sensation); HP (hot pain).
- Number of recorded trials.
- Averaged temperature In each of the sequence trials, the subject stops the temperature change once perceiving a predefined sensation. The temperatures at which the subject stopped the stimulations are averaged to yield the value of the threshold.
- Variance and STD (Standard Deviation) Statistical measures of the variation in the distribution of the temperatures that are averaged to give the threshold. Variance and STD are calculated for a finite sample.



It is possible to remove a Limits Trail using the Delete button on the PC key board. Refer to section 6.4.3, page 56 for details.

7.6.2 Levels

In this method, the system generates a number of stimuli of changing intensity according to the subject response. With each stimulus, the subject receives a stimulation of varying intensity which is determined according to his previous response. Test is terminated after predefined temperature step size and number of alternations is achieved. For a method explanation, refer to section 2.7.2, page 29.



Figure 33: Program Menu – Levels

When adding or editing programs, follow these rules:

Enter meaningful information in the Program Description section of the screen.

Define all the parameters of a sequence for that sequence to be valid.

If not all the sequences are displayed, use the horizontal scroll bar to display the ones that are out of sight.





Missing or incorrect parameters are identified by an ³² icon that flashes next to the relevant field. A short description of the malfunction can be

displayed by holding the cursor over the relevant $^{old O}$ icon.

The program menu varies according to the probe and the method of stimuli in use, as shown at the left side of the screen.

The parameters available for program creation at this stage are as follows:

Parameter	Description
Program Name	Displays the name of the program (editable).
Probe	Displays the probe that is used in this program.
Description	Under Description, click the arrow reader; then type free text to describe the program. The Description icon is indicated as active (
Update on	The date on which the latest changes were made and saved.
Start test/End test	Defines if these events are indicated by Sound.

Table 13: Program Details

Table 14: Program Parameters

Parameter	Description
Modality	Two different types of sensation can be delivered to the subject: Warm sensation; Cold sensation.
Baseline (°C)	Temperature from which each new stimulation sequence begins – between 28°C to 40°C for Q-sense thermode.
Trigger	Defines the trigger type: Auto or Manual.
Time Before Sequence (sec)	The time interval before the first trial of the current sequence between 0 to 600 seconds.
Destination Rate (°C/sec)	Defines the temperature rate of increase or decrease per second, from the baseline until the current destination temperature. Type an acceptable value: 0.1 to 2°C/Sec for Q-sense thermode Increasing temperature or 0.1 to 1°C/Sec for Q-sense thermode Decreasing temperature



Return Rate (°C/sec)	Defines the temperature rate of increase or decrease per second, from the destination temperature, down to the baseline. Type an acceptable value:
	0.1 to 2°C/Sec for Q-sense thermode Increasing temperature or 0.1 to 1°C/Sec for Q-sense thermode Decreasing temperature
Max. Time at Destination (sec)	Defines the maximum duration in seconds, between the time that the temperature reached the destination and the time it returned to the baseline. In case a YES response was detected, the maximum time will not complete, and the temperature will return to the baseline and continue to the next trial. Acceptable value: 0 to 600, and according to safety temperature – time limitations.
Min. Interval (sec) / Max. Interval (sec)	Variable time between trials. Actual time is set randomly in between the min. and max. values.
First Choice Step (°C)	The initial step size that is used until the first YES response is detected. Acceptable value: 0.2 – 5 °C
Final Choice Step (°C)	The fixed step size that is used in the final Turns Around Threshold. Acceptable value: 0.1 – 2 °C
Turns Around Threshold	The number of turns (YES to NO or NO to YES) at the Final Choice Step which is required to complete the test.
Number of Tests	The number of tests in a sequence – between 1 to 5.
Standard Events	Enables the system to alert you when stimulation procedure arrives at specific standard events.
Dummy Option	Enables dummy stimulus to increase test reliability. Dummy frequency is randomized, and it has average of 10%.
Randomize	Randomizes number of tests in order to increase sequence reliability.
Display Message	Determine when message to test subject is displayed on screen. Available options: Destination – Display message upon reaching trail destination temperature Return to Baseline – Display message upon trail return to baseline temperature
No Response on Destination	Define program behavior if subject responds after the Maximum Time at Destination is over. Relevant only if Maximum Time at Destination is larger than 0 and Display Message is set to Destination. Always Accept – Accept subject's response and continue the test Always Repeat – Ignore subject's response and repeat the same trial again Manually Select – Manually select whether the response should be accepted (press 'A') or ignored (press 'I').





Click the **Tack** icon at the upper right of the screen to lock or unlock the statistics display (indicated by the circle in Figure 31).



Following are the statistic details:

Statistics		
Sequence	1	2
Modality	WS	WS
No. of stimuli Turns	4 1	2 0
Threshold	0	0
Dummies Total Yes No	0 0 0 0	0 0 0

Figure 35: Statistical Analysis of the Results

The following information is displayed:

- Sequence index.
- Used modality WS (warm sensation); CS (cold sensation).
- Number of recorded stimuli and number of turns.
- Threshold In test trial, the subject response is recorded. The threshold is calculated as the average response of the last two trials.
- Dummies The total number of dummy stimuli and their response type (Yes or No).

7.6.3 Thermal Sensory Limen (TSL)

In this method, the system generates warm sensation and cold sensation stimuli, alternating according to the subject response. The stimuli direction changes after each subject response. For a method explanation, refer to section 2.7.3, page 29.



Figure 36: Program Menu – TSL



When adding or editing programs, follow these rules:

Enter meaningful information in the Program Description section of the screen.

Define all the parameters of a sequence for that sequence to be valid.

If not all the sequences are displayed, use the horizontal scroll bar to display the ones that are out of sight.



Missing or incorrect parameters are identified by an \bigotimes icon that flashes next to the relevant field. A short description of the malfunction can be displayed by holding the cursor over the relevant \bigotimes icon.

The program menu varies according to the probe and the method of stimuli in use, as shown at the left side of the screen.

The parameters available for program creation at this stage are as follows:

Parameter	Description
Program Name	Displays the name of the program (editable).
Probe	Displays the probe that is used in this program.
Description	Under Description, click the arrow , then type free text to describe the program. The Description icon is indicated as active ().
Update on	The date on which the latest changes were made and saved.
Start test/End test	Defines if these events are indicated by Sound.

Table 15: Program Details

Table 16: Program Parameters

Parameter	Description
Baseline	Temperature from which each new stimulation sequence begins
Temperature (°C)	– between 28°C to 40°C for Q-sense thermode.
Time Before	The time interval before the first trial of the current sequence.
Sequence (sec)	Ranges between 0 to 600 seconds.
First stimulus	Determines whether the TSL test will start with warm or cold stimulus.



Increase Rate	Defines the temperature rate of increase per second, from the destination temperature down to the baseline.
(°C/sec)	Type an acceptable value: .0.1- 2°C/sec for Q-sense thermode.
Decrease Rate	Defines the temperature rate of decrease per second, from the baseline until a response is recognized.
(°C/sec)	Type an acceptable value: 0.1-1°C/Sec for Q-sense thermode
Number of stimuli	The number of required subject responses until test termination.
Standard Events	Enables the system to alert you when stimulation procedure arrives at specific standard events.

The following screen shows a sequence consisting of TSL Method Test:



Figure 37: A TSL Method Test



Test statistics are hidden by default.

The bottom of the screen displays the total records (sequences) in the current program.

To view the statistics details, place the cursor at the bottom end of the statistics tab and drag it down.

Click the **Tack** icon at the upper right of the screen to lock or unlock the statistics display (indicated by the circle in Figure 37).


Following are the statistic details:

1
41.35
28.16
13.19
3
0
0

Figure 38: Statistical Analysis of the Results

The following information is displayed:

- Sequence index.
- WS (warm sensation) Threshold
- CS (cold sensation) Threshold
- D the absolute difference between WS and CS
- Number of recorded stimuli.
- Variance and STD (Standard Deviation) Statistical measures of the variation in the distribution of the temperatures that are averaged to give the threshold.

7.6.4 Ramp and Hold

In this method, the temperature rises (or falls) to a predetermined destination, at a predetermined rate, remains there for a predetermined duration, and proceeds to the next destination. For an example graph and an explanation of the method, refer to section 2.7.4, page 30.



Figure 39: Program Menu – Ramp and Hold

The table below describes the program parameters available for Ramp and Hold method:



Parameter	Description	
Baseline (°C)	Temperature from which each new stimulation sequence begins.	
	- Between 28°C to 40°C for Q-sense thermode.	
Time before	The time interval before the first trial. Valid range: 0 to 600 (sec).	
sequence (sec)		
Trigger	Trigger mode:	
	 Automatic – the stimulus will start automatically after the defined Inter Stimulus Interval (ISI). 	
	 Manual – The operator must initiate the onset of the stimulus after the defined ISI 	
Destination temperature (°C)	Defines the destination temperature for all trials in the current sequence. Valid range is limited by safety limitations and depends on the duration time.	
Destination rate (°C/Sec)	Defines the temperature change rate from the baseline to destination temperature.	
	0.1 to 1°C/Sec for Q-sense thermode.	
Destination criterion	Temperature: The system recognizes that it has reached the target temperature, by monitoring the temperature.	
	Time: The system recognizes that it has reached the target temperature, by referring to the theoretical time that is required to reach the destination temperature according to the selected rate.	
Duration time (sec)	Defines the duration for Thermode temperature to remain at destination temperature before returning to baseline.	
	Valid range: 0 to 600 (sec)	
Return option	Defines whether the temperature returns to the baseline or proceeds to the destination temperature of the next sequence. Select the relevant option from the drop-down menu.	
	Note that when the "next destination" option is selected certain fields become inactive, because they are no longer relevant (for example, the Number of Trials turns to $1'$).	
Return Rate (°C/Sec)	Defines the temperature change rate from the destination temperature back to baseline.	
	0.1 to 1°C/Sec for Q-Sense thermode	
Number of Trials	The number of trials per sequence. Valid range: 1 to 100	
Enable Pain Rating	Enables pain rating patient response at the end of the duration time of each trial	
	Advanced options are available for configuration with specific license. See section 25.4 page 180	
Randomize with next	Randomize trials with next sequence	
ISI Min (sec)	Minimum Inter-Stimulus-Interval (ISI). Valid range: 0 to 600 (sec)	
	l	

Table 17: Program Parameters



Parameter	Description	
ISI Max (sec)	Maximum Inter-Stimulus-Interval (ISI). Valid range: 0 to 600 (sec). Actual ISI will be randomized between min and max values.	
Standard events Enables the system to alert you when stimulation procedure at specific standard events.		



Note that this program is available only with special license and not a part of basic Q-Sense configuration

7.6.5 VAS Search

In this method, the temperature of the first stimulus rises (or falls) to a predetermined destination, at a predetermined rate, remains there for a predetermined duration, and returns to baseline. Patient estimation of perceived magnitude of sensation is recorded for the first stimulus. The intensity of the next stimulus is determined by patient response being higher or lower than the desired pain rating. For a method explanation, refer to section 2.7.5 page 31.

TEST		Utilities Help		Medoc User Name:Administrator
Current Selection Probe Q-Sense Method VAS Search	Program Name: test Update Probe: Q-Sense Descrit	d On: 07/03/2016 Start Test: Sound		
Program test Designer Administrator	Sequence	1		
	Baseline	32		
	Time Before Sequence (sec)	3		
	Trigger	Auto		
	First destination temperature (°C)	45		
	Destination Rate	2		
	Duration Time (sec) Return Rate	3		
	Search Mode	Auto		
	Target Pain Rating	6		
	Step size (°C)	1		
	Waiting time for response (sec)	10		
	ISI Min (sec)	5		
	ISI Max (sec)	7		
	Standard Events			
0 🕑 🛑				
	Number of sequences in the test : 1			
Device:Q-Sense Status:Demo	Device:Q-Sense Status:Demo Com: 🗮 State:Rest Mode — Thermode:Q-Sense-30x10 TEC:32.00 °C H52:10.00 °C H52:10.00 °C H52:10.00 °C Fin Mode:OFF Safety: =			

Figure 40: Program Editor – VAS Search

The table below describes the program parameters available for the VAS Search method:

Table	18:	Program	Parameters
-------	-----	---------	------------

Parameter	Description			
Baseline (°C)	Temperature from which each new stimulation seque range:	Temperature from which each new stimulation sequence begins. Valid range:		
	28°C to 40°C	28°C to 40°C		
Trigger	Trigger mode:			
	 Automatic – the stimulus will start automatically after the defined Inter Stimulus Interval (ISI). 			
A CONTRACTOR AND A CONT	Proprietary 75 of 188			

	Manual – The operator will initiate the onset of the stimulus after the defined ISI.		
First destination temperature (°C)	Defines the destination temperature for the first trial. Valid range is limited by safety limitations and depends on the duration time.		
Destination Rate (°C/sec)	Defines the temperature change rate from the baseline to destination temperature. Valid range: 0.1 to 2°C/Sec		
Duration Time (sec)	Defines the duration for Thermode temperature to remain at destination temperature before returning to baseline. Valid range: 0 to 600 (sec)		
Return Rate	Defines the temperature change rate from the destination to baseline temperature. Valid range: 0.1 to 1°C/Sec		
Search Mode	Search mode:		
	 Automatic – starting from the second trial, the stimulus parameters will be defined automatically by the predetermined target pain rating and step size. Manual – At the end of each trial the user will define the destination temperature of the next trial, until the desired pain rating is reached. The user also decides when to finish the program (i.e. when the desired pain rating is achieved). In manual mode the 'Target Pain Rating' and 'Step size' parameters become dimmed. 		
Target Pain Rating	Relevant for Automatic search mode only.		
	Defines the pain rating that needs to be reached. If the actual pain rating of the current trial is lower than 'Target Pain Rating' value, the temperature of the next trial will increase by the Step Size (see below) and vice versa.		
	Valid range: 1 - 100		
Step Size (°C)	Relevant for Automatic search mode only.		
	Determines the step size by which the temperature will increase/ decrease in every new trial. The direction of the temperature change will depend on the difference between the target and the actual pain ratings (see above in 'Target Pain Rating' description)		
	Valid range: 0.1 – 10 (°C)		
Waiting time for response (sec)	Time for which the pain rating dialog will remain on the screen waiting for user to respond.		
ISI Min (sec)	Minimum Inter-Stimulus-Interval (ISI). Valid range: 0 to 600 (sec)		
ISI Max (sec)	Maximum Inter-Stimulus-Interval (ISI). Valid range: 0 to 600 (sec). Actual ISI will be randomized between min and max values.		
Standard Events	Different evens which can be indicated during the test via TTL or Sound. See section 7.6.7		



Note that this program is available only with special license and not a part of basic Q-Sense configuration



7.6.6 Chaining Programs

The purpose of the Chain Method is to chain several programs one after the other. The result is a master program which runs the selected programs continuously, each program repeated by the number of cycles defined and in order according to the indexed list.

For a method explanation, refer to section 2.7.6, Page 32.

As can be viewed in Figure 41, a program list is present for the user to choose from, when constructing the master program.

Each sub-program is defined by the following factors:

Parameter	Description	
Probe	The probe to be used in that specific sub-program.	
Method	The method to be used in that specific sub-program.	
Program name	A meaningful name for the specific sub-program.	
Cycles	The number of cycles to be executed in that specific sub- test.	

Table 19: Chain sub program parameters



Figure 41: Program Chain Editor

The program menu varies according to the probe, and the method of stimuli intended for use, as shown on the left side of the screen.

The parameters available for program creation at this stage are as follows:

Table 20: Program Details

Column	Description
Program name	The program name (editable).

edoc

Column	Description
Probe	The probes used with this program.
Description	Under Description, click the arrow right with the type free text to describe the program. The Description icon is indicated as active (right with the text).
Update on	The date on which the latest changes were made and saved.

Table 21: Program Table

Parameter	Description	
Program list	A list of the available programs.	
	The serial number of the chained program. The serial number indicates the execution order of the programs.	
Probe	The thermode type to be used in the chained program.	
Method	Test method of the chained program.	
Program name	Name of the chained program.	
Cycles	Number of times the chained program is to be performed. To change the value, type the new one, or use the Up/Down arrows.	

The following additional functions are available:

•

Adds the program selected from the Program List to the chained list, below the selected program in the chained program list.

•

Deletes the selected program from the chained program list.

•

Moves the selected program up the index (up the list).

•

Moves the selected program down the index (down the list).

7.6.7 Next Sequence Definition

In Limits, TSL and Ramp & Hold protocols additional settings are available in the top part of program editor, allowing different options for next sequence definition during the program run:

Program Name:	Limits CS & WS Updated On: 01/0	7/2012 Randomize sequences:	Start Test: Sound
Probe:	Q-Sense Description:	Manually define the next sequence:	End Test: Sound
A contract medical systems			

Figure 42: Program Editor Next Sequence Definition

Two checkboxes are available and unchecked by default:

- **Randomize Sequences:** If this checkbox is checked the sequences order during program run will be randomized
- **Manually define the next sequence:** If this option is selected, the operator will receive the following dialog window at the end of the sequence, allowing him to specify the next sequence he would like to perform:

	Spe	cify	next sequence
	#	0	Description
		2	Trials: 1, Destination Temperature (°C): 46
		3	Trials: 1, Destination Temperature (ºC): 47
			ок)
_			

Figure 43: Manual Next Sequence definition

Note: this option is available if the program contains more than 2 sequences; otherwise this option will be dimmed.

- If both checkboxes are unchecked the sequences order will be according to their order in program editor (sequence number)



7.7 Standard Events

Standard Events are pre-defined events that occur during the test such as the onset of a stimulus or reaching the destination temperature. Events are marked by a sound.

Every event is also marked on the screen during the test by an asterisk (*).

7.7.1 Standard Event Configuration

To configure standard events in the required program, follow the steps detailed below:

1. From the **Programs** screen, select the program you want to configure standard events for and

double click to open the program editor or click on the edit program button ${}^{ extsf{W}}$

2. From the list of parameters, select the Standard Events option

ISI Min (sec)	4	4
ISI Max (sec)	6	6
Standard Events		

Figure 44: Standard Events in Program Editor

3. Click the arrow on the left to open the standard event editor.

For each event that you want to configure mark the Sound column to activate.

- 4. Click anywhere on the screen to close the Standard Events dialog.
- 5. The available standard events vary according to the test method. For a list of the available standard events according to the program type continue to the following section.

7.7.2 List of Standard Events according to Program Type

The available standard events according to program type are detailed in the table below:

Table 22: List of standard events according to program type

Program Method	Standard Events	Description
Limits	Onset	The start of each trial in the sequence
Levels	Onset	The start of each trial in the sequence
TSL	Destination	Subject response was detected
Ramp and Hold (optional)	Onset	The start of each trial in the sequence Note: In current release that is the only Standard Event available for Ramp and Hold.
	Destination	Thermode temperature reached target temperature <i>Note: Will be available in next release</i>
	End of duration	End of duration at target temperature Note: Will be available in next release



7.8 Test Instructions

In some cases, the general program instructions do not fully apply to specific programs, and unique instructions are required.

To enter test instructions, perform to the following procedure. All other (standard) tests will be assigned the standard instructions, as entered in section 10.3.2, page 129.

11. From the program list at the left, select the required program, and double-click, or click the

Edit program details button . The **Programs menu** is displayed.

12. Click the Accesses test instructions button The following screen is displayed:

Test Instruction Editor			*
Text Instruction Use Default Instruction		View	\bigcirc
Audio Instruction		Dies	
Use Default Instruction		Play	
	C	Close	\supset

Figure 45: Test Instructions Editor

- 13. To edit the text instructions,
 - 13.1. Deselect the relevant **Use default instruction** checkbox. The **Edit** button becomes enabled.
 - 13.2. Click the **Edit** button. The text editor is displayed, displaying the standard instructions.
 - 13.3. Edit the instructions as required, as described in section10.3.2.1, page129.
- 14. To replace the audio instructions,
 - 14.1. Deselect the relevant **Use default instruction** checkbox. The **Play** button becomes an enabled **Browse** button.
 - 14.2. Click the **Browse** button. The Audio instruction screen is displayed.
 - 14.3. Replace the instructions file, as described in section 10.3.2.2, page 130

7.9 Import / Export Program

Program details can be imported to and exported from MEDOC Main Station using a secure and unique format which is readable from MEDOC software only. The file extension is ats.



Note that importing and exporting program data between different software versions may result in compatibility errors and loss of data. Check database compatibility before proceeding.

In order to export Q-SENSE programs follow the procedure below:

1. Open the program list in the **Programs** screen.



- Q-Sense Technical Reference Manual
- 2. Select the programs to export. Multiple selection is possible by holding down the **Ctrl** and **Shift** buttons on the keyboard
- 3. Click **File** \rightarrow **Export**.
- 4. Select the location and name of the exported programs.
- 5. Click the **Save** button. The selected programs are saved to the selected location under the selected name.
- In order to import Q-SENSE programs follow the procedure below:
- 1. Click **File** \rightarrow **Import**.
- 2. Browse to the location of the exported programs and select the program you want to import.
- Click on the **Open** button. The imported programs are added to your program list.



MEDOC Main Station identifies and allows you to manually solve any conflicts that may arise when importing programs. For example, programs with the same name.



8 Patient Management

8.1 Patients List

The Patients screen enables you to manage patients (add, remove, edits etc.) and department information.

1. To access the patient database, click the **Patients** button at the toolbar. The following screen is displayed:

MMS - v.6.0.8.0							0 0 0
	O PATI	ENTS 👔 PI	ROGRAMS	RESULTS			
	Select a Patient						
Current Selection	Department	D	Last Name 🌑	First Name	Gender	Date of Birth	History
First Name: Undefined Last Name: Undefined	Department 1	12345678	Smith	John	Male	18/11/1982	
ID: Undefined Department: Undefined							
							Ģ
Device: Q-Sense Status:Online	Com: 😁 State:Re	st Mode Thermod	e:Q-Sense-30x30	TEC:32.00 °C HS	1:26.64 °C H52:2	6.27 °C Fan Mode	::High Safety: 🗬

Figure 46: Patients List

2. To narrow down the displayed list use the filter row.



The more criteria you define using the filter, the shorter the list that will be displayed.

3. To clear the filtered list, select the (**All**) option from the filter drop down list.



In order to add / remove fields from Patient table, see section **10.1.2**, page **106**.



It is possible to change column order by drag and drop it to required place.



The following options are available in the Patient Management screen:

()	New Patient
	Delete Patient
	Edit Patient details
or 🗎	Show or Hide Patient preview

• Current Selection control - displays the selected program information.

When you open the Patient Preview bar, the following screen is displayed:

MMS - v.6.0.8.0							
						Help	
	0 P		PROGRAMS	RESULTS			
	Select a Patien						
urrent Selection	Department	O ID	Last Name	First Name	Gender	Date of Birth Histo	ory (
st Name: John st Name: Smith : 12345678	Department 1	12345678	Smith	John	Male	18/11/1982	
partment: Department 1							
Clear							
							5
							6
	View patient of						0
	View patient of Department:		partment 1	SSN:			<u>[</u>
		_	partment 1	SSN: Gender:	O Male	e G Female	5
	Department: Last Name:	Dep	partment 1	Gender:			2
	Department:	Dep Smith John	partment 1		@ Male 18/11/198		2
	Department: Last Name:	Dep	partment 1	Gender:			2
	Department: Last Name: First Name:	Dep Smith John	partment 1	Gender:			<u>k</u>
	Department: Last Name: First Name:	Dep Smith John	partment 1	Gender:			<u>,</u>
	Department: Last Name: First Name: ID:	Dep Smith John	partment 1	Gender:			<u>k</u>
	Department: Last Name: First Name: ID:	Dep Smith John	partment 1	Gender:			<u>,</u>
	Department: Last Name: First Name: ID:	Dep Smith John	partment 1	Gender:			
	Department: Last Name: First Name: ID:	Dep Smith John	partment 1	Gender:			

Figure 47: Patient Management – Patient details preview





Patient details cannot be edited from the preview control screen

8.2 Add New Patient

To add a patient to the patients list:

15. Click the **Add** icon (). The **Patient details** screen is displayed:

MMS - v.6.0.8.0					C 🗈 S
				Utilities Help	
	ОРА		RESULTS		
	View patient de	ətails			
Current Selection	Department:	Department 1	▼ SSN:		
First Name: John Last Name: Smith	Last Name:	Luz	Gender:	Male O Fema	le
ID: 12345678 Department: Department 1	First Name:	Tzachi	Date of Birth:	14/02/1980	
Clear		12341234			
	ID:	12341234			
	History:				
Device: Q-Sense Status:Online			20-20 750-22 00 20 1		
Device: Q-Sense Statustoning	e com. State:	kest mode mermoderQ-sense	-50X50 TEC132.00 °C 1	151.20.92 °C 1152.20.58 °C F	an model might Safety:

Figure 48: Patients Editor Screen

16. Insert all the relevant patient data.



Missing or incorrect parameters are identified by an \bigotimes icon that flashes next to the relevant field. A short description of the malfunction can be displayed by holding the cursor over the relevant \bigotimes icon.



- Q-Sense Technical Reference Manual
- 17. Click the **Save** icon (
- 18. After adding and saving all the required patients, click the **Return** icon (



Adding a department can only be done by an administrator. See section 8.5 page 86.



The information displayed on the screen (field names) can be customized according to the setup defined by each user. Refer to section 10.1.2, page 106.

8.3 Remove Patient

To remove a patient from the patients list:

- 3. Click the **Delete** icon (
- 4. Click **Yes** to the prompt. The selected patient is deleted.

8.4 Edit Patient Details

To edit a patient from the patients list:

- 5. Click the **Edit** icon (
- 6. The selected patient parameters screen is open and can be edited.

8.5 Department Management

MEDOC Main Station enables you to mange (Add, Edit, Remove) your patient department.



Department management is enabled according to the authorization level of the current user. For further information, refer to section 10.1.5, Page 116.

1. From the **Utilities** menu at the top of the screen, select the **Department** option. The *Edit departments* menu is displayed:





Figure 49: Edit Department Menu

- 2. To define a new department, click the **New** button; then, type the department name.
- 3. To rename a department, select the required department; then click the **Rename** button. (Available only for System Administrators and Medical Supervisors)
- To remove a department, select the required department; then click the **Remove** button. At the prompt click the **OK** button.
 (Available only for System Administrators and Medical Supervisors)
- 5. To close the dialog, click the **Close** button.



8.6 ICD 9 Management

MEDOC Main Station enables you to mange (Add, Edit, Remove) your ICD 9 code.



ICD 9 management is enabled according to the authorization level of the current user. For further information, refer to section 10.1.5, Page 116.

1. From the **Utilities** menu at the top of the screen, select the **ICD 9** option. The *Edit departments* menu is displayed:



Figure 50: Edit ICD Menu

- To define a new ICD 9 code, click the New button; then, type the ICD 9 code and description, and click OK.
- To edit an ICD 9 code, select the required code; then click the **Edit** button. (Available only for System Administrators and Medical Supervisors). A code and description window is opened. Edit the relevant fields and click OK.
- 4. To remove a code, select the required department; then click the **Remove** button. At the prompt click the **OK** button.

(Available only for System Administrators and Medical Supervisors)

To close the dialog, click the **Close** button.

8.7 Import / Export Patient Lists

Patient details can be imported to and exported from MEDOC Main Station using a secure and unique format which is readable from MEDOC software only. The file extension is ats.



Note that importing and exporting Patient data between different software versions may result in compatibility errors and loss of data. Check database compatibility before proceeding.



In order to export Q-SENSE patients follow procedure below:

- 1. Open the patient list in the **Patient** screen.
- 2. Select the patients to export. Multiple selection is possible by holding down the **Ctrl** and **Shift** buttons on the keyboard
- 3. Click **File** \rightarrow **Export**.
- 4. Select the location and name of the exported patients.
- Click the Save button.
 The selected patients are saved to the selected location under the selected name.

In order to import Q-SENSE patients follow the procedure below:

- 1. Click **File** \rightarrow **Import**.
- 2. Browse to the location of the exported patients and select the patients you want to import.
- Click on the **Open** button. The imported patients are added to your patients list.



MEDOC Main Station software identifies and allows you to manually solve any conflicts that may arise when importing patients. For example, patients with the same unique field.



9 Results Management

The Results List screen lists the results of the recent test.

9.1 Results List

The Results screen enables you to manage Results (view, print, delete, etc.).

1. To access the results database, click the **Results** button at the toolbar. The following screen is displayed:

Current Sele	ection	Results Last Name	First Name	Method	Name 🕚	Body Site	Time	Operator	Comments O
Departament	Undefined	Smith	John	Limits	Limits CS & WS	C1 Plaht	01/07/2012	Manufacturer	
irst Name	Undefined	Smith	John	Limits	Limits CS & WS		01/07/2012	Manufacturer	
ast Name	Undefined	Smith	John	Limits	Limits CS & WS		01/07/2012	Manufacturer	
	Undefined	Smith	John	Limits	Limits CS & WS		01/07/2012	Manufacturer	
robe	Undefined	Smith	John	Limits	Limits CS & WS		01/07/2012	Manufacturer	
lethod	Undefined	Smith	John	TSL	tzahitsa	C6, Right	12/07/2012	Administrator	
rogram	Undefined	Smith	John	TSL	tzahitsa	C6, Right	12/07/2012	Administrator	
0									

Figure 51: Results List

2. To narrow down the displayed list use the filter row.



The more criteria you define using the filter, the shorter the list that will be displayed.

3. To clear the filtered list, select the (**All**) option from the filter drop down list.



In order to add / remove fields from Results table, see section 10.1.3, page 110.



It is possible to change column order by drag and drop it to required place.

The following options are available in the results management screen:



Q-Sens	e Technical Reference M	anual
	e or	Show or Hide Results preview
		Print Selected Results file / files
		Delete selected Results file from Q- SENSE Data Base

• Current Selection control - displays the selected program information.

When you open the Results preview bar, the following screen is displayed:

		O PAT	IENTS 👔 P	ROGRAMS	RESULTS				1	
		Results								
urrent Selec	ction		First Name	Method	Name	Body Site	Time	Operator	Comments	0
epartament	Department 1	T Smith	John	Limits	Limits CS & WS	S1, Right	01/07/2012	Manufacturer		
rst Name			1.022430					The second second second		_
	Smith	Smith	John	Limits	Limits CS & WS	C6, Left	01/07/2012	Manufacturer		1
		Smith	John	Limits Limits	Limits CS & WS	C6. Right	01/07/2012	Manufacturer		
					Limits CS & WS	C8, Left	01/07/2012	Manufacturer		
rogram	Q Sense 30x30 Limits Limits CS & WS Manufacturer Clear	Statistics	John John	Limits Limits Show Norma	Limits CS & WS	C6, Right	01/07/2012	Manufacturer		
obe ethod ogram signer	Limits Limits CS & WS Manufacturer	Smith Statistics	John	Limits	Limits CS & WS			Manufacturer		
obe ethod ogram signer	Limits Limits CS & WS Manufacturer	Smith Statistics 50 50 50	John	Limits	Limits CS & WS			Manufacturer		
obe ethod ogram esigner	Limits Limits CS & WS Manufacturer	Smith Statistics	John	Limits	Limits CS & WS			Manufacturer		

Figure 52: Result preview display

The following options are available:

	View Patient details
	Display Results graph preview. Available on selected methods.
\bigcirc	Display test subject details preview.
	Display the Result Body Site. Available only if a Body Site was selected.



	Delete selected results file.
\bigcirc	Print Graph
Check box	Show / Hide Normative Data display.

9.2 Import / Export Results

Test Results can be imported to and exported from MEDOC Main Station using a secure and unique format which is readable from MEDOC software only. The file extension is ats. Alternatively, results can be exported as an excel file (.xls) for further analysis.



Note that importing and exporting test results between different software versions may result in compatibility errors and loss of data. Check database compatibility before proceeding.

In order to export Q-SENSE Results to **.ats** format follow the procedure below:

- 1. Open the results list in the **Results** screen.
- Select the results to export. Multiple selection is possible by holding down the Ctrl and Shift buttons on the keyboard
- 3. Click **File** \rightarrow **Export**.
- 4. Select the location and name of the exported results.
- Click the Save button.
 The selected results are saved to the selected location under the selected name.

In order to export Q-SENSE Results to **.xls** format follow the procedure below:

- 1. Open the results list in the **Results** screen.
- 2. Select the results to export.
- 3. Click **File** → **Export to Excel**.
- 4. Select the location for the exported results.
- Click the Save button.
 The selected results are saved to the selected location.

The name of each result file is automatically generated according to the patient name, date, and time of test. Chain results are exported to a new folder containing all sub-programs of the chain

The resulting Excel file is divided into two tabs:



program.

Description: containing general patient and test data:

	А	В
1	Summary	
2	Department	Department 1
3	Patient	Smith, John
4	Probe	Q-Sense
5	Model	30x30
6	Method	Limits
7	Program	Q-sense limits test
8	Start time	18/06/2012 11:25
9	End time	18/06/2012 11:30
10	Patient Details	
11	Department	Department 1
12	ID	12345678
13	Last Name	Smith
14	First Name	John
15	Gender	Male
16	Date of Birth	18/11/1982
17	History	
18	Program Details	
19	Program Name	Q-sense limits test
20	Probe	Q-Sense
21	Description	
22	Updated on	18/06/2012 11:24
23	Body Site	
24	Body Site Main	Undefined
25	Body Site Side	Undefined
26		

Figure 53: Export Result File Opened In Excel – Description Tab Sample

Data: containing the detailed information of the test:

	A	В	С	D	E	F	G	н	1	J
1	Timestamp [msec]	Heater [C]	Tec [C]	Water [C]	PCB [C]	COVAS	Baseline [C]	LimiterMax [C]	LimiterMin [C]	Events
2	15		31.99			0				
3	30		31.99	0		0	32	50	20	
4	35		31.99	0		0	32	50	20	
5	55		31.99	0		0	32	50	20	
6	75		31.99	0		0	32	50	20	
7	80		31.99	0		0	32	50	20	
8	95		31.99	0		0	32	50	20	
9	115		31.99	0		0	32	50	20	
10	135		31.99	0		0	32	50	20	
11	155		31.99	0		0	32	50	20	
12	175		31.99	0		0	32	50	20	
13	190		31.99	0		0	32	50	20	
14	195		31.99	0		0	32	50	20	
15	215		31.99	0		0	32	50	20	
16	235		31.99	0		0	32	50	20	
17	245		31.99	0		0	32	50	20	
18	255		31.99	0		0	32	50	20	
19	275		31.99	0		0	32	50	20	
20	295		31.99	0		0	32	50	20	
21	300		31.99	0		0	32	50	20	
22	315		31.99	0		0	32	50	20	
23	335		31.99	0		0	32	50	20	
24	355		31.99	0		0	32	50	20	
25	375		31.99	0		0	32	50	20	
26	395		31.99	0		0	32	50	20	
27	410		31.99	0		0	32	50	20	
28	415		31.99	0		0	32	50	20	
29	435		31.99	0		0	32	50	20	
30	455		31.99	0		0	32	50	20	
31	465		31.99	0		0	32	50	20	
32	475		31.99	0		0	32	50	20	

Figure 54: Export Result File Opened In Excel – Data Tab Sample

In order to import Q-SENSE results, follow the procedure below:

- 1. Click **File** \rightarrow **Import**.
- 2. Browse to the location of the exported results and select the results you want to import (only for .ats format)
- Click on the **Open** button. The imported results are added to your results list.





MEDOC Main Station identifies and allows you to manually solve any conflicts that may rise when importing results. For example, results with same patient name date and time.



Saved Results cannot be edited or modified.



9.3 Print Results and Reports

MEDOC Main Station has a powerful Analysis and Results Report Generator that can be customized for user needs.

Follow these steps to print your results:

- 1. Select the relevant results file or files from the list in the results screen.
- 2. Click on the **Print** button The following screen is displayed:



Figure 55: Analysis Reports Dialog

- 3. Configure the Reports options according to your needs.
- 4. Press on the button of the Report Type you want to use from available options. Only available reports type are enabled according to your results file selection.
- 5. Report is generated. Report may have Preview or not according to your preferences.
 - 5.1. Print with Preview

Print Preview is displayed according to the active template and test method. Refer to **10.1.3**, page **110** for details regarding Printing Options. Press the **Print** icon or select one of the export options.

5.2. Print without Preview Report is sent to a connected printer. Select your Printing preferences according to the available printers and print your report.

9.3.1 Analysis Report Dialog

The Analysis Report Dialog allows Q-SENSE user to generate Reports of Q-SENSE tests.



Analysis Reports		e
Analysis Reports	Configuration	
Singles	Display Preview	
	Print Without Preview	"
	Number of Trials	9
(titateral	Start Time (sec)	0
	End Time (sec)	86400
	Min. Value Max. Value	-16
PrePost	Add Comments	
	Mode	
	Automatic	mode
	Manual me	ode
	Save Donfig	Default
		Cancel

Figure 56: Analysis Report

9.3.1.1 Analysis Report Type

Report type defines the template and content of the report and analysis to be displayed.



Report template can be edited and customized according to specific needs. Refer to section 10.1.3, page 110 for details

The following templates are available:

Report Type	Description	Restrictions
Singles	Print a single report for each selected Test Results according to active template.	None
Multiple	Print results statistic summary of selected tests.	 Test Method: TSL, Levels, Limits Selection of 2 or more result files
Bilateral	Compare test statistics of same body site of two sides (Left and Right) for same body site and test subject.	 Test Method: TSL, Levels, Limits Selection of 2 result files Identical Test Subject Identical Test Program Identical Main and Specific body site Opposite body site sides

Table 23: Report Analysis Configuration



Report Type	Description	Restrictions
Site to Site	Compare test statistics of same test subject in two different Body Sites.	 Test Method: TSL, Levels, Limits Selection of 2 result files Identical Test Subject Identical Test Program Different Main and / or Specific Site
Pre / Post	Compare test statistics of same test subject and site, performed in different date and time	 Test Method: TSL, Levels, Limits Selection of 2 result files Identical Test Subject Identical Test Program Identical Body Site
Trend	Display test statistic trend vs. test date and time.	 Test Method: TSL, Levels, Limits Selection of 2 result files or more Identical Test Subject Identical Test Program Identical Body Site
Custom	Custom report with custom analysis and restrictions.	Vary according to report analysisContact Medoc Service for Details.



Reports Examples:

				0-9	SEI	٧S	E 1	CE 4	т	DE	D٨	RT	•		Ũ		
				Υ.) I I	ľ	ΓV						
Patient irst Nan				L	ast N	me:	Smi	th			Dec	artm	ent: I	Depar	tme	nt 1	
.D.: 1 istory:	23456	78		s	ex: M	ale					-		Birth::				
Progran																	
robe: Q					thod	Lev	els			1	Desig	ner:	Admir	histrat	or		
escription		: Q-si	ense	Levels													
Genptio																	
Body Si lain Site		e	e: J					- 634-									
ain site	: Unde	mnea	5100	e: onc	enne	u 91	Jecin	c site	•								
Test Res																	
est#1	Moda	lity	Base 32		Rat 1	e	Turns 2		VG .16								
est # 1	100				1		4	2,	.10								
fest Gra	ph —																
ſest Gra	ph —																
lest Gra	59																
lest Gra	59 54																
lest Gra	59																
lest Gra	59 54 49																
	59																
ę	59 54 49 44 39 34 29	2215				2 N											
ę	59	20.5	8.1 °			2 N											
	59 54 49 44 39 34 29 24 19	20 N			6.1 ¥												
ę	59 54 49 44 39 34 29 24 19 14	2010				2 N											
ę	59 54 49 44 39 34 29 24 19	22/5															
ę	59 54 49 44 39 34 29 24 19 14 9					2 1											
ę	59 54 49 44 39 34 29 24 19 14 9 4																
ę	59 54 49 44 39 34 29 24 19 24 9 4 -1 -1 -6 -11																
ę	59 54 49 34 39 34 29 24 19 14 9 4 -1 -6					2 21		7	8 9	10	11	12	13	14	15	36	
ę	59 54 49 44 39 34 29 24 19 14 9 4 -1 -6 -11 -16		2			2.2.1	· · · · · · · · · · · · · · · · · · ·		5 9	10	11	12	13	M	15	16	
ę	59 54 49 44 39 34 29 24 19 14 9 4 -11 -6 6 -11 -16 0		2			2 2				100	11	12	13	м	15		

Figure 57: Single Results Report



Figure 58: Multiple Results Report

Nomenes Version:

C/38.2.0.0

System Details Hedo: Hein Station Version: 4.0.2.0



Analysis Report Examples:





Figure 59: Site to Site Results Report



Figure 61: Trend Results Report



Figure 60: Pre / Post Results Report

16/07/2012 Smith John isst Results: Image: State Sta	Body Si 6, 8, 8, 8, 8, 8, 8,	te	Limits Limits Limits	Left N/A N/A 30.62	Right 29.42 34.64 N/A	(ND AVG + 2*ST N/A N/A	(ND AVG + 2.5*STI N/A
Nodality Body Site Method Left Right Righ	6, 6, 8, 8, BodySi 6,	_	Limits Limits Limits	N/A N/A 30.62	29.42 34.64 N/A	(ND AVG + 2*ST N/A N/A	(ND AVG + 2.5*STI N/A
Nodality Body Site Method Left Right Righ	6, 6, 8, 8, BodySi 6,	_	Limits Limits Limits	N/A N/A 30.62	29.42 34.64 N/A	(ND AVG + 2*ST N/A N/A	(ND AVG + 2.5*STI N/A
10 10 10 22.42 10 DVA VE + 2.5351 IND AVE + 2.5351 V5 C6 Limits 1VA 29.42 10 AVA N/A V5 C6 Limits 30.62 N/A N/A N/A V5 C6 Limits 24.92 N/A N/A N/A V5 C6 0 0 0 0 0 0 V5 C6 0 0 0 0 0 0 V5 C6 0	6, 6, 8, 8, BodySi 6,	_	Limits Limits Limits	N/A N/A 30.62	29.42 34.64 N/A	N/A N/A	N/A
WS C6, Lumits N/A 34.64 N/A N/A SS C6, Lumits 30.62 N/A N/A N/A WS C6, Lumits 34.92 N/A N/A N/A Modality Bodysite Z>2 Z>2.5 S S C6, 0 0 YS C6, 0 0 0 0 S	6, 8, 8, BodySi f	he	Limits Limits	N/A 30.62	34.64 N/A	N/A	
S C6, Limits 30.62 N/A N/A N/A Analysis Limits 34.92 N/A N/A N/A N/A Modality BodySite Z>Z Z>Z5. C6, 0 0 VS C6, 0 0 0 0 0 0 VS C6, 0	8, 8, BodySi ¹ 6,	he	Limits	30.62	N/A		
WS CS, Limits 34.92 N/A N/A N/A Madping Modality BodySite Z>2 Z>2.5 X </td <td>8, BodySit</td> <td>be a</td> <td>Limits</td> <td>34.92</td> <td></td> <td>I N/A</td> <td>N/A</td>	8, BodySit	be a	Limits	34.92		I N/A	N/A
Nodality BodySite Z>2 Z>2.5 S C6, 0 0 WS C6, 0 0 VS C6, 0 0 Sobt_Evels and Limits tests are abnormal.both at the inplicit and at the left side at a specific provide the same site, either reft, and at the other side one of the rest is abnormal. Rot Evels are abnormal store side and the same site, either right or left. Left 2 tests are abnormal site either right or left. Left 2 tests are abnormal site either right or left.	6,	he			N/A		
Value Meaning b Both Levels and Limits tests are abnormal, both atthe inith side and atthe left side at a specific b Both Levels and Limits are abnormal, both atthe inith side and atthe left side one of the test is abnormal. clstpl 2 tests are abnormal at one side (right or left), and at the other side one of the test is abnormal. clstpl 2 tests are abnormal at one side and the same site, either right or left. Left 2 tests are abnormal but at different sites. L1 tests is bnormal to use site either right or left.	8,		0	0	_		
is Both Levels and Limits tests are abnormal, both arther inhibitide and at the left side at a specific or both Levels and Limits are abnormal at one side (rightorieft), and at the other side one of the test is abnormal. Englst 2 tests are abnormal at one side and the same site, either right or left Left 2 tests are abnormal at one side and the same site, either right or left. It tests is abnormal at one site either right or left.	8,		0				
Both Levels and Limits are abnormal at one side (rightor left), and at the other side one of the tests is abnormal. Right 1 Zests are abnormal at one side and the same site, either right or left. Left Zests are abnormal but at different sites. Itests abnormal at one side either right or left.	Levels and Li	mitstest	s are abnor			e right side and att	the left side at a specific
Right 2 tests are abnormal atone side and the same site, either right or left Left 2 tests are abnormal but at different sites. 1 testis abnormal atone site either right or left.	Levels and Li	mits are	abnormal at	tonesid	e (right	corleft), and at the	otherside one of the
2 tests are abnormal but at different sites. 1 test is abnormal at one site either right or left.		nalaton	e side and th	ne same	site, e	ither right or left	
1 test is abnormal at one site either right or left.	ts are abnorm	albutat	t different sit	ter			
) No abnormalities.	tis abnormal					-	-
	onormalities.						
		Levels and Li Levels and Li is abnormal ts are abnorm ts are abnorm tis abnormal	Levels and Limits test Levels and Limits are is abnormal ts are abnormal at on ts are abnormal but a tis abnormal at one s	Levels and Limits tests are abnor Levels and Limits are abnormal at is abnormal at one side and th ts are abnormal at one side and th ts are abnormal at one side either rig tis abnormal at one side either rig	Nu Levels and Limits tests are abnormal, bot Levels and Limits are abnormal at one sid is abnormal ts are abnormal at one side and the same ts are abnormal but at different sites. Its abnormal at one site either right or left	Neaning Levels and Limits tests are abnormal soft at the Levels and Limits are abnormal at one side (right is abnormal to the side and the same site, e ts are abnormal but at different sites. It a abnormal but at different sites.	Nearing Levels and Limits tests are abnormal, both atthe right side and att Levels and Limits are abnormal at concesside (rightor left), and atthe is abnormal to are abnormal at one side and the same site, either right or left ts are abnormal but at different sites. Its abnormal at one site either right or left.

Figure 62: Custom Report Results





Medoc Main Station Version:	6.0.12.0	Firmware Version:	CTSA 2.0.0 (Demo Mode)

Figure 63: Bilateral Results Report

9.3.1.2 <u>Reports Configuration</u>

Report configurations allow defining basic reports settings and display options.

The following configurations are available:

Table 24: Repo	rt Analysis	Configuration
----------------	-------------	---------------

Parameter	Description
Display Preview	Displays report preview before sending to print.
Print Without Preview	Send report to print without a preview.
Mode	Automatic Mode – automatically calculate test graph display Manual Mode – manual settings of graph display
Number of Trails (Manual Mode)	Determine how many trails to display for trail based tests. Available for Limits and Levels.
Start Time (Manual Mode)	Determine the display of test graph start time for time based tests. Available for TSL.



End Time (Manual	Determine the display of test graph end time for time based tests.
Mode)	Available for TSL.
Min. Value. (Manual	Determine display of the test graph minimum temperature.
Mode)	Available for TSL.
Max. Value. (Manual	Determine display of the test graph maximum temperature.
Mode)	Available for TSL.
Add Comments	Allow adding comments for the generated report. Available for all report type beside Singles.



9.3.1.3 <u>Custom Reports Configuration</u>

- 1. Save the custom report you would like to use
- 2. Go to Settings --> Software Settings --> Report Options tab
- 3. Select the required template from the left column and click Copy Template



Figure 64: Custom Report Configuration

- 4. A copy of the Multiple template will be added to the list (can be renamed).
- 5. Select the copy and click Edit Template.





Figure 65: Custom Report Configuration

- 6. The report designer window will open. Go to the file menu and select "open report". Select the report template saved in step 1.
- 7. After the report template opens, close the report designer window with the X on the right corner. There is no need to save.
- 8. Back in the report options window, select the copy again and click the >> button (move right).



Figure 66: Custom Report Configuration

9. Click Apply and close the report options. This report will now replace the existing report of the selected type



10 System Configuration

Q-SENSE system configuration is performed via three different setting menus:

- 1. Software Settings see section **10.1**, Page **105**.
- 2. Hardware Settings see section **10.2**, Page **123**.
- 3. Test Settings see section 10.3, Page 125.



The available system configuration options depend on user authorization level. Refer to section 10.1.5.2, page 115.

10.1Software Settings

The software setting menu includes communication setup, user interface and preferences and user management (authorization levels).

To access the software settings screens, from the **Settings** option in the menu bar; select **Software Settings**.

The following screen is displayed:

Software Settings	۲
Q-Sense Authorization Report Options Results Fields Patient's Fields	
Q-Sense COM Port: COM10	
Disconnect	
Calibration COM Port: COM1 V	
Calibration Com Part.	
Check	
ОК	Apply Cancel

Figure 67: Software Settings Dialog

The software settings menu includes the following tabs (additional tabs may be visible according to license):

- Q-Sense
- Authorization
- Report Options
- Results Field
- Patient's Fields



10.1.1 Communication

The General Software Settings screen enables you to configure general software, database and communication settings (see Figure 67 above). Configuration options are described in the table below.

Parameter	Description
Q-Sense COM Port	 Set the COM port to which the Q-SENSE system is connected to the computer: Auto Detect – Automatically scan all available ports. Simulator – Work in Demo Mode (MEDOC Main Station will not connect to Q-SENSE system. COM X – Set the specific port the Q-SENSE is connected to. Click Connect to establish communication.
Calibration COM Port	Set the COM port to which the Thermometer is connected to. This is only required during the Automatic Calibration Procedure. Click the Check button to test communication between the computer and YSI thermometer. Green light indicates "pass", and Red indicates "fail". For further details refer to Automatic Calibration section in the Service Manual.

Table 25: Software Settings - General

10.1.2 Patient Fields

Configure the available fields for entering patient details and determine which fields will be displayed in the patient list. The configuration of patient fields affect all users (see section 10.1.5, page 116), however every user can determine what fields to see when viewing the patient list.

To access Patient Field settings: from the top menu bar, select **Settings** and then **Software Settings**.



Figure 68: Open Software Settings

In the Software Settings window that opens, select the **Patient Fields** tab.



Figure 69: Software Settings - Patient Fields tab selection

The following window is displayed:



Software Settings Communication Users Export Report Options Results Fields Patient Fields	
Unique Fields Name Input Mask ID Driver License Social Security Number	Restore
Patient Card Fields	Patient Table Fields
ОК	Apply Cancel

Figure 70: Software Settings - Patient Fields

Patient Fields consist of the following types:

Unique Fields – The selected fields are used for conclusive patient identification (more than one can be used). Any other field can be repeated in various patients (for example, John can appear as first name for more than one patient).

Patient Card Fields – the selected fields will be available when entering patient details.

- 1. **Mandatory fields** these fields will be mandatory when creating a new patient card and entering the patient details. It will not be possible to save the patient card if one of the mandatory fields is not filled.
- 2. **Optional fields** built in commonly used fields that can be added / removed from the patient card.
- 3. **User Defined fields** similar to the optional fields, these customizable fields can be added or removed from the patient card. Rename the fields as needed.

Patient Table Fields – these are the fields will show in the patient list.

10.1.2.1 Add/Remove Patient Field

To define and add Patient Card Fields to the **Patient Details** dialog:

- 1. From the **Patient Card Fields** list, select the required field or add field name.
- 2. Click the **Apply** button.

To add Patient Fields to **Patient Table**:

1. Select field to be added from the **Patient Card Fields** list.



- 2. Click the _____ button to add field to **Patient Table Fields**.
- 3. Click the **Apply** button.

To remove Patient Fields from **Patient Table**:

- 1. Select field to be removed from the **Patient Table Fields** list.
- 2. Click the button to remove field from **Patient Table Fields**.
- 3. Click the **Apply** button.

To restore the default Patient Fields:

- 1. Click the **Restore** button to restore default **Patient Fields**.
- 2. Click the **Apply** button.




10.1.3 Results Fields

This dialog configures the displayed fields in the Results list.



Figure 71: Results Fields Configuration Screen

Table 26: Patient's Field

Parameter	Description
Patient Fields	Available fields from patient card
Program Fields	Program related fields
Test Results Fields	Test related fields

To add field to the Results Table:

- 1. From **Optional Fields** list, select (check) field to be added to Results Table.
- 2. Click the **Apply** button.

To remove field from the **Results Table**:

- 3. In **Ofptional Fields** list, uncheck field to be removed from Results Table.
- 4. Click the **Apply** button.

To restore the default **Results Fields**:

- 1. Click the **Restore** button to restore **Results Fields** defaults.
- 2. Click the **Apply** button.

10.1.4 Report Options

MEDOC Main Station enables to generate test reports, which contain the test results and data according to the system definitions when the test was saved.



Each test method has a unique report template, which is used when the user selects to print the test results.

O Sho	ed Report W Body Site Print	User Logo Show User L	\equiv			
Ramp	mplate	np	City Timple Edit Timple Renove	Te. Ran	np an Singles np an Singles ses Singles	Met Ramp an Pulses-TSA Pulses-P

The Report Options tab enables users to manage their reports and edit reports templates.

Figure 72: Report Options tab

Available Options:

Parameter	Description
User Logo	Enables and configure display of customized user Logo to the report printouts
Report Templates Options*	 Enables editing Detailed Report templates: 1. Manage Report templates data base 2. Edit custom made reports 3. Manage Active Templates list * Available Only when Detailed Report Option is selected

10.1.4.1 Detailed Report

Detailed Report option contains all required test information and varies according to test method.

Every test method may have more than one template in the Q-SENSE data base, but only one template is used each time when printing a report. The template type is selected by MEDOC Main Station according to the selected test method.



Q-SENSE user may select the active template for each method from the templates data base.

Templates can be edited and modified according to custom needs.

iser: Adm				F TECT	REPO	рт		36	ense
Ger: Adri	1130 000	0	_CENC						
Patient D	oetails —	Ý	-SENS		NLF U				
First Nam	e: John		Last Name:	: Smith		oartment:			1
I.D.: 12 History:	2345678		Sex: Male		Dat	e of Birth:	18/11/	1982	
Program		000	Method: Lev	-1-	Dl-		-1-4 -4		
	-Sense - 3 Name: Q·			eis	Desig	ner: Admi	nistrato	or	
Description		501150 20							
Dealer Cit	_								
- Body Site		d Side	Undefined Sp	acific Site					
Main site.	ondenne	id sider	ondenned of	sectific side:					
Test Res	ults								
	Modality			Turns AVG]				
Test # 1		Baselin 32	e Rate	Turns AVG 2 27.16]				
]				
]				
Test # 1	CS]				
	CS]				
Test # 1	cs oh]				
Test # 1	oh]				
Test # 1	cs]				
Test # 1	cs]				
Test # 1	cs]				
Test # 1	cs	32	1						
Test # 1 Test Grap	CS		1						
Test # 1 Test Grap	CS	32	1						
Test # 1 Test Grap	CS	32	1						
Test # 1 Test Grap	CS	32	1						
Test # 1 Test Grap	CS	32	1						
Test # 1 Test Grap	CS	32	1						
Test # 1 Test Grap	CS	32	1						
Test # 1 Test Grap	CS	32	1						
Test # 1 Test Grap	CS 	32	1			2 13		15	
Test # 1 Test Grap	CS 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 40 59 54 50 50 50 50 50 50 50 50 50 50	32		2 27.16	9 10 11	22 13		13	
Test # 1 Test Grap	CS 29 54 40 54 40 54 40 54 40 54 40 54 54 40 54 40 54 54 54 54 54 55 54 55 55 55	32		2 27.16	9 20 11	12 13	14	15	16

Figure 73: Detailed Report example



Detailed Report		w User Logo				
Show Body Site			<i>10C</i>			
Simple Print		advanced med	cal systems			
rt Templates Options						
a Base Templates Options				Active Template		
a base remplate	_		_	Reave rempiate		
Template Name	Туре	Method	Custom 🔵	Template Name	Туре	Method
Ramp and Hold	Singles	Ramp and Hold-TSA		Ramp and Hold	Singles	Ramp and Hold-TSA
Ramp and Hold	Singles	Ramp and Hold-Pat		Ramp and Hold	Singles	Ramp and Hold-Pathway
Pulses	Singles	Pulses-TSA		Pulses	Singles	Pulses-TSA
Pulses	Singles	Pulses-Pathway		Pulses	Singles	Pulses-Pathway
	Cimelan	Search-TSA		Copy Template Search	Singles	Search-TSA
Search	Singles					
Search Search	Singles	Search-Pathway		Search	Singles	Search-Pathway
	-	Search-Pathway Limits-TSA		Edil Template Search Limits	Singles Singles	Search-Pathway Limits-TSA
Search	Singles	,			-	,
Search Limits	Singles Singles	Limits-TSA		Edit Template Limits	Singles	Limits-TSA
Search Limits Limits	Singles Singles Singles	Limits-TSA Limits-Pathway		Edil Template Limits Limits	Singles	Limits-TSA Limits-Pathway
Search Limits Limits Levels	Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA		Edit Template Limits Remove Limits Levels Levels	Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA
Search Limits Limits Levels Levels	Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway		Linits Linits Linits Levels Levels	Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway
Search Limits Limits Levels Levels TSL	Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA		Limits Limits Levels TSL	Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA
Search Limits Limits Levels Levels TSL TSL	Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA TSL-Pathway		Conference Pressee	Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA TSL-Pathway
Search Limits Limits Levels Levels TSL TSL Suprathreshold	Singles Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA TSL-Pathway Suprathreshold-TSA		Contention Content	Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA TSL-Pathway Suprathreshold-TSA
Search Limits Limits Levels Levels TSL TSL Suprathreshold Suprathreshold	Singles Singles Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA TSL-Pathway Suprathreshold-TSA Suprathreshold-Pat		Conception Conceptication Conception Conception Conception Conception Concept	Singles Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-TSA TSL-TSA TSL-TSA Suprathreshold-TSA Suprathreshold-Pathway
Search Limits Limits Levels Levels TSL TSL Suprathreshold Suprathreshold Staircase	Singles Singles Singles Singles Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA TSL-Pathway Suprathreshold-TSA Suprathreshold-TSA Suprathreshold-TSA		Contention Conten	Singles Singles Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-TSA TSL-TSA TSL-Pathway Suprathreshold-TSA Suprathreshold-TSA Suprathreshold-Pathway Staircase-TSA
Search Limits Limits Levels Levels TSL Suprathreshold Staircase Staircase	Singles Singles Singles Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-Pathway TSL-Pathway Suprathreshold-TSA Suprathreshold-Pat Staircase-TSA Staircase-Pathway		La Lancia Transformation Limits Levels TSL TSL TSL TSL TSL Suprathreshold Suprathreshold Staircase	Singles Singles Singles Singles Singles Singles Singles Singles	Limits-TSA Limits-Pathway Levels-TSA Levels-Pathway TSL-TSA TSL-Pathway Suprathreshold-TSA Suprathreshold-Pathway Staircase-TSA Staircase-Pathway

Figure 74: Detailed Report Options

The following actions are available in the Detailed Report:

Table 28: Report Options

Parameter	Description	
Show Body Site	When Show Body Site option is checked, the name of the body site that was used during a test (Main Site, Side and Specific Site) will be displayed in the final report.	
User Logo	User Logo button may be used in order to import a custom made logo that will appear on the Q-SENSE reports.	
	Logo preview will appear on the right side when custom logo is in use.	
	• Acceptable formats: .GIF; .JPG; .PNG; .BMP only.	
Reports data base management	Enables managing template options, Copy Template , Edit Template , Remove template and set Active template (>>).	
	Active Template is the template that is used for each test method and type when pressing the Print button and selecting the report type to be print.	



10.1.4.1.1 Data Base Templates

The data base templates option displays the available templates from the Q-SENSE data base. Each test method can have several templates according to the user needs.

Data base templates are characterized by the following parameters:

- 1. Template Name
- 2. Type determine the report type the template refers to.
- 3. Method determine the test method the template support.
- 4. Custom (checked / unchecked) determine whether the template was made by Manufacture or User.

The original Q-SENSE data base contains a template for each of the Q-SENSE methods. These templates cannot be customized, edited or removed from the data base.

Editing Data Base template:



Editing a report template is an advanced action that requires basic knowledge of data base structure and programming.

Q-SENSE enables the editing of custom made reports only. For editing a report, copy a template from the existing data base, rename it and then edit it. The following steps describe the process of editing a report:

- 1. Select a template for editing from the Data Base Template list.
- 2. Press the **Copy Template** button. the following screen is displayed:

Enter report template name	8
Levels Copy	
ОК	Cancel

Figure 75: Custom made Report Name dialog

- 3. Enter a name for the new report you create and press the **OK** button.
- 4. Select the report name you have entered from the Report data base list.
- 5. Press **Edit Template** button (only custom made templates can be edited).



- Q-Sense Technical Reference Manual
- 6. The Report Editor screen is displayed:



Figure 76: Report Editor Screen

- 7. Edit Report template according to your needs.
 - 7.1. All report parameters may be edited.
 - 7.2. All existing data may be accessed through the Q-SENSE data base.
- 8. Press the **Save** button when you complete editing your template.
- Use the >> button in order to set your template as active.
 Your template will replace the Active template of the selected method.
- 10. Use the **Remove** button in order to delete a template from the Q-SENSE data base (only custom templates can be deleted).



A special report design can be prepared upon request by Medoc.

Setting Active Template:

- 1. Select the template you want to use as Active template.
- Use the >> button in order to set selected template as active. The selected template will replace the active template of the selected method.



10.1.5 User Management

User management enables you manage and restrict the access of specific users to certain features of the software. For example, restrict a test operator from accidently deleting patient records.

There are four available authorization levels: Software Administrator, Biomedical Engineer, Medical Supervisor and Operator. Refer to Table 29 for a list of allowed and restricted actions according to user authorization level.

To access User Management panel: from the top menu bar, select **Settings** and then **Software Settings**.



Figure 77: Open Software Settings

In the Software Settings window that opens, select the **Users** tab.



Figure 78: Software settings users tab selection

The following window will be displayed:

🔍 Software Settings 🛛 🛞
Communication Users Export Report Options Results Fields Patient Fields
op1 User Name Last Name First Name Change Role: Operator New Password: Confirm New Password: Onfirm New Password: Show hidden users
ОК Арлу Cancel

Figure 79: Software Settings Users Tab



Q-Sense Techi 10.1.5.1	nical Reference Manual Add\Remove Users
1. Click the "a	add user button" at the bottom of the user list
	Communication Users Export Report Options Results Fields Patient Fields
	op1 User Name Last Name First Name Change Role: Operator New Password: Confirm New Password:
	Cancel

Figure 80: Software Settings Add New User

- 2. Enter the following information:
 - 2.1. User Name this will the name the user will use when logging in to Medoc Main Station.
 - 2.2. Last Name and First Name these will be used for display and to identify actions performed by this user
 - 2.3. Role Select a role or authorization level for this user. Note that a user can only create users with roles equal or lower than his/hers own authorization level. Refer to Table 29 for a list of roles and their allowed and restricted actions.
 - 2.4. Password choose a password for this user which will be required when logging in to Medoc Main Station.
- 3. Click Apply to confirm
- 4. Continue to add users as needed by clicking the "add new user" button and entering the relevant information (repeat steps 1-3).
- 5. Click OK to end and close the user management panel.



Medoc Recommendations:

- 1. Create Operator level users for each site
- 2. Create a Biomedical Engineer user for basic troubleshooting and calibration at the sites if needed.
- 3. Change the password to the default admin user



Communication Users Export Operator Op1	Report Options Results Fields Patient Fields User Name Op1 Last Name Site First Name Operator
	Change Role: Operator
(+) (×)	Cancel

Figure 81: Software settings filled

To remove a user from the list, select the user and click the "delete selected user" button $\displaystyle{\swarrow}$

10.1.5.2 <u>Authorization Levels</u>

Users must be assigned to one of the following Authorization Levels or Roles, which will determine which actions are permitted and which actions are restricted.

- 1. Administrator can perform all actions. Administrators have full control over the patients, programs and results, as well as software and hardware configuration (within the capabilities of Medoc Main Station).
- Biomedical Engineer intended for a user who may handle the technical operations such as Thermode calibration and certain hardware and software configurations. Biomedical Engineers cannot delete patients or results.
- 3. Medical Supervisor intended for a user who may handle the medical or clinical operations such as program design. A medical supervisor will have limited access to hardware configurations.
- 4. Operators intended for user who will only be operating the system, performing common tasks such as adding patients to the list and performing a tests. Operators cannot make any changes to software and hardware configurations and have limited access to patient, programs and results.

Table 29 below lists the main allowed and restricted actions for each role or Authorization Level.



Screen	Function	Software Administrator	Biomedical Engineer	Medical Supervisor	Operator
Patient	Add Patient	\checkmark	\checkmark	\checkmark	\checkmark
	Edit Patient	\checkmark	\checkmark	\checkmark	\checkmark
	Delete patient	\checkmark	X	\checkmark	X
Programs	Add Program	\checkmark	\checkmark	\checkmark	X
	Edit Program	\checkmark	\checkmark	\checkmark	X
	Delete Program	\checkmark	\checkmark	\checkmark	X
	Duplicate Program	\checkmark	\checkmark	\checkmark	X
	Program Preview	\checkmark	\checkmark	\checkmark	\checkmark
Results	View Graph	\checkmark	\checkmark	\checkmark	\checkmark
	View Program Details	\checkmark	\checkmark	\checkmark	\checkmark
	Delete	\checkmark	X	\checkmark	X
	Print	\checkmark	\checkmark	\checkmark	\checkmark
Test	View Patient Details	\checkmark	\checkmark	\checkmark	\checkmark
	View Program Details	\checkmark	\checkmark	\checkmark	\checkmark
	Clear	\checkmark	\checkmark	\checkmark	\checkmark
	Go to Test	\checkmark	\checkmark	\checkmark	\checkmark
Tool bar - File	Export	\checkmark	\checkmark	\checkmark	\checkmark
	Import	\checkmark	\checkmark	\checkmark	\checkmark
	Export to Excel	\checkmark	\checkmark	\checkmark	\checkmark
	Print	\checkmark	\checkmark	\checkmark	\checkmark
	Log Off	\checkmark	\checkmark	\checkmark	\checkmark
	Exit	\checkmark	\checkmark	\checkmark	\checkmark
	Switch device	\checkmark	\checkmark	\checkmark	\checkmark
Tool bar -	Test	\checkmark	\checkmark	\checkmark	\checkmark
View	Patient	\checkmark	\checkmark	\checkmark	\checkmark

Table 29: Authorization Levels (main actions)



Screen	Function	Software Administrator	Biomedical Engineer	Medical Supervisor	Operator
	Programs	\checkmark	\checkmark	\checkmark	\checkmark
	Results	\checkmark	\checkmark	\checkmark	\checkmark
Tool bar - Settings	Software Settings				
	<u>Q-Sense</u>				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	\checkmark	\checkmark	\checkmark	\checkmark
	Patient's fields				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	\checkmark	\checkmark	\checkmark	X
	Results fields				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	\checkmark	\checkmark	\checkmark	X
	Report options				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	\checkmark	\checkmark	\checkmark	X
	Authorization				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Add	\checkmark	\checkmark	\checkmark	X
	Edit	\checkmark	\checkmark	\checkmark	X
	Delete	\checkmark	\checkmark	\checkmark	X
	Hardware Settings				
	<u>General</u>				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Enable thermode	\checkmark	\checkmark	\checkmark	X
	Rest Mode temp.	\checkmark	\checkmark	\checkmark	X
	% duration	\checkmark	\checkmark	\checkmark	X



Screen	Function	Software Administrator	Biomedical Engineer	Medical Supervisor	Operator
	Safety				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	X	X	X	X
	Thermode				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	\checkmark	\checkmark	\checkmark	X
	General	\checkmark	\checkmark	\checkmark	X
	Edit LUT tables	X	X	X	X
	Manual calibration	\checkmark	\checkmark	\checkmark	X
	Automatic calibration	\checkmark	\checkmark	\checkmark	x
	Add new Thermode	\checkmark	\checkmark	\checkmark	X
	Test settings				
	Instruction Configuration				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	\checkmark	\checkmark	\checkmark	X
	Body sites				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	\checkmark	\checkmark	\checkmark	x
	status				
	Demo	\checkmark	\checkmark	\checkmark	\checkmark
	On-line	\checkmark	\checkmark	\checkmark	\checkmark
	Department				
	Add Department	\checkmark	\checkmark	\checkmark	\checkmark
	Delete Department	\checkmark	X	\checkmark	X



Screen	Function	Software Administrator	Biomedical Engineer	Medical Supervisor	Operator
	Rename Department	\checkmark	X	\checkmark	X
	Normative data				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Edit	\checkmark	\checkmark	\checkmark	X
	Delete	\checkmark	\checkmark	\checkmark	x
	Import/Export	\checkmark	\checkmark	\checkmark	X
	Black Box				
	<u>Software</u>				
	View	\checkmark	\checkmark	\checkmark	\checkmark
	Export	\checkmark	\checkmark	\checkmark	\checkmark
	Q-SENSE recovery	\checkmark	\checkmark	\checkmark	X
Tool bar - Help	About	\checkmark	\checkmark	\checkmark	\checkmark



10.2Hardware Settings

This menu enables you to set up and test the parts of the system hardware. This can include setting up parameters such as, pulse duration and polarity; timeout limitations; detection delay, etc.

19. From the **Settings** menu, select the **Hardware Settings** option.

The following screen is displayed:

Hardware Settings Seneral Safety Thermode	6
General Thermode Enabled Rest temperature (°C) Ramp Timeout (% of duration)	PRU Diagnostics YES 🔌 NO 🍽
Status Safety status TEC status HS1 status HS2 status HS2 status	Fan Mode High Medium Low Manual Fan disabled
	OK Asstr Cancel

Figure 82: Hardware settings – General tab

The following options are available:

Table 30: Hardware Settings Tab Sheets

Parameter	Description	
General	For setting default hardware parameters.	
Safety	For displaying safety parameters.	
Thermodes	For setting thermode parameters and calibration.	

10.2.1 General

The following options are available in the General tab:

Table 31: Hardware Settings – General tab options

Parameter	Description
Reset temperature (°C)	Sets the temperature in which the system remains, when it is on but not in use.



Parameter	Description
Ramp timeout (% of duration)	If Ramp temperature continues to be detected beyond this setting, an error message is displayed. The test is not aborted.
Thermode enabled	If the use of probe type is not possible (for example, because it is broken) its usage can be disabled by deselecting the checkbox.

10.2.2 Safety

The safety tab presents the software temperature time limitations which are allowed according to the program safety regulations. In addition, is presents a list of the self-tested parts, and built-in test (BIT) parameters.

The values in this tab can only be read, and cannot be edited.



Figure 83: Hardware settings – Safety tab

10.2.3 Thermode

The thermode tab presents all the thermode data and calibration tables. Refer to the Thermode Management section in the Service Manual for further explanation.

Additional available functions are:

- Thermode calibration.
- Thermode management.



Hardware Settings		e
General Safety Thermode		
Q-Sense thermode		
Q-Sense Thermodes	General TEC1 TEC2	HeatSink 1 HeatSink 2 Auto. Cal. Man. Cal. Cal. Check
Default Thermode Q-Sense Thermode	Name	Default Thermode
	Serial Number	
	Thermode Type	AirTSA
	Model	Large30x30
	Last Modification	6/19/2012 4:27:07 PM
$+\times$		
Load from file Save to file		Set as Corrent Load from TCU
		OK Assy Cancel

Figure 84: Hardware settings – Thermode tab

10.3Test Settings

The Test Settings menu includes settings and preferences which affect how the tests are performed and displayed.

To access the test settings options, from the top menu bar, select **Settings** and then **Test Settings**.

The following settings can be accessed from the Test Settings menu:

<u>Test configuration</u> – change how the test is performed and displayed

<u>Instructions Configuration</u> Instructions Configuration – modify text and audio test instructions for the different program types

Body Sites – customize list of body sites.

10.3.1 Test Configuration

Control test behavior and appearance.

To access the Test Configuration settings: from the top menu bar, select **Settings** and then **Test Settings**. In the Test Settings window that opens, select the **Test Configuration** tab. The following window is displayed:



🕲 Test Settings
Test Configuration Instructions Configuration Body Sites
Pre-Test Configuration
Low Temperature Margin (°C):
Success Time (sec):
Test Configuration
Low Destination Margin (°C): 0.1 High Destination Margin (°C): 0.1
Post-Test Configuration
Post Test Time (sec):
Test Display Configuration
Show Normative Data
Default Trial Scale: 16 Default Time Scale (sec): 100
Temperature Scale Configuration
min 15 max 55 Min. Temp Unit (°C) 5
Temperature Limits Pain rating Limits
ОК Арру Сапсе

Figure 85: Test Configuration Settings

The following options are available:

10.3.1.1 <u>Pre-test settings</u>

Table 32: Pre-test parameters

Parameter	Description
Low temperature marginThe bottom margin for a stabilized temperature allowed successful pre-test.	
High temperature margin	The top margin for a stabilized temperature allowed for a successful pre-test.
Success time	The stabilized temperature duration, required to render a successful pre-test.
Maximum time	The maximal time allowed for stabilizing trials (during the pre-test sequence).

10.3.1.2 <u>Test Configuration</u>

Table 33: Test configuration parameters

Parameter	Description
Low destination margin	The bottom margin, above which the system declares that it has ascended a high destination temperature.
High destination margin	The top margin, below which the system declares that it has descended to a low destination.



10.3.1.3 Post Test Configuration

Table 34: Post-test parameters

Parameter	Description
Post-test time	Post-test temperature stabilization is required by safety standards. This setup is used to add time for post-test monitoring.
Add comments	When the Add Comments option is checked, a dialog box for adding comments to the test open when a test ends. These comments will be displayed in the final report.
Save automatically	When Save Automatically option is checked, results will be saved automatically at the end of each test.

10.3.1.4 <u>Test Display Configuration</u>

Table 35: Test Display configuration

Parameter	Description
Show COVAS	When checked, the COVAS output graph is displayed, with the stimulation output graph by default. Can also be changed per test, from the Test View screen.
Show Normative Data	When checked, the Normative Data information graph is displayed, with the stimulation output graph by default. Can also be changed per test, from the Test View screen.
Default trial scale	Sets the trial scale of the output graph while using the Limits method. Can also be changed per test, in any Test View screen.
Default time scale	Sets the time scale of the output graph. Can also be changed per test, in any Test View screen.

10.3.1.5 <u>Temperature Scale Configuration</u>

Table 36: Temperature Scale Configuration

Parameter	Description
Temperature Scale Range	Sets the default limits for the test graph.
Minimum Temp Unit	Sets the minimum resolution of the grid



10.3.1.6 <u>Temperature Limits</u>

Parameter	Description
Temperature Limits	Sets the minimum and maximum temperature that is allowed during a tes. This parameter does not replace the safety limitation of the Q-Sense; it allows limiting the stimulus temperature to a specific range, according to unique requirements.
	In case the temperature violates the temperature limits, the following actions are taken by the Q-Sense system:
	Limits Protocol: stimulus is terminated and temperature returns to baseline temperature, and then continues to the next trial. The terminated trial is marked with an asterisk. The terminated trial will be taken into consideration when calculating test statistics.
	TSL Protocol: the temperature does not exceed the allowed range, according to the temperature limiter. If the temperature reaches the limiter the direction of the stimulus is changed automatically and the temperature is marked on the screen. The terminated trial will be taken into consideration when calculating test statistics.
	Ramp & Hold Protocol: upon verifying a program the test parameters are checked according to the current temperature limiter. In case a temperature limit of a test parameter is violated, an error message indicating the violated parameter is displayed.

Table 37: Temperature Limits



10.3.2 Instructions Configuration

The Instructions Configuration option sets the default files that will be available to provide information concerning the various test methods.

Figure 86: Instructions Configuration

Two formats are available:

- Text a text (written) description.
- Audio a vocal description, in wav, wma or mp3 format.

10.3.2.1 <u>Text</u>

To create text instructions for a test method:

- 20. Choose the desired test type -Limits test.
- 21. Click the **Text** button. The following screen is displayed:





Figure 87: Test Instructions Editor

- 22. Enter the required text.
- 23. Click the **Save** button.
- 24. Close the window.

For editing Text instructions use the text editor.

10.3.2.2 <u>Audio</u>

To enter vocal instructions for a test method:

- Create an audio file of the instructions in one of the following formats: WAV, WMA, MP3
- 2. Choose the desired test type -Limits test.
- 3. Browse to the desired folder, and select the audio file.
- 4. Click the **Open** button.



Figure 88: Audio save button

- 5. Click the **Save** button.
- 6. Close the window.

The following functions are available:

- Play Plays the audio file.
- Stop Stops the playback. **Play** will start from the beginning.

edoc

- Pause Pauses playback. **Play** will continue from paused place.
- Change Reopens browser to allow replacing the current file with another one.

10.3.3 Body Site Editor

The Body Site Editor option displays the complete Body Sites list as set by the Manufacturer, and enables the user to create a personal list of tested body sites.

The Body Site Editor is divided into two sections:

- 1. Manufacturer Body Sites
- 2. User Body Sites

Ма	nufacturer Body	structions Con	figuration Body Sites	
	Main site	Side	Specific Site	Normative D
	T12	Right	Side of Abdominal Wall	
	L3	Left	Gluteal	
	C8	Left	Hand Palmar Ring	
	C8	Left	Hand Dorsal Ring	
	1.5	L off	Low Dock	
	<u> </u>			
			Export	
Use	er Body Sites —		Export	
Use	Main site	Side	Specific Site Normative Da	ata Hide
, Use		Side		ata Hide

Figure 89: Body Sites Editor

10.3.3.1 <u>Manufacturer Body Sites</u>

The Manufacturer Body Sites section displays the complete list of body sites (Dermatomic areas) as set by the manufacturer.

The following information is available:

- 1. Main Site C1...C8; T1...T12; L1...L5; S1...S5.
- 2. Side Left or Right.
- 3. Secondary Site matches the Dermatomic areas.
- 4. Normative Data indicates if relevant normative data is available in the Normative Data list.
- 5. Hide enables the user to hide a site from the Body Site selection screen.



• To narrow down the displayed list use the filter row.

The following actions are available in the Manufacturer Body Sites Editor:

- 1. Export Body Site list to an excel file.
- 2. Import .ats binary Body Site list.
- 3. Select Hide or Display site in the Body Site selection screen by using the check box.

10.3.3.2 User Body Sites

The User Body site section enables the user to manage a customized Body Site list according to specific needs.

The following information is available:

- 1. Main Site C1...C8; T1...T12; L1...L5; S1...S5.
- 2. Side Left or Right.
- 3. Secondary Site matches the Dermatomic area.
- 4. Normative Data indicates if relevant normative data is available in Normative Data list.
- 5. Hide enables the user to hide a site from the Body Site selection screen.
- To narrow down the displayed list use the filter row.

The following actions are available in the User Body Sites Editor:

- 1. Export Body Site list to an excel file.
- 2. Import Body Site list from an excel file (Compatible files only).
- 3. Edit Body Site entry.
- 4. Delete Body Site entry.
- 5. Select Hide or Display site in the Body Site selection screen by using the check box.



11 System Information

The System Information menu displays the system details.

The following information is displayed:

- Software version.
- Firmware (Q-SENSE) version.
- Q-SENSE serial number.
- Supported Thermodes (according to license).
- MEDOC contact information.

In order to see the system information:

1. From the **Help** menu, select the **About Medoc Main Station** option. The following screen is displayed:

About MEDOC Main Sta	tion 🛞
12	Q-Sense
	Small Fiber Test
Version:	6.0.8.0
Firmware Version:	CTSA 2.0.0
Firmware Serial Number:	0
Supported Devices:	Pathway, AlgoMed, TSA-II, Q-Sense
Supported Thermodes:	CHEPS, ATS, Q-Sense, TSA, VSA
Supported Modes:	TSA, VSA
Supported Add-ons:	Cold Temperature, Parallel mode, TTL(TSA), Covas (TSA), USB Covas(TSA), Calibration(TSA)
Copyright ©	: 2005 Medoc Ltd. All rights reserved.
Telephones:	+972-4-903-8800 (Israel) +1-919-402-9600 (USA)
E-mail:	info@medoc-web.com
Web:	www.medoc-web.com
	OK Sysinfo

Figure 90: System Information

In order to send full system information to Medoc:

1. Click the **Sysinfo** button. A system information file similar to the following, is created and displayed:



🔄 systeminfo.txt - Notepad
<u>F</u> ile <u>E</u> dit F <u>o</u> rmat <u>V</u> iew <u>H</u> elp
16/07/2012 12:59:44
Pathway Advanced Thermal Stimulator
Version: 6.0.8.0 Firmware Version: CTSA 2.0.0 Firmware Serial Number: 0 Supported Devices: Pathway, AlgoMed, TSA-II, Q-Sense Supported Thermodes: CHEPS, ATS, Q-Sense, TSA, VSA Supported Modes: TSA, VSA Supported Add-ons: Cold Temperature, Parallel mode, TTL(TSA), Covas(TSA), USB Covas(TSA), Calibration(
System information: Operating System: Microsoft Windows 7 Professional version 6.1.7601 Service Pack 1
СРU's: СРU0: Intel(R) Core(TM) i5 СРU 760 @ 2.80GHz
Physical memory available to Windows: 3323 MB
Related products: Microsoft .NET Framework 4 Client Profile version 4.0.30319 Microsoft SQL Server 2005 Express Edition (ATS) version 9.4.5000.00 Microsoft .NET Framework 4 Multi-Targeting Pack version 4.0.30319 Microsoft .NET Framework 1.1 version 1.1.4322 Microsoft .NET Framework 4 Extended version 4.0.30319

Figure 91: Sysinfo File Sample

2. Save the file and send it by mail to your Medoc Service contact.



12 The Black box (Logger)

MEDOC Main Station is installed with a "black box" or logger that records and documents software and system information in case of error or failure. The files can be exported to excel and sent to Medoc support in order to identify the nature and cause of any technical malfunction.

12.1 Exporting the Black Box File

Follow these steps to export a black box file:

25. From the **Utilities** menu, select **Black Box**; The following screen is displayed:

Black Box		6
Time	Reason	Description
T		
06/19/2012 4:26 PM	Device fatal error	8021
		No thermodes are enabled
07/12/2012 9:27 AM	Device fatal error	1. Enable connected therm 2. Click OK, and resume Sel
	EXPORT	СКЕАТЕ

Figure 92: Black Box – Software

- 26. Select the entry to export and click **Export** button.
- 27. The **Save As** window is displayed.
- 28. Browse to the folder into which you want to save the Excel file, type the file name, and click the **Save** button.

The **Exporting data, please wait** prompt is displayed.

You can abort the exporting process at any time, by clicking the **Cancel** button.

- 29. After Export is completed, click **OK**.
- 30. To close the **Black Box** window, click the **OK** button.



13 Appendix A – Normative Data

Medoc Main Station includes a built-in database of Normative Data which has been collected from various studies. In addition, you can add your own or import normative data to the database by using the User Normative Data table. Normative data is categorized by Thermode size, age group, gender, program type, body sites and various other parameters depending on the source. Normative Data can be displayed in real-time during the test, or be viewed later with the test results or in the report. Normative data is only displayed if normative data exists for the selected body site, program parameters, and patient age and gender. The following parameters need to be matched between the current selection and normative data for it to be displayed:

- Patient age and gender
- Program Method, modality, baseline temperature and rate
- Body site (main site, side and specific site)

New normative data can be added to the Normative Data database. If there are matching parameters (according to the criteria above) between normative data added by the user and the normative data provided by the manufacturer, the former will be used.

Please note that the use of the built-in Normative Data is subject to the terms and conditions specified in the Normative Data license agreement.

13.1 Normative Data License Agreement

Use of the built-in normative data is subject to the terms and conditions as stated below.

PEASE READ THIS AGREEMENT CAREFULLY.

Medoc Ltd ("Medoc") is the sole proprietor of the software package known as the Medoc Main Station software or "MMS software".

Medoc has created an add-on feature for the MMS software, which is a Normative Data feature (the "ND Feature").

The built-in information contained in the ND Feature was collected from various publications and researches, and adapted by Medoc in order to enable use in conjunction with the MMS software.

You are welcome to use the ND Feature, but you may do so only in accordance with, and subject to, the following provisions. Use of the ND Feature by you means that you accept all the terms and conditions of this agreement, including all limitations. You further agree that this agreement is enforceable like any written, mutually negotiated agreement signed by you.

If you do not agree to any of the provisions of this agreement, please do not use the ND Feature.

Should you decide to install and use the ND Feature, you may do so under the following conditions:

1. Medoc is not responsible or liable for, and does not guarantee or underwrite, the accuracy, completeness or applicability of the information contained in the Normative Data.

2. Medoc is not responsible or liable for the use you make of the ND Feature, nor is it liable for your conclusions from such use.

3. Medoc is not responsible or liable for any clinical diagnostics reached or treatment given based on the Normative Data.

4. Any use of the ND Feature and the information contained therein is at the user's sole responsibility and liability.

Should you have any questions regarding the ND Feature or this agreement, please contact Medoc Ltd. at info@medoc-web.com



13.2 Enable Normative Data

Normative Data is automatically enabled if you agreed to the terms and conditions during Medoc Main Station installation (see section 4.6). However, even if you decided not to enable Normative Data during installation, you can always enable later.

To enable normative data follow the instructions given below:

1. To open the Normative Data editor select **Utilities** from the top menu bar, and then select **Normative Data**. The following screen is displayed:

🛞 Normative Data	8
Manufacturer Normative Data	
Data source	l
Method Modality Min. Age Max. Age Gender Main site Side Specific Sit Baseline Rate Lower limi Upper limi AVG STD N The	mode si
Import	
User Normative Data	
Method Modality Min. Agt Max. Agt Gender Main site Side Specific Si Baseline Rate Lower lin Upper lin AVG STD N (Thermode s U	ser Nam
Import EXPORT New Remove	
ОК Арру	Cancel

Figure 93: Normative Data Editor - Enable Normative Data

2. If normative data was not enabled during MMS installation, the tables will be blank. To enable Normative Data, click on **Enable normative data** (circled above). The following screen is displayed:



Normative Data Manufacturer Norm	ative Data					- de la com		
Data source			Enable	normative data				li
M 🔍 Install	Normative Data						i	O sí
1		Please review the lice	nse terms before i	nstalling Norma	ative Data			A
Dear Use PLEASE f Medoc Lt Medoc ha The built to enable	EAD THIS AGREEMENT C. d ("Medoc") is the sole pro- s created an add-on featu- in information contained use in conjunction with the	AREFULLY. oprietor of the software ire for the MMS softwar n the ND Feature was o ne MMS software.	e, which is a Normat ollected from various	ve Data feature publications and	(the "ND Fea i researches	ature"). , and adapted by M	edoc in order	
User N If you do Should y 1. Medoc informati 2. Medoc 3. Medoc	velcome to use the ND Fer y you means that you acc it is enforceable like any i not agree to any of the p ou decide to install and us is not responsible or liabl on contained in the Norm is not responsible or liabl is not responsible or liabl e of the ND Feature and t	tept all the terms and co written, mutually negoti rovisions of this agreem e the ND Feature, you r e for, and does not gua titve Data. e for the use you make e for any clinical diagno	onditions of this agree ated agreement sign ent, please do not u nay do so under the rantee or underwrite of the ND Feature, n stics reached or trea	ement, including ed by you. se the ND Featur following condition the accuracy, co or is it liable for coment given base	all limitation e. ons: ompleteness your conclus ed on the No	or applicability of t ions from such use rmative Data.	e that this he	
Should y	ou have any questions reg	arding the ND Feature	or this agreement, pl	ease contact Med	loc Ltd. at ir	fo@medoc-web.co	m	
	ccept the terms of agre it accept the agreemen							
						Аррју	Cancel	
Import		New						
				C	ОК	Apply		ancel

Figure 94: Normative Data License Agreement

- Read the license agreement and click **Apply** to accept.
 Normative Data is now enabled.



13.3

ta source	H	Hafner et a	al 2015					Learn moi							D i
Method	Modality	Min. Age	Max. Age	Gender	Main site	Side	Specific Sit	Baseline	Rate	Lower limi	Upper limi	AVG	STD	N	Thermode si
Lim	War	20	20	Male	L5	Left	Foot D	32	1	0	40.98	0	0	0	30x30
Lim	War	20	20	Male	C8	Left	Forear	32	1	0	35.09	0	0	0	30x30
	War	20		Male	L5	Left	Forear Foot D	32	1	0	41.08			0	30x30 30x30
Lim			21									0	0		
Lim	War	21	21	Male	C8	Left	Forear	32	1	0	35.1	0	0	0	30x30
Lim	War	22	22	Male	L5	Left	Foot D	32	1	0	41.17	0	0	0	30x30
Lim	War	22	22	Male	C8	Left	Forear	32	1	0	35.1	0	0	0	30x30
	ive Data -	EXPO		Mala ender Ma	in sit€ Sid	e Spe	ecific S Base	ο linε Rate	Lowe	n lin Upper	lin AVG	STD I	O N (Th	ermo	de s User Nam
er Normati	ive Data -	EXPO	RT												
er Normati	ive Data -	EXPO	RT												

Figure 95: Normative Data Editor - Learn More

13.4 Normative Data Editor

The Normative Data editor allows you to change the Normative Data source, view and export the available built-in normative data, to import, add or modify your own normative data.

13.4.1 Selecting the Normative Data Source (Built-In Tables)

The built-in Normative Data includes several normative data tables from different sources. Each Normative Data table is usually based on a different publication. The normative data tables may differ in age groups, body sites and even in normative values for same gender, age group and body site. Before using the built-in normative data select the source which best fit your application.

To select Normative Data source table following the instructions below:

- 1. Open the Normative Data editor: select **Utilities** from the top menu bar, and then select **Normative Data**.
- 2. Click the arrow to right of the data source name. A list will open displaying the list of available sources. See Figure 96.



ta source		Default N	ormative (Data										\mathbf{X}	l
Method	Modali	Blankenbu	urg et al 20	010			c Sit	Baseline	Rate	Lower limi	Upper limi	AVG	STD	N	Thermode si
Lim	War	Default N	ormative D	ata				32	1	33.55	36.99	3	0	9	30x30
Lim	Cold	Hafner et	al 2015					32	1	26.13	31.73	2	1.4	9	30x30
Lim	Cold	30	39	Male	S1	Ri	Foot D	32	1	26.72	33.28	30	1	4	30x30
Lim	Cold	20	29	Fe	C2	Ri	Maxill	32	1	31.41	31.65	3	0	1	30x30
Lev	Cold	60	69	Male	S1	Left	Foot D	32	1	26.97	33.33	3	1	1	30x30
Lev	Cold	50	59	Fe	C6	Ri	Hand	32	1	30.12	31.96	3	0	4	30x30
Lim	Cold	60	60	En	т1	Di	Uppor	22	1	25.10	22.21	2	1	0	20~20
				Gender M	ain site Sie	de Spe	ecific Si Base	line Rate	Lowe	r lin [.] Upper	lim AVG S	STD I	N (Th	ermoc	le s User Nam
Impor				Gender M	ain site Si	de Spe	ecific S Base	line Rate	Lowe	r lin [°] Upper	lin AVG S	STD I	N (Th	ermoo	ie s User Nam

Figure 96: Normative Data screen select source

- 3. Click on one of the sources in list to select. Note that changing the normative data source will affect the display of normative data in results and in the reports. Confirm your selection.
- 4. To learn more about the selected normative data source click on the "learn more" link. Please note this link may not be available for all sources. See Figure 95.



Changing the normative data source affects the display of normative data in saved test results and in generated reports.

13.4.2 Normative Data Tables

The Normative Data editor includes two separate tables:

- 1. Manufacturer Normative Data
- 2. User Normative Data

The Manufacturer Normative Data table is a 'read-only' table. The data cannot be edited or modified. If needed, you can export the data, modify it and import it again as User Normative Data or add new Normative Data to a blank table.

To open the Normative Data editor, select **Utilities** from the top menu bar, and then select **Normative Data**. The following screen is displayed:



ta source		Default No	rmative Da	ata											li
Method	Modality	Min. Age	Max Age	Gondor	Main site	Sido	Specific Sit	Pacolino	Pate	Lower limi	Upper limi	AVG	STD	N	Thermode si
Hethod	Hodalicy	Philli Age	Hax. Age	Gender	Plain Sice	Side	opecine on	Dasenne	Nate	Lower min	оррег шш		510		Thermode a
Lim	War	60	69	Male	C8	Left	Hand	32	1	33.55	36.99	3	0	9	30x30
Lim	Cold	60	69	Fe	L4	Left	Knee	32	1	26.13	31.73	2	1.4	9	30x30
Lim	Cold	30	39	Male	S1	Ri	Foot D	32	1	26.72	33.28	30	1	4	30x30
Lim	Cold	20	29	Fe	C2	Ri	Maxill	32	1	31.41	31.65	3	0	1	30x30
Lev	Cold	60	69	Male	S1	Left	Foot D	32	1	26.97	33.33	3	1	1	30x30
Lev	Cold	50	59	Fe	C6	Ri	Hand	32	1	30.12	31.96	3	0	4	30x30
Lim	Cold	60													
Impor er Normat	ive Data	EXPO		ender Ma	T1	Di	Linnor ecific SI Base	22 line Rate	Lowe	25 10 r lin Upper	lim AVG	STD 1	1 N (Th	ermoo	20v20 le s User Nam
Impor er Normat	ive Data	EXPO	RT										1 N (Th		
Impor er Normat	ive Data	EXPO	RT										N (Th		

Figure 97: Normative Data Editor

Normative data tables include the following parameters:

Table 38: Normative Data Fields

Parameter	Description	Mandatory
Method	The test method (Limits, Levels,)	✓
Modality	The test modality (Cold Detection Threshold, Warm Detection,)	~
Minimum Age	Lower limit of age for which this Normative Data is applicable	~
Maximum Age	Upper limit of age for which this Normative Data is applicable	~
Gender	The gender that this Normative Data is applicable	✓
Main Site	The dermatome that was used to collect the normative data.	✓
Side	The anatomic side that was used to collect normative data.	
Specific Site	The specific body site that was used to collect the normative data.	



Parameter	Description	Mandatory
Baseline	The Baseline Temperature	√
Rate	The temperature change rate	√
N	The number of subjects which the normative data statistics is based on.	
AVG	The average threshold of the subjects that were tested to collect the specified normative data.	See note bellow
STD	The standard deviation of the threshold of a sample subject who was tested to collect the specified normative data.	See note bellow
Upper Limit	The upper limit as calculated from the normative data. Upper Limit = AVG + $2xSTD$.	See note bellow
Lower Limit	The lower limit as calculated from the normative data. Lower Limit = AVG – $2xSTD$.	See note bellow
Thermode Size	Thermode size used to collect this Normative Data (30x30, 16x16,)	

Note: In order to apply the normative data to test results, it is necessary to provide the upper limit for Warm Detection Threshold and Heat Pain or the lower limit for Cold Detection Threshold and Cold Pain. However, if these parameters are left blank, the software will calculated them from the average and standard deviation. In this case of course, the average and standard deviation must be provided instead.

The following operations are available:

- **Export or import a list of normative data** The list of normative data can be exported or imported from a excel file. To see the required configuration of the excel file required for import, export the manufacturer normative data list and use the same format. The user normative data list can also be exported and important as an *.ats file for backup.
- Add or remove data from the list of User Normative Data add, edit or remove normative data from the user normative data list.

To add new normative data to the list:

- 1. Click the **New** button to add a new row to the User Normative Data table
- 2. Enter the details for the normative data (for information about the specific parameters see Table 38 above)
- 3. Click the **New** button to add more rows or **Apply** to save and apply the new data.
- 4. To modify existing data in the User Normative Data table, just change the values you want to modify and click **Apply** to confirm changes.

To import normative data from a file:

- 5. Normative data can be imported only into the User Normative Data table. The data must be arranged in a formatted table and saves as an excel file.
- To see the correct format, export one or more records from the Manufacturer Normative Data. To do so, select one or more records in the Manufacturer Normative Data table and click the Export button.



- 7. Open the exported file, delete the existing data except for the column headers and add your own data in exactly the same format. Save the file as .xlsx or .xls.
- 8. To import the file into the User Normative Data table, click the **Import** button below the User Normative Data table and select the file you want to import.

13.5 Display of Normative Data in Tests, Results and Reports

The Normative Data can be displayed during the test, overlaid on saved test results and in the reports. on the test screen when its parameters match the current test parameters.



The Data will be displayed only if the following test parameters match Normative Data parameters: Method, Modality, Baseline, Rate, Main Site, Side, Specific Site, Age Group and Gender.

You may choose to display Normative Data as a default. Go to Settings → Test
 Configuration → Presentation Configuration and mark the relevant check box.



14 Appendix B – Backing Up and Restoring the Database

The Data Base Utility enables Back-up and restore system date base.

During installation of MEDOC Main Station, a default data base is created. It is called: "Computer name/**ATS**". This data base allows backup and reconstruction of the data base.



It is highly recommended to backup the database, on a monthly basis.



It is highly recommended to define database accessibility via Windows (not SQL).

To backup or restore the Q-SENSE system database, perform the following procedure:

1. From the Start menu select All Programs → Medoc → MEDOC Main Station → Database Configuration Utility.

The following menu is displayed:

lesse Utility 🕘	×
Backup Restore	Overwrites existing data !

Figure 98: Database Utility – Backup/Restore Menu

To backup the data base, click the **Backup** button.
 A **Saving** dialog box is displayed. Use a meaningful name for better identification.



Due to Windows security, Database backup files can be saved only under: "C:\MEDOC\DataBaseBackup".

- To restore a database, click the **Restore** buttonAt the prompt, click the **OK** button. A dialog box that allows choosing the data base source will open and allow choosing the data base you'd like to use.
- 4. Choose the desired data base and click the **Open** button.


15 Appendix C - Accessories

The following list of accessories can be purchased from Medoc, and are listed with their Catalogue numbers, which should be quoted when ordering a specific accessory.

	e 39: Accesso	
Description	Cat. No.	
Standard Q-sense 30x30mm Thermode This is the standard thermal probe, which is capable of heating the skin. Used on large body sites.	AS 00402	
Temperature Calibration Kit This kit is used for checking and calibrating the temperature calibration of the <i>Q</i> -sense themodeII unit, which can be a requirement of certain grant- funded research/study protocols. See the service manual for operation details.	AS 00406	<image/>

Table 39: Accessories



Description	Cat. No.	
Response Unit This unit enables the tested subject to respond to the stimuli by pressing one of two buttons. Depending on the selected test, the tested subject feedback signal terminates any process or causes the test to continue following a Yes/No response.	AS 00403	<image/>
USB CoVAS – Computerized Visual Analog Scale While the Q-sense delivers a heat stimulus to a tested subject's body site, the subject simultaneously moves the CoVAS slide. The movement of the slide is between a predefined minimum and maximum (0 to 100), which depicts the level of pain he/she feels.	AS 00213	
Thermode Strap	AS 00404	



Description	Cat. No.	
		AS 00404
Wall Mount Used to hang the system on a wall.	MP 00501	
Power Supply Power supply is used to connect electricity port and to the power port in the Q-sense system.	DT 00017	
USB Cable This cable is used to connect the Q- sense com port with a computer.	WA 00278	



16 Appendix D – System and Module Description

16.1System Description

The Q-SENSE modular design makes it easy to troubleshoot and replace parts.

The system consists of main module:

• Q-SENSE Accessories

For a detailed description of the accessories, refer to section 15

- \circ $\;$ Thermode (to be attached to the tested subject's skin).
- Response Unit.
- Q-SENSE Utilities
 - O Laptop.
 - O Printer.



Figure 99: System Diagram



17 Appendix E – Preventive and Periodical Maintenance

This chapter describes periodic maintenance procedures.

17.1Periodical System Run and Visual Check

A periodical system run and visual examination are necessary for maintaining performance of Q-SENSE system. The periodical run keeps all Q-SENSE accessories and subunits in operational condition at all times. Periodical system run is required, if the system or the supplied Thermode is not used on a regular basis, or if it has not been in use for a period of one month or more,



Visual check is required in order to make sure that there is no physical damage to system components, particularly to the Thermodes.

- 1. Check all accessories and Thermodes visually.
 - 1. Check that accessories connector is intact.
 - 2. Check that accessories body is intact.
 - 3. Check that there is no visible damage to the Thermode heads.
- Install all available Thermode types (Q-sense type), according to your license type, to the Q-SENSE system.
- 3. Connect all available Q-Sense accessories (Response Unit) to the Q-SENSE system.
- 4. Connect the communication cable from the Q-Sense system to the computer running MEDOC Main Station.
- 5. Power ON the Q-SENSE system.
- 6. Start MEDOC Main Station and ensure that communication is established.
- 7. Run the system for about 10 minutes; check for any visible malfunctions.
- 8. Run a few different test protocols such as Limits.
- 9. Repeat steps 8 and 9 with all the Thermodes supplied with the system.



Switch Thermodes only when the Q-Sense system is turned OFF!



Remember to change calibration tables prior switching to different Q-sense Thermode type (30x30mm).

- 10. If the system is not planned for use for three weeks or more, perform the following instructions:
 - a. Turn OFF the Q-SENSE system through the main switch.
 - b. Disconnect all accessories from the Q-SENSE system, and store them in a dry place.
 - c. Disconnect all Thermodes from the Q-SENSE system, cover the Thermode heads with soft cloth and store in dark dry place.





Changes, additions or maintenance to the equipment carried out by persons without appropriate qualifications and training and /or using un approved spare parts may lead to serious risk of injury and damage to the equipment as well as making the warranty void.

17.2Periodical Thermode Calibration

A periodical Thermode Calibration is necessary for the Q-SENSE system performance.

Perform Thermode Calibration every six months, according to the Thermode calibration service manual.

17.3Periodical Thermode Maintenance

- 1. Inspect Thermode to confirm mechanical integrity and no visual damage before use on patient.
- 2. Replacing Thermode once a year is needed to keep System integrity. Please contact Medoc representative in your Area for more information on Thermode replacement and warranty extended plan.



18 Appendix F – Cleaning and Disinfecting

Perform this procedure before initially using the Thermode and after each use, to minimize the spread of pathogens between tested subjects.

- 1. Thoroughly clean the Thermode contact surface using alcohol wipes.
- 2. Allow to air-dry.
- 3. Make sure the Thermode Head Case Perforated is clean of dust.
- 4. Store the Thermode in a clean, dry area.



19 Appendix G - Thermode straps assembly instructions

1. The Q-sense Thermode strap is made out of two part – short and long



2. Insert the round end of the long strap into the slot on the Thermode side (when the hook side is facing to you) and fold it on the loop side.



3. Insert the round end of the short strap into the slot on the other side of the Thermode (when the hook side is facing to you) and fold it on the small loop square.



4. The Thermode is ready to use





20 Appendix H – List of Accompanying Documents

Table 40: Q-sense Accompanying Documents

- Quick installation guide.
- Release Notes.
- Acceptance Test Procedure (doc)
- Packing list
- Q-sense drilling model for assembly holder
- Q-Sense Operating Manual



21 Appendix I – Transportation and Storage conditions

The environment conditions for Q-SENSE system transportation and storage are as follows:

- An ambient temperature range of 0 °C to +70 °C.
- A relative humidity range of 10% to 95%, including condensation.
- An atmospheric pressure range of 500 hPa to 1060 hPa.



22 Appendix J – Environmental Conditions

The environment conditions for Q-SENSE system during standard use are as follows:

- An ambient temperature range of +18 °C to +24 °C
- A relative humidity range of 30% to 75%, including condensation
- An atmospheric pressure range of 700 hPa to 1060 hPa



23 Appendix K – Upgrading procedure

This section regards installation for users having older version of Medoc Main Station.

1. Users having previous version of Medoc Main Station on the laptop will have the following screen displayed:



Figure 100: Previous version message

2. To upgrade version, continue with installation and click OK. The following screen will be displayed:



Figure 101: Removing previos version screen

3. In order to remove older versions of MMS Click YES. The following screen will be displayed:

🗱 Medoc Main Station Uninstall	
DataBase backup path DataBase backup path	
Please select folder for database file backup	
El (Medoc)DataBaseBackup	
Unins .	tall Cancel

Figure 102: Data Backup Path

4. Choose folder for database file backup and click Uninstall. The following screen will be displayed:





Figure 103: MMS Uninstall

5. When uninstall is complete the following screen will be displayed:



Figure 104: Uninstall Completed

6. Click finish. The following screen will be displayed:



Figure 105: MMS successfully removed

7. Click OK. To continue installation refer to Section 4.6, Page 39.



24 Appendix L – Troubleshooting – Operator Level

The troubleshooting section describes some of the Q-SENSE most frequent Error messages and Warnings; it helps you resolve them at operator level.

There are two types of malfunctions: Software and Hardware.

- 1. Software malfunctions consist of lost or damaged files that can result, for example, in missing or incomplete tested subject information.
- 2. Hardware malfunctions manifest themselves either as occurrences that you become aware of, such as Thermode not responding, or as a malfunction message displayed on the screen.

If you detect a malfunction, do the following:

1. If the malfunction you have encountered is an Error or other message refer to section 24.1.1, page 158.

For other malfunctions, see alternative solutions in the following sections.

24.1 Numbered Software Error and Warning Messages

The design of all the Q-Sense Error messages and Warning messages is similar to this sample message:



Figure 106: Error and Warning Messages Sample

The text displayed in the message box varies according to the detected malfunction.

24.1.1 Frequent Warning Messages

Table 41: Warnings

Nº	Warning	Explanation
1.	The application is about to start Q-SENSE recovery and self test.	The application needs to perform a self test.
	Make sure the thermode is not attached to the patient and press OK.	 Remove Thermode from subject. Click the OK button to start performing a Self Test.



N⁰	Warning	Explanation
2.	Q-SENSE is in Safe Mode, and must perform Self Test before resuming operation. Click YES to perform Self Test, and resume work.	The Q-SENSE system is in Safe Mode and must perform Self Test before it continues with normal operation.Click YES if you wish to resume the
	Click NO to continue in Safe Mode.	 Self Test. Click NO if you do not want to resume the Self Test.
3.	Do you want to retry Q-SENSE recovery?	The Q-SENSE is ready to retry performing Self Test.
		1. Click YES to start the Self Test.
		 Click NO to continue working in Safe Mode.
4.	A conflict occurred while importing a user- defined field. Please resolve the conflict by performing one of the next actions: 1. Choose the local definition for this field. 2. Choose the imported definition for this field.	 There is a conflict between the existing user-defined entry in the data base and the one you have imported. Select the entry you wish to keep: local definition, or imported definition.
5.	 A conflict occurred while importing a user. Please resolve the conflict by performing one of the next actions: 1. Choose one of the local users. 2. Edit the conflicting fields in the imported user. 	 There is a conflict between the existing Q-SENSE user entry in the data base and the one you have imported. Select the entry you wish to keep: local user, or imported user, or modify the conflicting fields in the imported user.
6.	A conflict occurred in department name. Please resolve this conflict by performing one of the next actions: 1. Choose the local department. 2. Edit the imported department name.	 There is a conflict between the existing department name in the data base and the one you have imported. Select the entry you wish to keep: local department, or imported department, or modify the imported department name.
7.	A conflict occurred while importing a patient. Please resolve the conflict by performing one of the next actions: 1. Choose one of the local patients. 2. Edit the conflicting fields in the imported patient.	 There is a conflict between the existing patient in the data base and the one you have imported. 1. Select the entry you wish to keep, local patient, or imported patient, or modify the imported department name
8.	Configuration data was changed. Press YES to save the configuration data. Press No to discard changes. Press Cancel to continue editing.	The Q-SENSE configuration data has been changed. Select one of the presented options and continue.



N⁰	Warning	Explanation
9.	Program violates safety limitations, and cannot be saved. Modify program settings, and retry saving. Violations: Sequence: ##, Temperature: ##, Actual time (sec): ##, Allowed time (sec): ##	The program you have designed violates the Q-SENSE safety limitations and therefore can not be saved to the data base. You have to modify the program parameters that are stated as violation according to the Q-SENSE safety table before saving this program.
10.	Deleting current Thermode is not allowed. Please set other Thermode as current and then delete this Thermode.	 You can not delete a Thermode you are currently using from the data base. 1. Set another Thermode as the current (active) Thermode. 2. Delete the Thermode you wish to remove from data base
11.	Are you sure you want to save this Thermode in the local database and then set it as current? 1. Click YES to proceed. 2. Click NO to abort.	You are about to save the selected Thermode parameters to the Q-SENSE data base and set it as the current (active) Thermode of the Q-SENSE system. Select one of the presented options and continue.
12.	Are you sure you want to replace the selected Thermode tables with the current Q-SENSE tables? 1. Click YES to proceed. 2. Click NO to abort.	You are about to replace the selected data base Thermode parameters with the current (active) Q-SENSE Thermode parameters. Select one of the presented options and continue.
13.	Some of the program parameters are invalid. Press YES to continue editing. Press NO to discard all changes.	One or more of the program parameters is not valid and must be modified. Select one of the presented options and continue.
14.	The default instruction is not defined for this program.	No test instructions are defined for the selected method. You can add instructions through Settings → Test Settings → Instructions Configuration
15.	Thermode calibration data was changed but not saved. Press Ok to exit calibration and discard the new calibration data. Press Cancel to continue calibration.	The Thermode calibration data was not saved. Select one of the presented options and continue.



Nº	Warning	Explanation
16.	The configuration of the Thermode was changed. Selecting another Thermode will discard current changes. Press YES to select another Thermode and discard the changes Press NO to continue editing previous Thermode.	You have changed Thermode configuration but have not saved it. Select one of the presented options and continue.
17.	The imported Thermode has the same serial number as existing "Thermode Name" Thermode (serial number: #####). Press Yes to override the existing Thermode. Press No to create a new Thermode (serial number will be omitted).	The Thermode file you try to import to the data base has same serial number as existing data base Thermode. Select one of the presented options and continue.
18.	Sensor <sensor name="">*: missing calibration table entries in range 10 - 48. *<sensor name="">= TEC\HS1\HS2</sensor></sensor>	The sensor LUT can not be complete, because there are missing entries. Complete the missing entries and continue.
19.	Sensor "Sensor Name": The number of defined calibration table entries is less than the required 10.	In order to complete calibration there must be at least 10 entries in the calibration table. Complete missing entries and continue.
20.	The number of defined calibration table entries is more than the required 10, but less than the recommended 20. 1. Click YES if you want Q-SENSE to perform interpolation and fill out the missing values. 2. Click NO to enable return to the Hardware Settings and fill in missing values.	 When performing manual calibration, at least 10 entries are required, although 20 are recommended by Medoc. This message states that you have calibrated less than the recommended number of points. You can choose to complete 20 entries or let the program interpolate the missing values.
21.	You cannot change Q-SENSE (firmware) Thermode values directly	Q-SENSE Thermode firmware can only be viewed. Only Thermode data bases can be edited, and then set as current Q- SENSE Thermode firmware.
22.	Destination temperature is lower than the baseline. The interval between baseline and destination must be less than 15°C. Sequence 1 - The interval is 36 - 20 = 16°C	When performing Cold EP tests there are restrictions that must be followed: The interval between the baseline temperature and destination temperature must be less than 15°C. Make sure that your program complies with these restrictions; then continue.



Nº	Warning	Explanation
23.	Destination rate is out of range: Sequence 1 - For a destination	When performing a Cold EP test there are restrictions that must be followed:
	temperature of 23°C, the range must be [0.11].	 Temperature change rate must be within the specified range.
		Make sure that your program complies with these restrictions and continue.
24.	Destination temperature is lower than the baseline.	When performing a Cold EP test there are restrictions that must be followed:
	Sequence 2 - Return rate must be in the range [0.12]	 The temperature return rate must be within the specified range.
		Make sure that your program complies with these restrictions and continue.

24.1.2 Frequent Error Messages

Table 42: Errors

Nº	Error Message	Solution
1.		The appearance of this icon, beside a configurable field or parameter, states that the parameter value is illegal.
		1. Move the cursor over the sign for further information.



N⁰	Error Message	Solution
2.	Communication with Q-Sense was lost	Communication between Q-SENSE system and PC was lost.
	 Click OK to check if communication can be re-established automatically, run self-test and continue working with the system In case communication is not re- established automatically: Close Medoc Main Station and switch OFF the device Verify that the device Verify that the device communication cable is connected firmly to device and computer. Turn ON the device and start Medoc Main Station. If steps 1 & 2 fail, contact Medoc support for further assistance. 	 Follow the message instructions to solve the error. 1. Click OK to check if communication can be re-established automatically, run self-test and continue working with the system 2. In case communication is not reestablished automatically: In case communication is not reestablished automatically: Close Medoc Main Station and switch OFF the device Verify that the device Verify that the device and computer. Turn ON the device and start Medoc Main Station. 3. If steps 1 & 2 fail, contact Medoc support for further assistance.
3.	No Thermodes are enabled. Error Code: 800D 1. Enable connected Thermode via Hardware Settings, General tab. 2. Click OK, and resume Self Test when prompted.	The Q-SENSE does not have any enabled Thermodes. You must enable the connected Thermodes via Hardware Settings and then perform Self Test , before continuing working with the system.
4.	Failed to recovery from Safe Mode, Q- SENSE returns Acknowledge Code = ##.	 The Q-SENSE system was unable to complete Self Test recovery due to an error. The Q-SENSE system is reporting the acknowledge code that was generated. 1. Write the Acknowledge Code number or name. 2. Restart the Q-SENSE system and program. 3. Continue working. If the problem persists, contact Medoc service and state the Acknowledge Code number or name you experienced.



N⁰	Error Message	Solution
5.	Failed to recovery from Safe Mode, no response from Q-SENSE during timeout period.	The Q-SENSE system was unable to complete the Self Test recovery, because the program could not connect to the system.
		 Restart the Q-SENSE system and program.
		2. Continue working.
		If the problem persists, contact Medoc service.
6.	MEDOC Main Station can't locate Q-sense device. Please check the following:	MEDOC Main Station failed to connect to the Q-Sense system.
	 Verify that the correct port is selected. Refer to the computer device manager which port Q-Sense is connected. Make sure Q-Sense device is connected and turned ON. 	 Verify that the correct port is selected. Refer to the computer device manager which port Q-Sense is connected. Make sure Q-Sense device is connected and turned ON.If the problem persists, contact Medoc service.
7.	Unsupported TCU firmware version. Firmware version: ###	MEDOC Main Station has detected that the firmware version installed on the Q- Sense system is not compatible with the software.1. Make sure the HASP is well connected.
		Restart Q-SENSE system and software and try to log in
		 In case Restart did not solve the problem: 1. Load the firmware, .hex file. Use the installation CD, from which you installed the program; or the update folder in the Medoc FTP site, from which you downloaded from, see Service Manual for further information.
8.	User Name or Password are incorrect.	The username or password you have entered is not the right ones.
		1. Verify that you have typed the correct user name and password.
		2. Try to log in.
		If the problem persists, try the <i>admin</i> user name and password.



N⁰	Error Message	Solution
9.	Command: ### failed ###. No response from Q-SENSE.	The command failed, because no response from the Q-SENSE was detected.
		1. Write the command name or number.
		Restart the Q-SENSE system and program.
		3. Continue working.
		If the problem persists, contact Medoc service and state the Acknowledge Code number or name.
10.	Command: ### failed ###. Q-SENSE returns Acknowledge Code =	The command failed as a result of the displayed acknowledges code.
	###.	 Write the command name or number and the acknowledge code.
		Restart the Q-SENSE system and program.
		3. Continue working.
		If the problem persists, contact Medoc service and state the Acknowledge Code number or name you experienced.
11.	The Thermode cannot be saved because its verification failed.	One or more of the Thermode parameters are illegal and therefore can not be saved.
		Verify that all Thermode parameters are legal and modify the illegal ones; then save the Thermode parameters.
12.	The following safety limitation has been violated. Temperature: 51 °C; Allowed time: 0 (sec). Error Code: 8002	The Q-SENSE system detected that a safety limitation was violated during a test run, and switched automatically to safe mode in order to prevent any damage.
		The violated temperature and the time of the event are displayed, together with the error code.
		 Check that the parameters of the program you ran are valid.
		Check that the Thermodes are intact and undamaged.
		 Run Self Test to verify that the Q- SENSE system is enabled to continue work.
		If the problem persists, stop using the Q-SENSE system and contact Medoc service.



N⁰	Error Message	Solution
13.	Static sensor mismatch. Error Code: 8003	The Q-SENSE system has detected a mismatch between its sensors, and therefore switched to Safe Mode in order to prevent any damage.
		 Check that the Thermodes are intact and undamaged.
		 Run the Self Test to verify that the Q-SENSE system is enabled to continue work.
		3. Run Thermode calibration.
		If the problem persists, stop using the Q- SENSE system and contact Medoc service.
14.	Static sensor mismatch. Mismatch (difference) temperature 4 (°C). Error Code: 8003	The Q-SENSE system has detected a mismatch between its sensors during static temperature measurement, and therefore switched to Safe Mode in order to prevent any damage.
		The mismatched temperature that was measured is as reported in the message.
		 Check that the Thermode is intact and undamaged.
		 Run Self Test to verify that the Q- SENSE system is enabled to continue work.
		If the problem persists, stop using the Q- SENSE system and contact Medoc service.
15.	Dynamic sensor mismatch. Mismatch (difference) temperature: 10 (°C). Error Code: 8004	The Q-SENSE system has detected a mismatch between its sensors during a dynamic temperature change, and therefore switched to Safe Mode in order to prevent any damage.
		The mismatched temperature that was measured is as reported in the message.
		 Check that the Thermode is intact and undamaged.
		2. Run the Self Test to verify that the Q-SENSE system is enabled to continue work.
		If the problem persists, stop using the Q- SENSE system and contact Medoc service.



N⁰	Error Message	Solution
16.	Thermode TEC did not respond to temperature changes. Error Code: 8007 Make sure that the blue connector is connected properly	The Thermode has failed to pass TEC (part of the Thermode) test during the Self Test, and therefore the Q-SENSE has switched to Safe Mode in order to prevent any damage.
		 Check that the Thermode is intact and undamaged.
		Make sure that the blue connector is connected properly
		 Run Self Test to verify that the Q- SENSE system is enabled to continue work.
		 If self Test fails, Restart system and software and retry.
		If the problem persists, stop using the Q- SENSE system and contact Medoc service.
17.	TEC sensor is shorted or disconnected. Error Code: 800A	The Q-SENSE system has detected that the TEC (part of the Thermode) sensor is shorted or not connected, and therefore switched to Safe Mode in order to prevent any damage.
		 Check that the Thermode is intact and undamaged.
		 Run Self Test to verify that the Q- SENSE system is enabled to continue work.
		If the problem persists, stop using the Q- SENSE system and contact Medoc service.
18.	Empty command buffer. Error Code: 400C	An error has occurred in the Q-SENSE processor during run time.
		 Click OK and continue working as usual.
		If the problem persists, send the corresponding error code to the Medoc service team.



N⁰	Error Message	Solution
19.	Time mark missed. Error Code: 400D	An error has occurred in the Q-SENSE processor during run time.
		 Click OK and continue working as usual.
		Restart the Q-SENSE system and software.
		If the problem persists, send the corresponding error code to the Medoc service team.
20.	Sample buffer full. Error Code: 2001	An error occurred in the Q-SENSE processor during run time.
		 Click OK and continue working as usual.
		If the problem persists, send the corresponding error code to the Medoc service team.
21.	Finite ramp by temperature timeout. Thermode temperature: 35 (°C). Error Code: 2002	The Q-SENSE system was unable to complete the requested command in the allowed time, and has therefore switched to Safe Mode in order to prevent any damage.
		 Check that the Fan mode is according to requirement.
		 Run Self Test to verify that the Q- SENSE system is enabled to continue work.
		If the problem persists, stop using the Q-SENSE system and contact Medoc service.
22.	Safety cold duration limit violation. Temperature level: 20 (°C), Time entry: 10(sec). Error Code: 800E	The Q-SENSE system has detected that the safety limitation was violated, and therefore switched to Safe Mode in order to prevent any damage.
		 Make sure that the program you are trying to run complies with the safety limitations.
		 Run Self Test to verify that the Q- SENSE system is enabled to continue work.
23.	System could not establish communication! Retry: The system will retry to establish communication.	The program was unable to establish communication with the Q-SENSE system.
	Cancel: Continue working without connecting to device.	Select one of the presented options and continue.



Nº	Error Message	Solution
24.	The thermometer does not respond. 1. Check that the thermometer is turned on.	Automatic calibration had failed, because the program could not communicate with the thermometer.
	2. Check that the thermometer is connected to the correct COM (see setting > software setting > calibration COM port).	Follow the displayed instructions, and retry.
25.	The measured temperature 51(°C) is out of permitted range [2050].	During automatic calibration the measured temperature exceeded the allowed range.
		 Reduce calibration range (increase minimum and decrease maximum) through Automatic calibration properties.
		2. Retry calibrating.
		If calibration is still out of range, calibrate the Thermode manually to bring it back into range.
26.	Have not been able to read temperatures from the Thermometer for 60 seconds.	The program was unable to communicate with the thermometer.
	 Verify that the thermometer communication cables are connected properly. Verify that the thermometer is turned on. Restart the automatic calibration procedure. 	Follow the displayed instructions, and retry.
27.	 Check that the thermode is properly placed in the jig and connected to the Q- 	Warning notification before automatic calibration.
	Sense. 2. During automatic calibration software safety protection is lowered; therefore, it is not allowed to place Thermode on human subjects Click YES to continue calibrating. Click NO to discontinue calibration	Follow the displayed instructions, and continue.
28.	Can not open thermometer serial communication port, because it is used by	The selected calibration COM port is in use by the Q-SENSE system.
	Medoc Main Station	 Select the appropriate calibration COM port through Settings→Software Settings→General.
		2. Retry calibration.

N⁰	Error Message	Solution
29.	 Q-sense Thermode was not detected by Q-SENSE. The Thermode has been disabled. 1. Make sure that the Thermode is properly connected to the Q-Sense system. 2. Enable the Thermode through the Hardware Settings, General menu. 3. Click Yes to try again. Or 4. Click No to continue working. 	The Q-SENSE system could not detect one of the enabled Thermodes, and therefore disabled it in order to prevent any damage. Follow the displayed instructions to continue working.
30.	Some of the Safety parameters (Temp Vs. Time) are violating the Q-SENSE safety limitations. 1. Click OK to continue working with current non-approved safety parameters. 2. Click Restore to restore safety parameters to the Q-SENSE definition.	The Q-SENSE system has detected that the modified safety limitations do not comply with safety regulations. Select one of the presented options and continue. In is not allowed to perform any tests on human subjects when violating safety limitations without written approval!
31.	Communication between Q-SENSE and PC lost while in Test Run mode. Error Code: 800C	 Communication between the Q-SENSE system and the PC has been lost. 1. Restart the Q-SENSE system and program. 2. Run Self Test. 3. Continue working with the system. If the problem persists, contact Medoc service.



Nº	Error Message	Solution
32.	System AuliReferenceEucestoru. Object reference not set to an Instance of an object. at Nedoc Systifichieber celesienilverManned). Netocom State State Systifichieber celesienilverManned). Netocom State State Systifichieber celesienilverManned). Netocom State Stat	SQL server could not find COM port or Virtual COM port and cannot start SQL services.
	COK .	1. Press OK.
	Error start with: System.NullRefeenceException: Object reference not set to an instance of an object. at Medoc.SysInfoHelper.GetSeriaPortNanes()	 Copy the file Add SERIALCOM to registry.reg from Q-SENSE installation CD to your computer. The file located in Medoc Manufacture Remote Assistance Tools folder.
		 Run the Add SERIALCOM to registry.reg file by double click on it.
		 Connect the computer to Q-SENSE communication port using USB cable. Make sure that the driver is installed properly according to Q-SENSE installation instructions.
		5. Turn ON the Q-SENSE system.
		6. Start MEDOC Main Station.
		If the problem persists, contact Medoc service.

24.2 Faulty System Responses

Table 43: Faulty System Responses

Malfunction	Solution	
Thermode does not heat	The Thermode does not heat up, and remains at room temperature or cold. The Q-SENSE Temperature status, displayed on status bar, indicates a temperature other than the current Rest Mode temperature.	
	The malfunction may be caused by the Thermode, or by the voltage supplied from the power supply.	
	1. Verify that the Q-SENSE system is in Online mode.	
	Power OFF the Q-SENSE unit, and replace the Thermode.	
	3. Power ON the Q-SENSE unit and check if the Thermode heats up.	
	 Verify that the Q-SENSE system is not in Safe Mode (Safety indicator, in status bar, is GREEN) 	



Malfunction	Solution	
Power Failure	Perform power supply and cable check.	
	 Check that the power cable you are using is not damaged. Make sure that the cable and its connectors are undamaged. 	
	a. Check that the power cable that you are using is connected properly to Q-Sense Power COM.	
	Perform power-switch check.	
	 Turn ON the main power switch of the Q-SENSE system. A green LED indicates it is ON. 	
	* If the problem persists, contact Medoc service for support.	
Constant Communication	1. Perform COM port check.	
problems	a. Check that MEDOC Main Station is set for automatic COM port detection (Settings→Software Settings).	
	2. Perform Power checks (see above).	
	3. Perform communication cable check.	
	 b. Check that the communication cable is connected properly. 	
	 c. One end must be connected to the back of the Q- SENSE COM port; the other end to your laptop. 	
	4. Turn ON the Q-SENSE system.	
	 Disconnect and then reconnect the communication cable from the COM port of the Q-Sense System, while the Q- Sense system (software and hardware) is running. 	
	6. Restart the Q-Sense system.	
Automatic Calibration Failure	 Verify you selected proper COM port for Thermometer in Software Settings menu. 	
	2. Make sure that the operating system decimal symbol is set to dot.	
	To change decimal symbol go to Control Panel→Regional and Languages Options→Regional Options→Decimal Symbol.	
Display Settings	5. Make sure that the font setting on your laptop is set for 96dpi.	
	 Make sure that your screen resolution is set according to instructions in section 364.1, Page 36. 	
No TEC Response	1. Power and communication Check.	
	a. Check that the system is ON.	
	Proprietary 172 of 18	



Malfunction	Solution
	b. Check that communication can be established.
	2. Thermode connection check.
	 Make sure that the enabled Thermode is connected properly to the Q-Sense system.
	3. Restart MEDOC Main Station and Q-Sense system.
	4. TEC temperatures refresh check.
	a. Check that the TEC temperature in the software status bar is not freezing i.e., see that the temperature value changes; there should be a fluctuation of about 0.1°C.
	5. TEC response check for Q-sense Thermode only:
	a. Start the Q-Sense system and software.
	b. Go to Settings \rightarrow Hardware Settings \rightarrow Thermode.
	 Select the Q-sense Thermode then press the calibration tab.
	 Change the Thermode temperature using the temperature spin box.
	e. Check if the temperature in the status line has changed according to the set temperature.
	6. Consult Medoc service for further instructions.



25 Appendix M – Licensed Options

25.1FTP Support

Enables to upload test results to FTP location for collaboration, multi-center studies or backup. There is an option to automatically save and upload test results immediately after test completion (requires internet connection).

25.1.1 Export Configuration

Configure the FTP server which will be used when FTP Server is selected during manual export, or when automatic export option is enabled.

When exporting results to an FTP server the results will be automatically named according to the following rule: <Site Name>_<Patient Name>_<Test Date & Time>

To access the Export Configuration panel: from the top menu bar, select **Settings** and then **Software Settings**.



Figure 107: Open Software Settings

In the Software Settings window that opens, select the **Export** tab.



Figure 108: Software settings export tab selection

The following window will be displayed:



Software Settings	O
Communication Users Export Report Options Results Fields Patient Fields Auto Export Settings Export automatically	
Upload Server	
Server URL:	Check connection
Password:	
]
0	K Apply Cancel

Figure 109: Software Settings Export Tab

25.1.1.1 Configure the FTP Upload Server

To configure the server enter the following details:

- 1. Server URL the FTP address. Use a specific address including the folder to which you want the results to be uploaded to. For example: ftp://ftp.my-ftp-server.com/FolderName
- 2. User name and password required to access the FTP server
- 3. Site name an identifier which will be used in the name of the uploaded result file to distinguish the uploaded file from other files uploaded from different locations.
- 4. Click Apply to confirm
- 5. Check the Connection by clicking the "check connection" button the red indicator to the right will change to green if the connection is successful.

25.1.1.2 <u>Automatic Export</u>

If automatic export is enabled test results will be automatically uploaded to the FTP server at the end of a test. A message will be displayed at the end of the test, indicating whether or not the upload attempt was successful or not. The computer must be connected to a network in order for the results to be successfully uploaded at the end a test.

To enable automatic export:

- 1. Make sure the FTP server details are entered correctly and check the connection.
- 2. Mark the "Export automatically" checkbox.
- 3. Make sure auto-save option is enabled (see 10.3.1).



25.2CoVAS

Enables continues pain rating using the USB CoVAS device.

25.2.1 CoVAS Calibration

Calibrating the CoVAS ensures the CoVAS range 0 to 100 corresponds with the left most and right most point of the slider, respectively.

The CoVAS should be calibrated when:

- It is used for the first time
- Software or firmware are re-installed

To Calibrate the CoVAS:

- 1. Connect the CoVAS to the Computer (USB connection).
- 2. In the Hardware Settings windows, select the USB CoVAS tab.

🕘 Hardware Settings	8
General Safety USB COVAS Thermode	
COVAS calibration wizard	
Welcome to the COVAS calibration wizard	
This COVAS calibration wizard will allow you to calibrate, modify or repair your COVAS hardware. To continue, click Next.	
ОК	Apply Cancel

Figure 110: CoVAS Calibration Tab

- 3. Click the Next button and follow the instructions shown on screen (a short version is given here)
- 4. Move the CoVAS slider all the way to the left and click the Next button.
- 5. Move the CoVAS slider all the way to the right and click the Next button.
- 6. Click Next to save the calibration results.



- Q-Sense Technical Reference Manual
- 7. Click the Finish button to complete the calibrating process; then click OK.

25.3 Patient Record Control

Allows full customization of patient fields including the options to add, remove, rename and control input.

25.3.1 Control Patient Field Entry Format with Input Masks

When you have several users entering patient data, you can control how the data must be entered in specific fields in order to maintain consistency. For example, you can set an input mask for the ID field so the ID numbers will have a very specific format.

Input Masks

An input mask consists of a string of characters which represent the format in which the data should be entered. The input mask strings consists of **Regular Characters**, **Special Characters**, and **Literal Characters**.

The **Regular Characters** represent the alphabetic or numeric characters that must or may be used to enter data in the field. Refer to Table 44 for a list of Regular Characters and their definitions.

The **Special Characters** represent the case and various delimiters and symbols. Refer to Table 45 for a list of Special Characters and their definitions.

The **Literal Characters** are characters that are not Regular Characters or Special Characters. They are inserted as is into the edit box of the field and the curser will skip over them during editing. The use a Regular Character or a Special Character as a Literal Character, add a backslash (\) before.

Character	Definition	
L	User must enter a letter (A-Z, a-z)	
1	User can only enter a letter (A-Z, a-z) but it is not required	
А	User must enter a letter (A-Z, a-z) or a number (0-9)	
а	User can only enter a letter (A-Z, a-z) or a number (0-9) but it is not required	
С	User must enter any character	
с	User can enter any character but it is not required	
0	User must enter a number (0-9)	
9	User can enter only a number (0-9) but it is not required	
#	User can enter only a number (0-9) and a plus (+) or minus (-) but it is not required	

Table 44: Regular Characters

Table 45: Special Characters

Character	Definition
>	All the characters that follow are converted to upper case
<	All the characters that follow are converted to lower case
<>	No case checking is performed



/	Used to separate months, days and years in dates. The character used depends on the regional settings of the computer
:	Used to separate hours, minutes and seconds in time values. The character used depends on the regional settings of the computer
\$	Currency symbol. The symbol used depends on the regional settings of the computer

Examples

- A mask for entering a telephone number: (999)000-00-00. Each '0' Regular Character requires a number in the corresponding position. No number in these positions can be omitted. The Regular Character '9' represents an optional area code. Only a number can be entered in these positions but it is not required. The '-' and '(' and ')' characters in the mask are literal. A telephone number entered could be: (555)222-33-22 or 222-33-22.
- 2. A mask for entering a sequence with numbers and letters: \A>LL-00. The `\A' represents a Literal Character which will be entered as `A'. The `>' specifies that following characters will be converted to upper case. The `LL' indicates that two letters must be entered in this position. The `-` is a Literal Character. The `00' is a placeholder for 2 numbers. An entry to this field can be: ASD-88.

25.3.2 Add/Remove Patient Fields

To define and add Patient Card Fields to the Patient Details dialog:

- 1. From the **Patient Card Fields** list, select the required fields from the Mandatory or Optional fields, or add a new field from the user-defined list. Add an input mask if needed.
- 2. To remove a Patient Field, remove the selection mark from the field
- 3. Click the **Apply** button to confirm changes.





To add Patient Fields to the Patient Table:

- 4. Select fields to add from the Patient Card Fields list and click the **Move Right** button
- 5. Click the **Apply** button to confirm changes

To remove Patient Fields from Patient Table:

- 1. Select fields to remove from the Patient Table Fields and click the **Move Left** button
- 2. Click the **Apply** button to confirm changes

To restore the default Patient Fields:

- 1. Click the **Restore** button to restore default Patient Fields.
- 2. Click the **Apply** button to confirm changes



25.4Pain Rating

Enables advanced options for monitoring Patient Response: customized range for numerical pain rating scale, multiple ratings per trial.

25.4.1 Pain Rating Parameters Configuration

Pain rating is available only for Ramp & Hold program.

Advanced parameters available with pain rating license can be configured while creating or editing any Ramp & Hold program.

The following options can be configured:



Figure 113: Pain Rating Parameters

Enable Pain Rating checkbox – When checked enables pain rating patient response

First pain rating -determines the time of the first pain rating.

Default value is "0", meaning that the first pain rating will occur at the end of the trial.

Pain rating interval – determines the interval between pain ratings.

Default value is "0", meaning there will be only one pain rating in trial, at time determined by first pain rating parameter value.

<u>Note:</u> The amount of paint ratings in trial is determined by combination of first pain rating, pain rating interval and duration time parameters value. For example, if the duration time of the trial is 20 sec, first pain rating value is 5 sec and pain rating interval value is 5 sec, there will be 4 pain ratings in the trial 5, 10, 15 and 20 sec after reaching the destination temperature.

Waiting time for response – determines the time for which the pain rating dialog will remain on the screen waiting for user to respond.

In addition, **Pain rating limits** can be configured from Test Configuration tab in Test Settings window:



🗶 Test Settings 🔞					
Test Configuration Instructions Configuration Body Sites					
Pre-Test Configuration					
Low Temperature Margin (°C):					
Success Time (sec): 1 Maximum Time (sec): 20					
Test Configuration					
Low Destination Margin (°C): 0.1 High Destination Margin (°C): 0.1					
Post-Test Configuration Post Test Time (sec): Add Comments Save automatically					
Test Display Configuration					
Show Normative Data					
Default Trial Scale: 16 Default Time Scale (sec): 100 :					
Temperature Scale Configuration					
min 15 max 55 Min. Temp Unit (°C) 5					
Temperature Limits Pain rating Limits					
min 20 max 50 min 0 max 100 min					
ОК Арруу Салсе!					

Figure 114: Pain Rating Limits



26 Appendix N – fMRI functionality

26.1 fMRI Thermode



Q-sense fMRI Thermode and accessories are defined and marked as MR Conditional equipment according to the ASTM F2503-08.



Q-sense MR filters are defined and marked as MR Unsafe equipment according to the ASTM F2503-08.



WARNING:

Make sure that the Thermode is unblemished in order to prevent potential harm to the test subject.



WARNING:

To avoid the risk of electric shock, this equipment must only be connected to a supply mains with protective earth.



The fMRI Q-sense system was tested in the following MR environments: Field Strength: 3T and 7T; Max. Spatial gradient: 72 mT/m Max. dB/dt: 346 T/m/s; RF Fields: max. power 35KW pep; TX bandwidth: 800kHz; RX bandwidth: 500Hz-1MHz; maximal RF field strength: 24,7 µT CP SAR: whole body 4 W/kg, head 3.2 W/Kg



The fMRI Q-sense system was tested in the following MR Sequences:

- 1. Echo Planer Imaging (EPI) bold (fMRI)
- 2. rf_noise and rf_noise spectrum electromagnetic noise interference (Siemens protocol)

26.1.1 *Q-sense system configuration in fMRI mode:*

The fMRI Q-sense system includes the following components:



Control room thermode cable

• **AR** Q-Sense fMRI Thermode



The Q-sense fMRI thermode incorporates an EMI filter designed to reduce and prevent electromagnetic interference noise from penetrating into the scanner room. The fMRI filter includes:

- Minimal insertion loss : 10MHz/13dB, 100MHz/50dB, and 1GHz/70dB.
- Capacitance: 4000pF
- Non-ferromagnetic filter components



The filters are to be connected via appropriate 9-pin D-type cutouts in the penetration panel, with the thermode air tube to be passed through a waveguide of appropriate dimensions (according to the MRI scanner manufacturer design).

Loosely connected cable connectors or use of conductive tape or braid may lead to interference noise and imaging artifacts.

The fMRI Thermode is "MR Conditional". An extended 13 [m] cable connects between the Q-sense system, which is located outside of the MRI control room, and the Thermode that is located in the MRI room. The part of the thermode inside the room is 10 [m] long, and the part outside the room is 3 [m] long.



"MR Conditional" means an item that has been demonstrated to pose no known hazards in a specified MR environment with specified conditions of use.

26.1.2 Installation – MR Filter

- 1. Complete the setup and installation procedure as described for a regular Q-sense system, according to the System Block diagram below.
- 2. Connect the system devices according to the following stages:





Figure 115- fMRI room setup

- 2.1. Connect the Q-sense system so it would be placed in the MRI control room, while the Thermode itself is placed in the MRI room, as shown in the diagram:
- 2.2. Make sure your PC, Q-sense fMRI unit are all OFF.
- 2.3. Install the fMRI thermode Control Room part (see Figure 116) according to the following instructions:
 - 2.3.1. Connect the D type panel filter connector on your MRI room penetration panel.
 - 2.3.2. Connect Q-sense fMRI Connectors to the Q-sense fMRI unit





Figure 117: Q-sense fMRI Thermode Shielded room part AS 00412

2.4. See Figure 117, for description of the fMRI Thermode- Shielded room part components:

- 2.4.1. Q-sense fMRI Thermode
- 2.4.2. Q-sense fMRI Thermode connector
- 2.4.3. Connection to MRI
- 2.5. Connect the main power cord to the Q-sense system.
- 3. The fMRI Thermode- Shielded room part should be placed inside the fMRI room. All the other devices' connectors should be placed outside of the fMRI room.
- 4. Connect the required devices and accessories, as described in Figure 115
- 5. Conduct three phantom tests:
 - 5.1. Conduct the test, when the Q-sense is OFF,
 - 5.2. Conduct the test, when the Q-sense is ON and software is OFF,
 - 5.3. Conduct the test, when the Q-sense and software are ON and software test is running.
- 6. The installation is finished if all three tests are passed successfully.



The SIP/SOPs (signal input and outputs) must be connected to medical grade safety approved equipment for Canada and the United States, by a recognized certification agency, which is connected to the common protective earth of the system.



26.2 TTL functionality

Q-sense systems with FMRI thermode functionality support TTL functionality which can be used for synchronization with external systems.

26.2.1 Q-sense fMRI side panel



Figure 118: Q-sense fMRI side panel

26.2.2 TTL set up

Q-sense can send 1-bit TTL from the side panel TTL OUT BNC connector (see Figure 118)



Figure 119: HW settings for Q-sense fMRI

The polarity of the TTL can be configured in Q-sense hardware settings. See figure 119



1. TTL Polarity:

- **1.1.TTL OUT:** Configure the Polarity of TTL OUT pulse: 5V or (-)5V. For 5V: The TTL is 0V when Idle and 5V when triggered. For (-)5V: The TTL is 5V when Idle and 0V when triggered.
- **1.2.TTL IN:** The polarity of the TTL IN does not need to be configured. Q-sense will automatically detect if it is 5V or (-)5V.

2. TTL Pulse out Duration:

2.1.Configure the duration of the TTL OUT. The allowed range is: 50 – 1000 msec.

26.2.3 Standard Event Configuration



TTL trigger shall be sent only if the standard event is configured in the program

To configure standard events in the required program, follow the steps detailed below:

1. From the Programs screen, select the program you want to configure standard events for and

double click to open the program editor or click on the edit program button



2. From the list of parameters, select the Standard Events option

ISI Min (sec)	8 0
ISI Max (sec)	Ø 0
Standard Events	
Temperature Events	

Figure 120: Standard Events in Program Editor

3. Click the blue arrow on the right to open the standard event editor.

ISI Max (sec)	🖸 🖸 0
Standard Events	
Temperature Events	Standard Events TTL Sound
umber of sequences in the test : 1	• Onset 1 🔲 🗆

Figure 121: Standard Event Editor

4. Mark TTL option to activate TTL OUT signal output for various standard events (Onset, Temperature UP/DOWN, End of duration, End of test).

26.2.4 TTL specifications



Parameter	Description
Synchronization options TTL Input	Externally trigger the onset of stimulus Voltage: ± 5V Current: 10 – 15 mA Minimum Duration: 10 msec.
Synchronization options TTL Output	Can indicate various events during stimuli (depending on protocol used) Voltage: ± 5V Current: 2 mA Duration: 50 – 1000 msec

26.3 Blower control

Regular thermode 30x30 is cooled by an internal fan in the thermode enclose. fMRI 30x30 thermode is cooled via the air tube by a blower inside the main unit.

It is possible to turn off the blower for work in an office environment.

In order to do that there is a need to defined two types of thermodes in the HW settings. Thermode type "30x30" and thermode type "fMRI 30x30". By setting thermode type 30x30 as current you will disable the blower and activate the regular thermode fan. Setting thermode type as fMRI 30x30 will activate the blower and disable the regular thermode fan.

For additional info please refer to section 10.2.3



Use of a thermode without appropriate cooling may result in the thermode providing inaccurate temperatures, and may result in the test being stopped due to safety mechanisms being activated, especially in prolonged cold detection threshold protocols.

Ensure that the correct thermode type is selected in HW settings.

